UNCLASSIFIED SÉCRÉN

LA-14066-H History

Not for public release: distribution is limited.

Tracing the Origins of the W76: 1966–Spring 1973 (U)

November 3, 2003

Redacted Version

R WEARDN PATA NUCL Sigma 1 Nuclear Weapo Critical Design Information DoD Directive 5219.2 Applies

RESTRICTED DATA This accument contains Restricted Data as defined in the Atomic Energy Act of 1964. Linauthorized disclosure subject to administrative and cruminal sanctions.

Classifier: Michael Pankratz Derived from: LA-4000, Rev. 8, 9/02 July 14, 2003



Los Alamos National Laboratory is operated by the University of California for the United States Department of Energy under contract W-7405-ENG-36.

LASSIFIED

LA-14066-H History UNCLASSIFIED

November 3, 2003

An Affirmative Action/Equal Opportunity Employer

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the Regents of the University of California, the United States Government nor any agency thereof, nor any of their employees make any warranty, express or implied, or assume any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represent that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise does not necessarily constitute or imply its endorsement, recommendation, or favoring by the Regents of the University of California, the United States Government, or any agency thereof. The views and opinions of authors expressed herein do not necessarily state or reflect those of the Regents of the University of California, the United States Government, or any agency thereof. Los Alanos National Laboratory strongly supports academic freedom and a researcher's right to publish; as an institution, however, the Laboratory does not endorse the viewpoint of a publication or guarantee its technical correctness.



ii

CLASSIFIED

LA-14066-H History

Tracing the Origins of the W76: 1966–Spring 1973 (U)

Betty L. Perkins

November 3, 2003

ersio ted "

NUCKAR WAAPON DATA Sigma 1 Critical Nuclear Wedpor Design[Information DoD Directive 5210.2 Applies

RESTRICTED DATA This document contains Restricted Data is defined in the Atomic Energy Act of 1954. Unauthorized disclosure subject to administrative and criminal sanctions.

Classifier: Michael Pankratz Derived from: LA-4000, Rev. 8, 9/02 July 14, 2003



Los Alamos NM 87545

ASSIFIED

iii



This Page Intentionally Left Blank



ICLASSIFIED

ACKNOWLEDGMENTS

John C. Hopkins read most of this document in its initial draft form. Dale Reid Worlton and Harold H. Rogers, Jr. read several chapters. Stephen A. Becker, Thomas E. Larson, B. Pruitt Ginsburg, Paul S. Dunn, and J. David Olivas read sections that relate to their areas of expertise. Several others, noted as personal communication in the source references, provided relevant information. The author must thank all of these people for their helpful corrections and comments.

The author also wishes to thank Michael C. Pankratz for designating the proper level of classification; this required the reading of the complete document. The editing group at Los Alamos must be commended for their faithful efforts. In addition, the staff at the Records Center, Archives, and Report Library must be thanked for maintaining and allowing the author to have access to the historical records. Without primary sources the document would not have been possible.

Explanation of Primary Sources

The author used as the primary database for this report sources (such as reports, letters, TWXs, memos, progress reports, and minutes of meetings) that were available to the author within the Los Alamos National Laboratory. It would be expected, considering the extremely large number of relevant documents and their diverse distribution, that not all the "in-house" information was retrieved. (It will be noted that because the author is not an X-Division employee, information in the X-Division on-line database was not accessed nor were the device drawings located in the X-Division vault.)

The author takes full responsibility for locating the primary sources, for their interpretation, and for the report as written. However, it will be noted that sources may differ or fail to give complete information, or the author may have made an error. If the reader is engaged in a specific project, the author recommends review of the applicable sources. In addition, the original engineering drawings should be located and used.

In this report, the author in citing the source has indicated the location of that source at the time the report was submitted for publication. It will be noted that B refers to the area in which the document is located in the Los Alamos Records Center. The numbers after the letter refer to the location of the document. (All of the cited documents having B locations are part of the Agnew director files.) The use of C again refers to the area location of the specific file in the Records Center, and the numbers note the specific location. Use of SLSL or GAMF indicates film; the number that follows is the number of the film roll; these are also in the Records Center. Those sources with A and a series of numbers are located in the Los Alamos archives; the numbers after the A relate to the box and folder. Reports for which a specific location is not given are, in general, in the Los Alamos report library.

LA-14066-H

UNCLASSIFIED

v

This Page Intentionally Left Blank

SECRETIRI UNCLASSIFIED

LA-14066-H

vi

CONTENTS

ABSTRACT	xiii
CHAPTER I. INTRODUCTION	I-3
 A. Explanation 1. Assignment 2. Overview 	. I-3 I-3 I-3
 B. Los Alamos Scientific Laboratory Management Structure and Philosophy 1. Norris Bradbury 2. Harold M. Agnew 	I-4 I-4 I-5
 C. Weapon Group Designations/Responsibilities, Support and Basic Research Groups, and Committee Functions	. I-9 I-9 I-13 I-14 I-15
 D. Naming Names. 1. Operation. 2. Event 	. I-15 . I-15 . I-16
LIST OF FIGURES	. I-1/ I-1

(b)(3)

6. Lessons Learned...... II-37

(b)(3)

FIED U

LA-14066-H

vii

UNCLASSIFIED (KE/IRA SF/ (b)(3) (b)(3) 1. Initial Considerations II-58 (b)(3) 2. Field Tests of the Model 2 II-99 (b)(3) 1_Proposal_____ II-105

(b)(3)

H. The Los Alamos Position May 1973	П-111
LIST OF FIGURES	II-2
LIST OF TABLES	ІІ-2

(b)(3) ·

1. Livermore: A Small, Lightweight Secondary	III-3
2. Los Alamos: 1959–1964	III-4
B. Los Alamos 1965–1968 Design Program Involving a Small Secondary	<u>III</u> -10
2. Early Mk 18 Program (b)(3)	III-13

ļ

C.	BM	IP Program and the Mk 18	Ш-15
	1.	Concerns With the Primary in Terms of Secondary Design	Ш-15

,	١.,	•	10	•
ſ	n	n	63	n
٩	-	,	~،	

T. Background	
(b)	(3)
F. Looking Ahead	
LIST OF FIGURES	Ш-1
LIST OF TABLES	Ш-2

:

CHAPTI	ER IV. MATERIALS, ENGINEERING, AND VULNERABILITY	IV-3
А.	Materials	
	1. Considerations	IV-3
	(b)(3)	
	4. PBX 9501	IV-22
	(b)(3)	
	7. Pu Alloys	IV-34
	8. Material Compatibility	IV-54
В.	Engineering	IV-54
	1. Tools	IV-54
	(b)(3)	
	3. Reservoir Designs to Provide Minimum Helium in the Boost Gas.	IV-59
	4. Detonators	IV-63
	5. Engineering Computer Codes	IV-65
	6. Assembly	IV-65
С.	Vulnerability	IV-66
	1. Considerations	IV-66
	2. NTS Vulnerability Tests	IV-67
	3. Calculations	IV-90
LA-1406	6-H SECKET/BID	ix
	UNCLASSIFIED	

LIST OF FIGURES	 -1
LIST OF TABLES	 '-1

CHAPTI	R V. WEAPON PROGRAMS AND CHANGES IN THE STOCKPILE:	
196	5–MAY 1973	V-3
Α.	Phase 3 Programs at Los Alamos	V-3
	1 Assignments	V-3
	2 Weapon Programs	V-4
	 Conclusions Concerning the Phase 3 Program 	V-48
B.	Advanced Development and Pre-Phase 3 Programs	V-49
	1. 1965	V-49
	2 1966	V-53
	3 1967	V-57
	4 1968	V-60
	5 1060	V-71
	J. 1909	V-82
	0. 1970	V-01
	7. 1971	V-91 V 100
	8. 1972 and First Quarter of 1973	¥-100
	9. Conclusions	V-112
C.	Trends	V-113
	1. Change in Composition	V-113
	2. New Weapon Systems	V -116
	3. Decreased Levels of Funding	V -120
	4. Tighter Test Requirements/Cost Increases	V-121
	5. Result	V-121
LIST OF	FIGURES	V-1
101.01	TOOTED	

	STITOR A STRATEOR MISSIEL WARDEND	
A.	Mk 18	VI-3
	1. The Ouestion of a Pie Split	VI-3
	2. The Question of the Primary	VI-4
	3. Uncertainty in the Program	VI-12
	4. The Question of the Secondary	VI-16
	5. Where Do We Go From Here?	VI-17
	6. Phase-Out?	VI-21
	7. Transformation	VI-23

SECRETIED UNCLASSIFIED LA-14066-H

x

B.	Mar	k 400 Program	VI-24	
	1.	Livermore vs Los Alamos: The Navy's Submarine-Launched		
		Ballistic Missile (SLBM) Warheads-W47, W58, and W68	VI-24	
	2.	Advanced Planning: 1966-August 1969	VI-26	
	3.	Advanced Systems Studies: September 1969–September 1971	VI-27	
	4.	A New Ballistic Missile System	VI-34	
	5.	Approval for a Mk 400	VI-36	
	6.	At Last, a Phase 2	VI-40	
		(b)(3))
	8.	Where is the Phase 3?	VI-51	
	9.	Continued Hope for a Phase 3? Maybe	VI-58	
	10.	Finally—A Phase 3 Assignment	VI-61	
	11.	Significance	VI-64	
С.	Hard	old Agnew	VI-64	•••
	1.	Proponent	VI-64	
	2.	Reliability	VI-65	
	3.	Advantages of Diversity	VI-66	
	4.	Engineering Philosophy	VI-67	
	5.	Cooperation	VI-67	
	6.	Program Balance	VI-68	
	7.	Yield: The Confetti Argument	VI-69	
	8.	Capability	VI-69	
	9.	Promise of the Next Strategic Missile Warhead	VI-70	
	10.	Morale	VI-71	

SFARENRY UNCLASSIFIED

xi



This Page Intentionally Left Blank

UNCLASSIFIED

LA-14066-H

xii

TRACING THE ORIGINS OF THE W76: 1966–SPRING 1973

by

Betty L. Perkins

ABSTRACT (SRD)

The objective in writing this report was to place the development of the W76, before it entered Phase 3, in a historical perspective. The author has rather arbitrarily chosen to consider for this pre-Phase 3 history, the history of the weapon program at Los Alamos during the years 1966–May 1973.

The report tries to provide some understanding as to why, in the spring of 1973, the Los Alamos Scientific Laboratory received the Phase 3 assignment and why the assignment was important to the future of Los Alamos. In addition, the report provides insight into why historically the design of the W76 evolved as it did.

Chapter I provides general information including the organization of the Laboratory during the time-period of interest and the definition of what is included in the different phases in weapon development.

Chapter II discusses the work on primary design.

(b)(3)

ASSIFIED

xiii

UNCLASSIFIED SECRET/RD

(b)(3)

Chapter IV briefly describes the early development effort for several of the materials that would be important in the W76 program,

(b)(3) The engineering status of several ancillary components such as detonators and gas storage systems is reported. Chapter IV notes the vulnerability tests relevant to the early LASL XW76 weapon program.

Chapter V includes a brief discussion of the history of the weapon systems assigned to Los Alamos as Phase 3 programs during the 1966-spring 1973 period. The extensive effort that was required for the various Phase 1 and 2 programs and the early advanced development programs under consideration during these years is also discussed. It is noted that despite this effort, the Los Alamos weapon teams failed during 1966-1972 to win a viable Phase 3 assignment to develop a warhead for a strategic missile weapon system. The chapter also includes some trends in the U.S. nuclear stockpile that are important in understanding the 1970-1980s weapon programs.

Chapter VI outlines the Los Alamos effort for the Mk 18 and the later Mk 400 programs; programs that served as the precursor programs to the W76. This chapter provides insight on the Los Alamos effort to obtain a Phase 3 assignment for a strategic warhead and the success in this effort that resulted in the long desired award of the XW76 program to Los Alamos.

AFAREDMAN LA-14 UNCLASSIFIED

UNCLASSIFIED

CHAPTER I TABLE OF CONTENTS

CHAPTE	ER I. INTRODUCTION I	-3
Α.	Explanation I	-3
	1. Assignment I	-3
	2. Overview	-3
B.	Los Alamos Scientific Laboratory Management Structure and Philosophy I-	-4
	1. Norris Bradbury	-4
	2. Harold M. Agnew	-5
C.	Weapon Group Designations/Responsibilities,	
	Support and Basic Research Groups, and Committee Functions I-	.9
	1. Weapon Groups: 1966-September 1972 I-	-9
	2. Weapon Groups: Reorganization and September 1, 1972, Designations I-	-13
	3. Support and Basic Research.	-14
	4. Committees I-	-15
D.	Naming Names	-15
	1. Operation	-15
	2. Event	16
Е.	Phases in Weapon Development I-	17

LIST OF FIGURES

Figure I-1. Laboratory Organization August 1967	I-6
Figure I-2. Laboratory Organization November 1970	I-8

SECRET/SON UNCLASSIFIED

I-1

UNCLASSIFIED

This Page Intentionally Left Blank

SEPERATION LA-14 UNCLASSIFIED

I-2

CHAPTER I. INTRODUCTION

A. Explanation

1. Assignment

The assignment given to the author was to outline the history of the development of the W76 warhead (presently carried on both the Navy's Trident C4 and D5 submarine-launched ballistic missiles). Because the Los Alamos Scientific Laboratory [LASL] received the Phase 3 assignment for this warhead in the spring of 1973, it would be reasonable to assume that a history of the W76 would cover only the period from the Phase 3 assignment until the initial operational capability of the W76 was achieved in October 1979 (Poseidon back-fit). But history is continuous. What happens at one point in time is dependent upon what happened earlier.

In order to set the development of the W76 in the necessary perspective, give some understanding as to why in the spring of 1973 LASL received the Phase 3 assignment and why the assignment was important to the future of Los Alamos, and indicate several reasons why the design of this device evolved as it did, a history of work prior to 1973 is required. The author has rather arbitrarily chosen to consider for this history, the history of the weapons program at Los Alamos during the years 1966–May 1973. (However, to give continuity, some aspects of the program are also described for work completed before 1966.) This pre-Phase 3 effort at Los Alamos is the focus of this report.

However, the author must insert a warning to the reader. It must be noted that to further increase the complexity that is history, it is almost impossible to identify all the factors that go into determining actions during a specific era. In addition, the description of an event is dependent upon the available "data set" of historical documents. Moreover, how an event is described in a point in time is dependent on what happens later and on our own personal experiences, knowledge, and "mindset." Thus, no history can be completely objective.

2. Overview

Before the award of the design effort for the W76 to Los Alamos, the U.S. nuclear weapon designers had been required—by the introduction of MIRVed (Multiple Independently Targeted Reentry Vehicle) missiles into the U.S. weapon arsenal—to develop lightweight/small warheads for use in the missiles' reentry vehicles.

(b)(3)

LA-14066-H

LASSIFIED

NCLASSIFIED SEACAREAT/BD

(b)(3)

Chapter IV will briefly describe the early development effort for several of the materials that would be important in the W76 program.

(b)(3)

The engineering status of several ancillary components such as detonators and gas supply systems will be reported. Chapter IV will also note the vulnerability tests relevant to the LASL XW76 weapon program.

Chapter V will outline and briefly discuss the history of the weapon systems assigned to Los Alamos as Phase 3 programs during the 1966–1972 period. In addition, mention will be made of Phase 1 and 2 programs and early development programs under consideration during those years. This chapter will attempt to inform the reader as to the extensive effort that was required. However, as Chapter V will also describe, the Los Alamos weapon teams failed during 1966-1972 to win a viable Phase 3 assignment to develop a warhead for a strategic missile weapon system. The W62 for the Minuteman III with a Phase 3 of 1964 went to Livermore. The W68 for the Navy's Poseidon submarine with a Phase 3 of 1966 also went to Livermore. Earlier, the W56 (the warhead for the Minuteman I, II) and the W58 (the warhead for the Navy's Polaris) had also gone to Livermore. The Chapter will also note some trends in the U.S. nuclear stockpile that were important for the weapon programs at the Livermore, Sandia, and Los Alamos laboratories.

Although the program was finally canceled, of particular importance to the later W76 development was the Mk 18 program. This program will be covered in some detail in Chapter VI. The Navy's Mk 400 program was the precursor program to the W76. The history of the Mk 400 program will also be outlined in Chapter VI. This chapter will discuss the vital question: who would win the Phase 3 for the Mk 400 (XW76) Los Alamos or Livermore?

B. Los Alamos Scientific Laboratory Management Structure and Philosophy

1. Norris Bradbury

Norris Bradbury served as the director of the Laboratory at Los Alamos from October 1945 until September 1970. When he accepted this job and became director in October 1945 just after the end of WWII, he promised that he would serve for six months. But the six months of service stretched into twenty-five years.

In a January 1967 letter to Charles Winter, Deputy Director of the Division of Military Application, Bradbury described the Laboratory, "Los Alamos is organized on a facility and technology basis; LRL is organized more on a project basis." Bradbury also noted, "Internally in the Laboratory, the weapon program is steered by a committee chaired by the Laboratory Director and comprised of Assistant Directors and relevant Division Leaders. Basic decisions are made by this group, the members of which carry the authority within their respective areas of responsibility to implement them. More detailed discussions and decisions within the framework

SECRETHE

I-4

UNCLASSIFIED

of agreed upon programs is made by the Weapons Working Group which is chaired by an Assistant Director of the Laboratory. Matters of detailed design, scheduling of tests, analysis of results, and so on are dealt with within this group. Still more specialized groups ...cover areas of weapon related research. Since much of the work of the Laboratory must be coordinated with that of the Sandia Laboratory, cross Laboratory groups exist again with membership having authority to implement their decisions." Figure I-1 outlines the Laboratory organization in August 1967 and indicates the organization in terms of the Assistant Directors and Technical Division Leaders noted in Bradbury's letter to Winter.

In his letter to Winter, Bradbury also outlined what he felt was "essential progress" in the U.S. nuclear weapon program. He stated, "To me this means that the United States should be and always remain ahead of any potential enemy with respect to every aspect of nuclear technology for military defensive or offensive use. This includes everything from maximizing the efficiency (yield per pound), relating nuclear devices to military needs and delivery systems in the optimum way, production of and resistance to specialized effects of nuclear explosions (hardening, x-rays, ER, blast, etc., etc.), the search for new possibilities in both explosive and effects technologies, as well as potential new military uses or the application of nuclear devices to existing, proposed, or hypothetical military systems. It clearly includes space as a possible theater for military nuclear activities as well as the sea and the sea bottom."

After serving so faithfully for so many years, by the end of the 1960s Bradbury was tired; he felt that a new director should be appointed. This person proved to be Harold M. Agnew.

2. Harold M. Agnew

Harold M. Agnew was born in Denver, Colorado, on March 28, 1921. After obtaining, in 1942, an undergraduate degree with a major in chemistry from the University of Denver, Agnew was immediately recruited to work at the University of Chicago as part of the Manhattan District's team. Here on December 2, 1942, under the direction of Enrico Fermi, he was involved in bringing, at Chicago's Stagg Field, the first atomic pile into a self-sustaining nuclear chain reaction.

Agnew's association with the Los Alamos Laboratory (Project Y) began on April 1, 1943, when he joined the Experimental Physics Division. (Agnew has recalled that he was sent to Los Alamos as a "time-out" period because he had received a fairly high exposure during his work at Chicago. He has also joked that he was sent to Los Alamos because Los Alamos demanded his wife; she was a very good secretary.) In 1945 Agnew flew, as a scientific advisor, with the 509th Bombardment Group on the first nuclear weapon strike.

After the war, in 1946, Agnew returned to the University of Chicago to again work under Fermi. He received his MS in 1948 and a PhD in physics in 1949.

Agnew returned to Los Alamos in 1949. From October 1951 until May 1953 Agnew served as the assistant to the technical associate director. In February 1956, he became the assistant division leader in the weapons division; he held this position until October 1957, at which time he became the alternate division leader. He continued in this position until 1961. At that time he took a leave of absence to become the Scientific Advisor to the NATO Supreme Allied Commander in Europe. In August 1964, when he returned to the Laboratory at Los Alamos, he was given the assignment of W-(Weapon) Division leader (see Figure I-1).

/SECRETINE

INCLASSIFIED

¹N. E. Bradbury to Mr. Charles Winter, DIR 2062 (SRD) (January 9, 1967), 8 pp., A99-019, 198-8.

I-5





VIEWGRAPH 2

Figure I-1. Laboratory Organization August 1967

Source: "Summary, SAC Briefing, August 15, 1969," W-9-481 (U) (August 25, 1969), p. 8, A99-019, 215-17.

CRE/PORDE

LA-14066-H

JNCLASSIFIEI

UNCLASSIFIED

Agnew's ties to the Military before and during his service as Laboratory Director included not only his assignment with NATO, but in addition many other assignments. He served as a member of the U.S. Air Force Scientific Advisory Board from 1957-1968. He was a member in 1960 of the Von Karman Study Group and also served as a consultant for the Joint Committee on Atomic Energy. In 1961, he became a member of the USAF Minuteman Planning Committee. From 1964–1970 Agnew served as Chairman of the U.S. Army Scientific Advisory Panel. In 1965, he became chairman of the U.S. Army Combat Developments Command Scientific Advisory Group. He was a member of the President's Scientific Advisory Committee from 1962-1972. He served as a member of the Defense Science Board, 1966-1970. In addition, he served as a member of the NASA Aerospace Safety Advisory Panel during 1968–1974. (In 1971, he received the NASA Public Service Award for his service on the Aerospace Safety Advisory Panel.) From 1974-1982 he served on the General Advisory Committee, U.S. Arms Control and Disarmament Agency. In 1966, Agnew received the Ernest Orlando Lawrence Award for his contributions to the development of nuclear weapons and for his outstanding success in working with the armed services to ensure the maximum safety and effectiveness of atomic weapons systems.²

With his weapon and technical background and his long association with Los Alamos, it was natural that Agnew would be chosen as the new director at Los Alamos. Agnew would serve as director from September 1970 until March 1979.

When Agnew became director, as shown in Figure I-2, in general he maintained the major organizational structure of the Laboratory. This was very important in ensuring a functioning organization with a smooth transition between directors. The position of Technical Associate Director continued. Two new staff positions at the Assistant Director level were created: Assistant Director for Weapons and Assistant Director for Research. The job of Assistant Director for Production had already been discontinued when the incumbent retired earlier in 1970. In late 1970 and early 1971, a new Chemistry and Nuclear Chemistry Division, CNC, was formed and Agnew created the Theoretical Design Division, TD, by combining parts of T-Division and W-Division. In 1972 the position of Technical Associate Director level and the incumbent assumed the new post of Deputy Director. The positions of Assistant Director for Weapons and Assistant Director for Research were raised to the Associate Director level and were given line responsibility for specific divisions. As of September 1, 1972, the GMX- and W-Divisions were abolished and two new divisions known as M (Dynamic Testing) and WX (Weapons Engineering) were created.³

Agnew had a deep commitment to the weapon program and believed that the weapon laboratories played the key role. In a June 15, 1970, briefing before the DMA [Division of Military Application] Agnew stated, "You hear today many statements by the Secretary of Defense and other members of the Defense Department of how in spite of progress made by the Soviets we have been maintaining our deterrent. The reason we are maintaining this deterrent is because of the technology that is being provided by the weapons laboratories of the AEC— Sandia Corporation, Livermore, and Los Alamos—and not because of tremendous strides being

LA-14066-H

SECRETARIA

I-7

UNCLASSIFIED

²Jaques Cattell Press editor, American Men & Women of Science (U) (R. R. Bowker Company, New York and London, 1982), Volume I, pp. 40-41. The Atom, Vol. 15, No. 2 (U) (March 1979). Harold M. Agnew, personal communication (U) (March 2, 2002).

³Alison Kerr, et al., two-volume informal history of the organizational structure of the Los Alamos Laboratory, (U) (no date), located in the Los Alamos archives.



LOS ALAMOS SCIENTIFIC LABORATORY ORGANIZATION



November 1970

Figure I-2. Laboratory Organization November 1970 Source: "Los Alamos Scientific Laboratory Organization," (U) (November 1970), 1 p., A88-011, 2-30.

SECREMAN /

UNCLASSIFIED

I-8

made by the Defense Department. It we weren't providing the technology that allows these high yields in smaller packages to be made we wouldn't be keeping up with the Soviets because the number of Minuteman we have is the same and the number of submarines is the same. It is the warhead technology that enables this country to keep up its deterrent, and that is only because of the technological base which the weapons laboratories supported by the Committee and the Commission provide the country."⁴

As part of his new job as director at Los Alamos, Agnew would continue and intensify his campaign for Los Alamos to receive the Phase 3 assignment for the W76. He was successful in this effort.

C. Weapon Group Designations/Responsibilities, Support and Basic Research Groups, and Committee Functions

As noted in the organization charts of Figures I-1 and I-2, the work at Los Alamos took place along the lines of various disciplines. A set of Divisions, each concerned with a particular interest was set up. In each Division there were groups where the work was again more narrowly specified. However, to produce a specific weapon, members from all the different divisions came together as needed. The following sections will attempt to explain how the Laboratory functioned in terms of organization.

1. Weapon Groups: 1966–September 1972

a. Weapon Design

Very early in the history of the various groups in the Laboratory (1948–December 1970), W-4 was designated as the small weapons theoretical design group responsible for the design of single-stage devices and primaries. However, in January 1971, a division known as TD-Division, responsible for the theoretical design of nuclear weapons, was created. Group members in W-4 then became group members in what was designated TD-4.

Until the formation of TD- and C-Divisions, members of T-Division were responsible for computing, code development, theoretical problems in mathematics, and some aspects of weapon design. For a number of years, until 1973, Carson Mark was the Division Leader. Another important member of the division office at that time was secondary designer Robert Thorn. Until Thorn became TD division leader in 1971, he also headed T-2. (A "new" T-2 group called the Nuclear Data group was then formed in April 1971.) Formed in September 1959, T-3 was the hydrodynamics group. Until it became TD-1 in January 1971, T-4 (which had become a group in May 1970) was also a weapons group in T-Division. (Beginning in October 1971, the "new" T-4 became the group concerned with equation of state and opacity.) The group T-5 members were concerned with numerical analysis; in January 1971, this group became TD-5. From July 1963 until January 1971, T-6 was the fission weapons design group. [Author's note: This group under Dave Woods was apparently a backup design group for the other design groups. By having multigroups, it was possible to see if the design teams agreed.] Group T-7 was the computer research and development group. In April 1968, its members joined C-Division. Members of T-8 were concerned with applied mathematics (mathematical methods). Their well-known group

LA-14066-H

SECRETARIA UNCLASSIFIED

⁴"Remarks by H. M. Agnew Concerning Need for Testing," (June 15, 1970 Briefing), DIR-2244 (SRD) (October 9, 1970), pp. 7.6–7.7, A99-019, 269-1.

leader was Stanislaw Ulam. In the January 1971 reorganization, members of this group were split into TD-5 and TD-6. The weapon effects group was T-12—until the group was dissolved in July 1968. Once TD-Division and C-Division were formed, the interests of members of T-Division were those required to provide theoretical and analytical support to most of the major programs of the Laboratory.

C-Division was formed in April 1968 from parts of T-Division and Data Processing. This division was called the Computing Sciences and Services Division. Thus the name indicated the type of work for which the division member's were responsible. Nicholas Metropolis served as an advisor from April 1968 to March 1974. In 1972, the members of C-Division were responsible for maintaining and operating one IBM 7094, two CDC 7600s, three CDC 6600s, two IBM 1401s, and one IBM 1360. In addition, the members' interests included research in statistical theory and development of methodology, consultation, assistance in numerical procedures and techniques for problem solving, numerical analysis, and applied mathematics and programming.

On January 1, 1971, a new division was formed that included the "old" T-2, T-4, T-5, T-8, and W-4 groups. This division was called TD, or the Division of Theoretical Design. Members of the division office included several members of the Los Alamos weapon design team. Robert Thorn was the division leader with Harry Hoyt the alternate division leader. TD-1 was called the Thermonuclear Weapons Physics group. TD-2 was the Thermonuclear Weapons Design group. It was this group that was chiefly concerned with the design of secondaries. TD-3 was Weapons Outputs. Another very important group was TD-4, Small Weapons Design. This group was responsible for the design of single-stage weapons and the primary in multistage weapons. TD-5 was called Codes Development. TD-6 was the Monte Carlo group. Concerning TD activities as of June 1972, it has been reported, "This Division is responsible for the theoretical design of nuclear weapons. The Division is responsible for work on the physical principles of nuclear weaponry, research and development on new concepts, and output effectiveness studies of various classes of weapons. An important portion of the effort is directed toward design and interpretation of nuclear weapon tests. ... The personnel of the Division perform calculations based on fundamental theory but use as a database experimental data from other groups of the Laboratory."

Most of the work in T, C, and TD took place in the main technical area, called TA-3.

b. HE Production and Development and In-House Field Tests

A division called GMX was formed in 1948. This division, divided into a large number of groups, was responsible for explosives and "their interaction with metal." Duncan MacDougall was division leader from August 1948 until September 1970 when MacDougall became Assistant Director for Weapons. [Author's note: It should also be noted that another strong figure in the early weapon program was Max F. Roy. He served in the director's office as Assistant Director for Production from August 1948 until his retirement in June 1970. There is a story that circulates in the laboratory that Max Roy wanted to contract work out, but MacDougall wanted the work done in-house.] Eugene H. Eyster served as alternate GMX division leader between 1954 and 1970 when at that time he became division leader.

GMX-1 was the nondestructive testing group. As will be noted, this group became, in September 1972, M-1. Their main work site was TA-8 (Technical Area 8). This site is known as

SEC BEAUSIA

LA-14066-H

UNCLASSIFIED

I-10

GT site in honor of Gerold Tenney. X-ray techniques were important in the diagnostics conducted by this group. The group also had test facilities at TA-40, also known as DF Site.

Group GMX-2 was the explosives research and development group. It would become WX-2. Included in the GMX-2 work was the development of new types of explosives. Nemo development was also successfully accomplished. The group members worked at the site known as TA-9 or Anchor Ranch. The group also had test facilities at TA-14, also called Q Site.

GMX-3 was the large high-explosives and implosion-systems group that would become WX-3. This was an important group that was responsible for much of the work relevant to high explosives. The large site known as S-Site, TA-16, was the site at which work on explosive manufacture, machining, and testing took place. There was also an HE burning ground. (TA-16 included several sites that had previously, during the Project Y period, had specific names.) The group also had test facilities (including a drop tower) at TA-11, or K-Site.

GMX-4 was the pin techniques group and as such its members were responsible for the pin shots conducted at TA-15 (also known as R Site). Eric L. Peterson was group leader from 1948 until 1971. This group would become M-4.

GMX-11 was the Phermex group. As the group name implies, the members of this group used the Phermex facility to provide important diagnostics on weapon behavior. They too used the TA-15 site. Douglas Venable was group leader from November 1963 until September 1972. Under reorganization in September 1972, the group become M-2.

GMX-6 was the group concerned with optical techniques. It would be this group that would in general, as part of the weapons program, do case diagnostic and related shots. Their test facilities were at TA-39, known as Ancho Canyon Site.

GMX-7 was the group responsible for detonators, firing, and cables. This group would become WX-7. The main area for operation of this group was TA-22, known as TD (Trap Door) Site. Test facilities were also located at TA-40, DF Site.

GMX-8 was the explosives phenomena group. In the 1972 reorganization, this group became M-3. Their test area was at TA-36, known as Kappa Site. The specific areas at this site had names such as Eenie, Meenie, Minie, and Lower Slobbovia.

GMX-9 was the photography group, known as the fast cameras in optics. Their group leader was Berlyn Brixner. Their laboratories were at TA-8.

GMX-10 was called the statistical mechanism and detonation theory group. This group was dissolved as part of the 1971 reorganization.

The GMX field-test groups, and later the equivalent M field-test groups, gave their field-test shots numbers. Thus, as will be noted in the following chapters, each shot record is identified by a specific shot number.

c. NTS Test

The division responsible for the preparation and completion of tests at NTS, including certain diagnostics, was J-Division. This work included ensuring that all the tests were conducted safely and, for the underground tests, the use of proper stemming techniques to ensure containment. The division members also were responsible for the construction and maintenance of field-test facilities. Thus, the work included mechanical design engineering, structural analysis, vacuum technology, and underground phenomenology. Several members were concerned with nuclear weapon effects. William Ogle was division leader from 1965 until October 1972 at which time Charles I. Browne became division leader.

LA-14066-H

SECRET/BU

UNCLASSIFIED

SECRET/RD UNCLASSIFIED

J-1 was concerned with personnel and administration. Group J-3 was responsible for plans, operations, and administration, NTS. Beginning in 1965 the group was located in Nevada under the leadership of Robert Beiler (who left the group in 1979). J-6 had the responsibility of engineering and construction, or facility production. Equipment, engineering and specifications, including downhole design were the responsibility of J-7. J-8 was the electrical engineering group, and their responsibilities included overseeing the timing and firing of the test device.

From March 1971 until September 1979, the group J-9 was known as the underground test phenomenology group. The diagnostics based on radiochemistry were performed by members of J-11 from 1951 until January 1971 when the nomenclature of this group became CNC-11 (Nuclear Chemistry). The laboratories and analytical equipment required for the J-11 radiochemistry program were located at TA-48. (Calibration of equipment and similar activities were conducted at the reactor at TA-2.) Group J-12 was responsible for neutron measurements. In July 1971, the name of the group was changed to Neutron Measurements—PINEX. During the same time period, members of Group J-14 were responsible for the reaction history diagnostics. (During 1966, J-14 had been formed from personnel from J-10.) As part of their responsibilities, J-15 members were responsible for hydrodynamic yield. In general, members of the J-Division groups were greatly assisted by organizations such as EG&G and REECO that were contractors to the Laboratory.

In 1972, the division was known as J-Division Field Testing. The various weapon-related groups in this division as reported in November 1972 are noted below:

Group	Name		
J-1	Operations		
J-3	Operations NTS		
J-6	Facility Production		
J-7	Downhole Design		
J-8	Timing and Firing-Phenomenology Support		
J-9	Underground Test Phenomenology		
J -12	Neutron Measurements—Pinex		
J-14	Reaction History		
J-15	Diagnostic Design Hydrodynamics		

d. Engineering and Design

W-Division was the designation of the nuclear weapons engineering division. The division members accomplished Phase 3 development for all non-HE components, built prototytpes for NTS shots and performed tests to see how a weapon might behave. In addition, W-Division was the principal point of contact within the laboratory for all nuclear weapons programs. As previously noted, Harold Agnew was the division leader from August 1964 until he became director in the fall of 1970. He was then replaced with Robert G. Shreffler. The division was dissolved in September 1972 when most groups joined WX-Division.

The W-1 group was known as weapons engineering. In September 1972, it became WX-1. For many years, Jacob J. Wechsler headed this group. The group was located in the canyon at

SECRETINI

LA-14066-H

UNCLASSIFIED

SPARANES/

TA-41. Group W-7 was the group concerned with the physical and chemical properties of weapon materials. They too were located in Los Alamos canyon at TA-41. This site had a tunnel in the hillside used for secure storage. The tunnel was known as the Ice House, a carryover from Project-Y days when the storage facility was a former ice house. The group had a varied assignment program that covered such diverse activities as responding to accidents involving nuclear weapons to the study of pit hydriding. This group became WX-5.

W-3 was the group concerned with gun-device engineering. The work of this group will be noted in Chapter V in the discussion of the LASL gun-type weapon programs. This group was located at TA-33.

Group W-8, before it was dissolved in February 1972 and absorbed by P-3, was the group responsible for vulnerability and neutron physics. Group W-10, which became a group in 1970, was designated X-Ray Effects on Weapons. It became WX-6.

As will be noted in Chapter V, W-9 was formed in 1968 to provide an interface between the Laboratory and the Military. This group was called the Department of Defense Liaison group. In effect, the group members had the responsibility of "explaining" laboratory programs to the Military and responding to the large number of requests, such as input for the Phase 2 reports, from the DOD and related departments.

e. Materials

In 1972, CMB-Division personnel were responsible for both basic and applied research and development in the fields of chemistry, metallurgy, and chemical engineering.

f. Radiochemistry

In 1971, the former J-11 group was moved into the CNC-Division. The members of CNC, as well as providing the radchem test yields, were interested in low-temperature physics, the study of transuranium elements, and radioactive half-lives.

2. Weapon Groups: Reorganization and September 1, 1972, Designations

The following changes were made when the two new divisions WX and M were formed on September 1, 1972:⁵

WX-Division		M-Division	
New Designation	Old Designation	New Designation	Old Designation
WX-1	W-1	M-1	GMX-1
WX-2	GMX-2	M-2	GMX-11
WX-3	GMX-3	M-3	GMX-8
WX-4	W-3	M-4	GMX-4
WX-5	W -7	M-5	GMX-9
WX-6	W-10	M-6	GMX-6
WX-7	GMX-7		

⁵"Minutes of the 177th X-Unit Steering Committee Meeting (U), September 21, 1972," WX-7-72-4 (SRD) (October 13, 1972), pp. 2–3, B11, Drawer 53, Folder 1 of 2.

LA-14066-H

SECREMEN UNCLASSIFIED

UNCLASSIFIED staren min

E. H. Eyster was appointed WX-Division Leader with B. L. Moore, M. L. Brooks, and R. W. Drake in the Division Office. For M-Division, W. E. Deal was appointed Division Leader, with Douglas Venable, W. W. Wood, J. J. Erpenbeck, and F. R. Parker in the Division Office.

M-Division was known as the Dynamic Testing Division. Thus, the members of this large division were responsible for all the on-site field tests so necessary in the weapon program. The new M-1 group was known as Nondestructive Testing, and its members continued to be at TA-8. M-2 was the Phermex group. Members of this group were, of course, located at TA-15. M-3 was Detonation Physics (located at Kappa Site). M-4 was Pin Diagnostics and Neutron (located at R Site). M-5 was Optical Engineering and Repair. M-6 was Shock Wave Physics (located at Ancho Canyon).

After it was formed, WX-Division was originally designated the Weapons Engineering Division. The division office included a staff responsible for the overall management of such areas as engineering, plans and budgets, operations, hydrodynamics, testing, weapons systems, advanced development, new technologies, and reimbursable programs. WX-1 was given the name Nuclear Components and Engineering. Again, Wechsler headed this important group at TA-41. WX-2 (TA-9) under Louis C. Smith was called Explosive and Other Materials Development. Staff members would, as pointed out in Chapter IV, play a critical role in the development of new materials for the XW76. With the group headed by Jesse Aragon, Group WX-3 members were concerned with high-explosive implosion systems development. The group continued to operate the facilities at TA-16 and TA-11. The gun group, WX-4 (TA-33) was absorbed into WX-5 in April 1973. A new WX-4 group formed in December 1975 was responsible for design systems. It had formerly been ENG-6.

The material development group formerly W-7 was renamed WX-5. The group continued to work at TA-41. W-10 became WX-6, and the group members continued to be concerned with vulnerability and lethality. Although GMX-7 was renamed WX-7, the group members continued to be concerned with detonators and detonating systems at TA-22. Their work on the XW76 will also be noted in Chapter $IV.^7$

Support and Basic Research 3.

Not included in this list are the various required support groups. These included groups whose members were involved in personnel, payroll, procurement, engineering, component fabrication, health and safety, the technical library, and similarly important functions.

Moreover, there has always been the philosophy at Los Alamos that to have a viable weapons program the laboratory also had to be a first-rate scientific research facility. Thus, there were several groups whose members were interested in basic research in mathematics, physics, biology, materials, and similar scientific disciplines. There was also a great deal of interest in the development of new diagnostic tools, including accelerators.

⁷Alison Kerr et al., two-volume informal history of the organizational structure of the Los Alamos Laboratory, (U) (no date), located in the Los Alamos archives. Applicable Los Alamos phone books (U). "Nuclear Technology and Analysis Report (U)," Field Command Defense Nuclear Agency, Kirtland Air Force Base, New Mexico 87115 report FC/06720008 (SRD) (June 1, 1972), pp. 56-61, B11, Drawer 57, Folder 1 of 1. LA-14066-H I-14

SHORET RAM

UNCLASSIFIED

⁶"Minutes of the 177th X-Unit Steering Committee Meeting (U), September 21, 1972," WX-7-72-4 (SRD) (October 13, 1972), pp. 2-3, B11, Drawer 53, Folder 1 of 2.

4. Committees

As noted by Bradbury, various Committees were set up at Los Alamos. These functioned to provide an information and decision network between the various groups that had been set up along the lines of various disciplines.

As previously reported, Bradbury stated that internally in the Laboratory, the weapon program was steered by a committee chaired by the Laboratory Director and comprised of Assistant Directors and relevant Division Leaders. He also noted that the basic decisions were made by this group, the members of which carried the authority within their respective areas of responsibility to implement them. This committee was known as the WLPC or the Weapons Laboratory Policy Committee.

Bradbury also noted that the more detailed discussions and decisions within the framework of agreed-upon programs was made by the Weapons Working Group, the WWG. This committee was chaired by an Assistant Director of the Laboratory. Matters of detailed design, scheduling of tests, analysis of results, and so on, were dealt with within this group. It is interesting to note that this committee is still in operation today.

Bradbury reported that there were still more specialized groups that covered areas of weapon-related research. These committees were concerned with "specialized" information. For example, there was the Hydrodynamics Committee (HWG), the Weapons Group (TWG), and the Vulnerability Working Group. The Hydrodynamics Committee met to hear various presentations on the weapons programs. The forthcoming field tests or their results were also discussed. The Theoretical Weapons Group met to hear discussions by representatives from the design groups. These presentations focused on codes, the theoretical design of weapons, and the results of tests in terms of the design implications. The Vulnerability Working Group, as its name implies, was concerned with vulnerability issues.

Bradbury noted that because much of the work of the Laboratory had to be coordinated with that of the Sandia Laboratory, cross-Laboratory groups existed, again with membership having authority to implement their decisions. There was, for example, the X-Unit Steering Committee, the TX Committee, and the X-R Committee (Zipper).

Under Agnew, the Weapons Program Review Committee (known as the WPRC) came into existence. The members of this Committee attended the WWG and TWG meetings. During their meetings, they heard overviews of the various weapon programs. The responsibility of the members was to advise the WLPC on shot schedules and on the weapons program in general.

D. Naming Names

1. Operation

As the reader will observe, throughout this report various tests are mentioned. These are reported by operation and event name. The operation name is the first name given; this is followed by the event name.

Early-on, testing was not continuous. In general, a series of tests, or campaigns, were planned that were to extend over a certain period. To designate the specific campaign being planned, the test series was given an operation name. Tests to be included were proposed by Livermore, Los Alamos, and the DOD. In general, these would be the tests that actually took place. Sometimes it was necessary to make substitutions at the last minute, and the proposed test list for the series would change. Presidential approval was required before each test.

LA-14066-H

SECRET/RP

UNCLASSIFIED

Beginning with the Niblick operation, testing went on more or less continuously. However, the same program of deciding on a test list and the use of an operation name was continued. The named operation extended from one fiscal year (FY) to the next. Thus, the Storax operation extended through June 1963. Niblick operation tests continued from July 1963 through June 1964. Whetstone tests continued from July 1964 through June 1965. Whetstone was followed by Flintlock that took place from July 1965 until June 1966. Flintlock was followed by Latchkey, FY1966–FY1967; followed by Crosstie, FY1967–FY1968; followed by Bowline, FY1968–FY1969; followed by Mandrell, FY1969–FY1970; followed by Emery, FY1970–FY1971; followed by Grommet, FY1971–FY1972; and this operation was in turn followed by Toggle, FY1972–FY1973. The reader will notice these names throughout the remaining chapters.

The name of each operation was chosen in Washington.⁸ DMA [Division of Military Application] staff member Ken Adney recalls that while he was at DMA in the 1960s–1970s, he and staff member Irv Williams would propose names for the operation. They tried to think of names that might be related to the particular service of the Military that the person in charge belonged to. Once the list was presented to the person in charge, [such as the Director of Military Application or later the Assistant General Manager for Military Application] this person then selected the operation name from the list.⁹ The Whetstone through Toggle series appear from their names to represent small, but important, items that were used, or had been used by those in the service. The name Niblick was perhaps a reminder that someone liked to play golf.

2. Event

With as many tests as the U.S. conducted, it was a nontrivial task to specify a suitable event name for every test.

After the early test program, a formal procedure for naming names was initiated. In order to make the task more organized, the decision was made to designate a family class of nouns. The family type (along with names representing this family) was submitted to the Atomic Energy Commission (AEC). The AEC in turn would announce which names had been approved. These families of names included San Francisco streets, types of cheeses, games, ñaūtical terms, plants, animals, Indian tribes, and tools.¹⁰

Perhaps one of the best sources of names for the Los Alamos group was to make use of the place names in New Mexico. In 1965, the University of New Mexico press published a small book called *New Mexico Place Names, a Geographical Dictionary*, edited by T. M. Pearce, assisted by Ina Sizer Cassidy and Helen S. Pearce. Most place names found in New Mexico are listed in this publication, and a short paragraph explains where the named location is and how the name originated. This dictionary contains more than 5,000 individual items. With such a dictionary in hand, the Los Alamos weapon groups found it easy to obtain shot names.

(b)(3)

Patricia Nolin Bodin through John C. Hopkins, personal communication (U) (February 11, 2003).

¹⁰John C. Hopkins, personal communication (U) (June 7, 2002).



LA-14066-H

UNCLASSIFIEI

⁸John C. Hopkins, personal communication (U) (January 15, 2003).

To add to the multiplicity of names, the DOD sponsored shots had double names. The DOD management decided to generate two name lists. For each test, a name was chosen from the first list and then from the second list, and the combination was used as the event name. Thus, this procedure created a double name, such as Dido Queen and Distant Mist. To further confuse the naming process, the DOD might give the event one name and the Laboratory supplying the device might employ another name. An example is Stilton/Hushed Echo. United Kingdom (UK) tests were also sometimes in this category employing both a UK and a U.S. name.¹¹

E. Phases in Weapon Development

In the United States weapons program there are various "stages" that a weapon goes through from conception to retirement. These "stages" have been formalized into steps called phases. The phase definitions summarized in a 1970 report are as follows:

"Phase 1—Weapon Conception

Information is gathered and exchanged by the DOD and the AEC to determine if the weapon concept warrants formal study.

Phase 2—Program Study

This is sometimes called 'feasibility phase.' Desired military characteristics are provided the AEC by the DOD and development and procurement requirements are studied. A formal joint 'feasibility' report is prepared recommending either proceeding or dropping the idea.

Phase 3-Development Engineering

This phase authorizes the full-scale development program, assigns responsibilities to various agencies or laboratories, and culminates in a complete engineering release.

Phase 4—Production Engineering

This is the preproduction phase; however, some modification of design may be included herein.

Phase 5—First Production

This phase is marked by delivery of the first production units of weapon assemblies. <u>Phase 6—Quantity Production and Stockpile</u>

In this phase, the AEC/ALO production facilities provide the weapon or system at the level required to meet DOD stockpile requirements. This phase includes quality assurance and surveillance.

Phase 7—Retirement

In this final phase, a program for physical elimination of a weapon from the stockpile is established on a timely basis."

Additional information on the activities that took place during the various phases is available in the cited reference.¹²

LA-14066-H

SECRETIPIC

UNCLASSIFIEI

¹¹John C. Hopkins, personal communication (U) (June 7, 2002).

¹²Betty L. Perkins, "Tracing the Origins of the Modern Primary: 1952–1970 (U)," Los Alamos National Laboratory report LA-13755-H (SRD) (April 2, 2001), pp. IV-3–IV-7.

This Page Intentionally Left Blank

LA-14066-H AV CBETYNE **UNCLASSIFIED**

UNCLASSIFIED SECRET/BD/

CHAPTER II TABLE OF CONTENTS

(b)(3)

LA-14066-H

/SECREMBLY

UNCLASSIFIED

II-1

UNCLASSIFIED SECREA/BO (b)(3) II-105 I. Proposal.....

(b)(3)

H. The Los Alamos Position May 1973 II-111

LIST OF FIGURES

(b)(3)

SPACE MED

UNCLASSIFIED

UNCLASSIFIED (b)(3) SECRET/BD/ LA-14066-H П-3 UNCLASSIFIED

stere was

SEA BEARE

LA-14066-H

UNCLASSIFIED

(b)(3)

ÎI-4
statted/

(b)(3)

LA-14066-H

AFXARE/1/BDL UNCLASSIFIED

(b)(3)

The basic idea came

from Famularo, Juveland, and Cremer; Bernard and Jacoby contributed to the development of the principle in HE-driven systems. It started with a study of gun devices in an attempt to make them lighter and to drive them with lower reactivities." Additional historical information is available in the cited reference.¹⁴

(b)(3)

¹⁴Beverly A. Wellnitz, "Weapons Working Group, Minutes of the 215th Meeting," WWG-215 (SRD) (October 29, 1969). DD. 5-6. A99-019. 92-19.

(b)(3)

ARCRA/RIX

LA-14066-H

II-6

LA-14066-H

ARCHENRE

(b)(3)

II-7

SECRATION UNCLASSIFIED

ALCHER /BD

LA-14066-H

11-8

SECRET MUNCLASSIFIED

LA-14066-H

A CARAMAN

(b)(3)

UNCLASSIFIED



II-10

/sacreat/sab LA-14066-H UNCLASSIFIED

SECORE MED

LA-14066-H

SECRETA/RA

(b)(3)

п-11

And UNCLASSIFIED

(b)(3)

II-12

SEGRET/BO

UNCLASSIFIED

UNCLASSIFIED

(b)(3)

LA-14066-H

AELTBATY/RD(UNCLASSIFIED

П-13-

UNCLASSIFIED

Steloskie 1/BAD

(b)(3)

SHCKET RO

LA-14066-H

UNCLASSIFIED SECRETARD

LA-14066-H

SAARAAAAAA UNCLASSIFIED

(b)(3)

п-15

UNCLASSIFIED

SECREMENTS /

II-16

sacke and

UNCLASSIFIED

CLASSIFIED Å

LA-14066-H

AACHER MO UNCLASSIFIED



UNCLASSIFIED

LA-14066-H

MACHINEL UNCLASSIFIED

(b)(3)

UNCLASSIFIED SECORECTORY

(b)(3)

II-20

SALAR AND UNCLASSIFIED

LA-14066-H

ATTARA TARA (b)(3)

(b)(3)

П-21

UNCLASSIFIED

(b)(3)

II-22

SPARAMED

LA-14066-H

LA-14066-H

SACREMAN UNCLASSIFIED

(b)(3)

拍-23

UNCLASSIFIED SECRETIAN/ (b)(3) 2 UNCLASSIFIED II-24

(b)(3)

LA-14066-H

SACKET BAD

II-25

(b)(3)

Standition

LA-14066-H

UNCLASSIFIED

LA-14066-H

•

;

ALLAANAL UNULANNA

(b)(3)

ĨI-27

UNCLASSIFIED

ARCKETARD

II-28

UNCLASSIFIED

(b)(3)

LA-14066-H

· 1

ÂJ ASSILLED

(b)(3)

Attake Map

LA-14066-H

ASSILLD

II-30

(b)(3)

SEAREMAN

II-31

-LA-14066-H

UNCLASSIFIED

(b)(3)

11-32

State Avaria

UNCLASSIFIED

LA-14066-H

ALCRAAM UNCLASSIFIED

(b)(3)

(b)(3)

ARABAA/AM

UNCLASSIFIED

II-34

(b)(3)

LA-14066-H

SECRETARIA UNCLASSIFIED

UNCLASSIFIED SRCBAMPARD/

(b)(3)

11-36

ALCRA ALASSIFICIO

UNCLASSIFIED

6. Lessons Learned

(b)(3)

LA-14066-H

SPC BEAVER UNCLASSIFIED

11-37

UNCLASSIFIED *S*E

However, there were even problems in getting some of the necessary input data. For example, the meeting minutes of the January 7, 1971, Hydrodynamics Working Group report that Deal [representing GMX at the meeting] had said, "...we apparently do not know how to

(b)(3), (b)(1)

II-38

SPAREWAR/

LA-14066-H

UNCLASSIFIED AFARAA/RAD

LA-14066-H

SAGRAAMAD/ UNCLASSIFIED

(b)(3)

II-39

(b)(3)

ALCEACTION LA-12 LA-14066-H
UNCLASSIFIED

LA-14066-H

(b)(3)

11-41

CLASSI

FURTH

UNCLASSIFIED AAARAAA

(b)(3)

The history of the early Livermore design program is the subject of LA-13755-H (SRD), and for more information the reader is referred to this document.

(b)(3)

The Bradbury memo was followed by the establishment of the Small Systems Group. During their first meeting on August 29, 1957, the group members decided that a delegation from Los Alamos should visit Livermore in September.

(b)(3)

SELERA/ARA UNCLASSIFIED

II-42

(b)(3)

Staff members visited Livermore to discuss Livermore systems. Various reports from Livermore were sent to Los Alamos. (As an example, the director files at Los Alamos contain reports that describe the Livermore program in the late 1958–1965 time period.) The two design groups met at various meetings such as the JOWOG meetings where information on the Laboratory design programs was exchanged. There was the design verification program that took place between the laboratories during the moratorium (see LA-12950-H (SRD)).

(b)(3)

UNCLASSIFIED

/stappetrickits/

LA-14066-H

(b)(3)

a. Military Requirements for Small, Lightweight Warheads As noted previously, in the mid-1960s the Los Alamos design group had begun work on 10-inch diameter or less primaries.

(b)(3)

The reason for this great interest on the part of the design laboratories in the 10-inch and less diameter was the fact that the Military was pushing for small, lightweight systems. By this period, the missile/guidance/nose-cone establishment in the United States had developed their systems to where it appeared that it would be possible to put several warheads on one intercontinental ballistic missile (ICBM), deploy the missile, and have each of the warheads hit a different target. This concept is referred to as use of multiple independent reentry vehicles (MIRV). It was felt at that time that the USSR was also going into these types of systems. Because a warhead is much less costly than a missile, the Military wanted to pack as many warheads as possible into each missile. This desire for as many warheads as possible on one missile pushed the nuclear weapon groups to achieve as small as possible in terms of diameter. Moreover, the Military wanted as long a range as possible for each missile; this requirement pushed the weapon groups to try and design minimum-weight warheads.

A request for multiple-carriage capability for the forthcoming improved Minuteman system was formalized in a January 1963 revision to the Phase 1 study. Three reentry vehicles were to be carried in this system—designated the Mk 12 (L). On February 12, 1964, Phase 3 authorization was given for the Mk 12 (L). Livermore and Sandia Corporation, Livermore, were to receive the assignment (the warhead would carry the designation XW62). In November 1964, the Military Characteristics were amended to provide a warhead "compatible with a MIRV application on the advanced Minuteman missile system."⁹⁵

(b)(3)

On August 31, 1964, in a letter to AEC Chairman, Glenn Seaborg, Harold Brown, Director of Defense Research and Engineering, formally proposed the lightweight warhead program. Later, a paper titled "MIRV on Minuteman

SEGRETIED

LA-14066-H

UNCLASSIFIED

⁹⁵Betty L. Perkins, "Tracing the Origins of the Modern Primary: 1952-1970 (U)," Los Alamos National Laboratory report LA-13755-H (SRD) (April 2, 2001), pp. XII-7–XII-14.

UNCLASSIFIED

and Titan II' and dated March 3, 1965, was provided to the administrations at the weapon-design laboratories.

(b)(3)

ARCRATIBO

UNCLASSIFIED

LA-14066-H

П-45

(b)(3)

∱I-46

ANGRAMBDA

UNCLASSIFIED

UNCLASSIFIED SECRANKY

(b)(3)

c. Los Alamos—Livermore Discussions, March 1965

A Joint Working Group (JOWOG) 21 meeting was held at Livermore on March 16-18, 1965.

(b)(3)

In addition to their attendance at the meeting, the Los Alamos attendees, George White, S. R. Orr, Eldon Pequette, and Robert Osborne, apparently visited privately with Sack. Their classified notes on this meeting were forwarded to Los Alamos¹⁰/₂

(b)(3)

¹⁰⁰С. Т. Brockett to Helen Redman, (U) (March 23, 1965). p. I. <u>А99-019, 273-4.</u> (b)(3) LA-14066-H II-47 IINCLASSIFIED

(b)(3)

By June 1966, the Livermore teams were almost certain that they were going to receive the Phase 3 assignment for the Poseidon Mk 3.

(b)(3)

II-48

SECTEMPTICAL

UNCLASSIFIED

(b)(3)

LA-14066-H

AERTHEMME

II-49

UNCLASSIFIED

(b)(3)

1. Loss of the Poseidon C-3 Warhead to Livermore

In March 1965, a paper titled "USN Missile Force Improvement Summary" was published under the sponsorship of the Institute for Defense Analyses. This paper suggested a design for a small reentry vehicle designated the Mk 100 and reported that eight of the Mark 100s could be included in the Navy's Polaris A-3 system.¹⁰⁷

By April 1965, the Navy decision makers had decided on a new missile to be known as Poseidon. Compared to the Polaris A-3, the Poseidon was to be longer and have a larger diameter, carry a heavier payload, and achieve a greater range. Each missile would carry multiwarheads and would use a space bus to carry and distribute the warheads on target.¹⁰⁸

A December 6, 1965, letter from the Chief of Naval Operations requested AEC participation in the Poseidon conceptual studies. On the cover sheet of this letter, there is a note written by a person in the Los Alamos management to the effect that Los Alamos had requested that the Navy invite Los Alamos to compete on the Poseidon assignment. The note stated, "...we [LASL] should really go after the business."¹⁰⁹

On January 13, 1966, Director of Defense Research and Engineering, John S. Foster, Jr., in a letter to AEC Chairman Glenn T. Seaborg indicated that the Navy was favoring the Mk 100-type, small-reentry vehicle with its multiwarheads for use on the new Poseidon C-3 fleet ballistic missile.¹¹⁰

Representing the various groups at Los Alamos, on April 28–29, 1966, Peaslee, Aragon, Horpedahl, and Hoverson attended a meeting in Washington on the Poseidon C-3 missile system.

(b)(3)

¹⁰⁷"USN Missile Force Improvement Summary (U)," Institute for Defense Analyses Pen-X Paper 59, IDA/HQ 65-3610 (SRD) (March 1965), pp. 9–10.

¹⁰⁸"Weapon Development Status Report (U)," Headquarters Field Command Defense Atomic Support Agency, Sandia Base, Albuquerque, New Mexico report FC/04650121 (SRD) (April 1, 1965), p. 14, A99-019, 160-1.
¹⁰⁹Harry B. Hahn to Director, Division of Military Application, U.S. Atomic Energy Commission (SRD) (December 6, 1965), 2 pp., A99-019, 217-15.

(b)(3)

¹¹⁰John S. Foster, Jr. to Honorable Glenn T. Seaborg (SRD) (Januarv 13, 1966), 2 pp., A99-019, 217-15.

LA-14066-H CRET/RAV NCLASSIFIED

In May 1966, the AEC published a Phase 2 feasibility study for a warhead for the Poseidon C-3 missile system. Proposals from the Los Alamos Laboratory were included in this study.

(b)(3)

Although the Los Alamos group very much wanted the assignment of the Navy's warhead (to be known as the W68), such was not to be the case/

(b)(3)

Bradbury wrote a letter dated June 1, 1966, to Livermore director Michael M. May.

(b)(3)

Bradbury proposed a "pie

split"—Livermore would take the Mk 3 Poseidon warhead and Los Alamos would take the warhead for the Mk 18.113

(b)(3)

(b)(3)

This decision must have been reinforced after hearing from the Peaslee delegation, when they returned from Washington, as to how badly Los Alamos had come off in the bid for the W68,

(b)(3)

On April 4-5, 1966, a group from Los Alamos composed of William Deal, E. L. Peterson, D. M. Mosher, and Gene Eyster from GMX Division and Bill Davis from W-1 visited Livermore,

(b)(3)

LA-14066-H

¹¹³N. E. Bradbury to Dr. Michael M. May (SRD) (June 1, 1966), 2 pp., A99-019, 186-2.

(b)(3) 784/1/B4X

UNCLASSIFIED

(b)(3)

The meeting minutes then state, "T Division and W-4 will make calculations, and GMX will make the usual types of local tests. W-1 will order mock pits and cases from Oak Ridge, and the HE parts (9404) can be made at Pantex. Fifteen sets are being ordered to start with, for pin, optical and Phermex shots."¹¹⁶

(b)(3)

Detailed information on these shots is available in the cited references.

(b)(3)

¹¹⁶Beverly A. Mohr, "Weapons Working Group Minutes of the 160th Meeting," WWG-160 (SRD) (May 11, 1966), p. 10, A99-019, 92-11.

SACRAT/BAD

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

LA-14066-H

2

\$\$F\$7\$F\$1/\$C

(b)(3)

П-53

UNCLASSIFIED

UNCLASSIFIED SFICREWRAD

(b)(3)

II-54

SERGAAN/AD

UNCLASSI

It would appear that sometime in late April or May of 1967, groups from Los Alamos again visited Livermore. In a letter dated June 2, 1967, Bradbury thanked Livermore Director, Michael May, for the help that the Livermore staff had given to staff from GMX and to Jim Frank and George Fogelson from T-Division.¹²⁷ Included in the help given to Fogelson were the transmittal of five Livermore codes, including the tapes and instructions on how to use them.¹²⁸

The WLPC members met on June 13, 1967. Included in their discussions was the Mk 18.

(b)(3)

This effort would include layout drawing, including weight and center-of-gravity calculations and RV synthesis, as well as studies on vulnerability.¹³¹

(b)(3)

¹²⁷N. E. Bradbury to Dr. Michael M. May (U) (June 2, 1967), 1 p., A99-019, 273-4. Jack W. Rosengren to Dr. Jane Hall (U) (June 8, 1967), 2 pp., A99-019, 273-4.

(b)(3)

¹³¹Beverly A. Wellnitz, "Weapons Working Group Minutes of the 179th Meeting," WWG-179 (SRD) (July 5, 1967), pp. 6, 8-9, A99-019, 92-14.

LA-14066-H

SEQUETIVED

(b)(3)

UNCLASSIFIED

SECRETYRA'

(b)(3)

UNCLASSIFIED

The Agnew letter was followed by a letter dated August 17, 1967, to Livermore Director May, from Bradbury.

LA-14066-H

UNCLASSIFIED

(b)(3)

In his reply, dated August 23, 1967, May agreed to Bradbury's request. He urged a joint diagnostic program using the Los Alamos PHERMEX facility.¹³⁷

(b)(3)

(b)(3)

Yet another Livermore meeting, that involved E. C. Dudziak, H. R. Lehman, M. D. Torrey, nd J. K. S. Walter representing GMX and W divisions, took place on September 27, 1967. 'he requests for information made by the Los Alamos delegation were similar to the earlier requests covering such items as calculational procedures, mounting recommendations and standoff restrictions, hydrodynamic data, drawings, etc.

(b)(3)

(b)(3)

Michael M. May to Dr. N. E. Bradbury (SRD) (August 23, 1967), 1 p., A99-019, 273-4.

(b)(3)

П-ў6

UNCLASSIFIED

Appendict and

II-57

UNCLASSIFIED

RETURN

LA-14066-H

(b)(3)

The Hydrodynamic Committee was

SECRETHE

(b)(3)

1. Initial Considerations

١.

(b)(3)

assigned the job of naming the proposed primary.144

(b)(3)

2. Class Name

Towards the end of November 1967, the problem of what primary to use in the Mk 18 was somewhat resolved.

(b)(3)

¹⁴⁴Jane H. Hall to Distribution, Subject: "Memorandum of Understanding – WLPC No. 21," AD-1814 (SRD) (November 13, 1967), p. 1, A99-019, 91-10.

(b)(3)

\$\$,\$78,677,18451/

LA-14066-H

UNCLASSIFIED

NCLASSIFIED SECRETVE

(b)(3)

This means that every detail will have to be worked out in order to design local tests, attachments, etc. He has requested help from Sandia Corporation in this work."

(b)(3)

For example, at the WWG meeting of February 21, 1968, it was reported that LASL had been assigned development responsibilities for the Mk 18. Bradbury commented that LASL had better pursue all versions of the Mk 18. During the meeting various possibilities for the warhead's design were discussed by Peaslee

(b)(3)

a. Design Program

(b)(3)

During the WWG meeting of March 13, 1968, Osborne explained his work up to that point.

(b)(3)

LA-14066-H

REVIEW

II-59

٠7



(b)(3)

On March 15, 1968, Osborne wrote Sack at Livermore

(b)(3)

Osborne indicated that

he would like to visit Sack and bring Martin Torrey and Al McKnight with him to discuss details of two-dimensional hydrodynamic calculations

(b)(3)

SECREARD

LA-14066-H

UNCLASSIFIED

Adamad

LA-14066-H

UNCLASSIFIED

(b)(3)

APARETRA UNCLASSIFIED

(b)(3)

II-62

AFAR FAMA

UNCLASSIFIED

AF CRAFFIC

(b)(3)

LA-14066-H

ARCHART

UNCLASSIFIED

SECRETARIO/

II-64

LA-14066-H

UNCLASSIFIED

(b)(3)

(b)(3)

5

The GMX-4 report on the shot indicates that the isochrones obtained from the data of Shots 2545 and 2563 showed "Irregularities which looked as if they fundamentally were connected with the weapon's design."¹⁷¹

(b)(3)

^{1/1} Weapons Test Section GMX-4 Progress Report, 16 February through 15 March 1969," GMX-4-1124 (SRD) (no date), pp. 5-6, 48-55, A86-016, 47-8.

(b)(3)

LA-14066-H

/\$4,4,4,4,4,4,1,1,1,1,1

ĨI-65

UNCLASSIFIED

(b)(3)

SALAAAAAA

LA-14066-H

UNCLASSIFIED

AAAAAAAA

(b)(3)

LA-14066-H

Stat GRAA/BAN

II-67

UNCLASSIFIED

(b)(3)

LA-14066-H

UNCLASSIFIED

(b)(3)

The usually mild Bradbury even stated, "I do not care

(b)(3)

LA-14066-H

-

/\$#\\\bt\\\bt\

ĪI-69

UNCLASSIFIED

UNCLASSIFIED \$15/C/RGF/C/RCB/

particularly how Messrs. Agnew [W-Division] and Mark [T-Division] establish their joint review procedures except that they had better be doing so promptly as no metal will get cut or anything else happen until I (or the WLPC) receives their joint blessing."183 During the WLPC meeting of August 21, 1969 it was agreed that in order to avoid delays and confusion, the members would review, on a more frequent schedule, the overall test program philosophy, the need for various tests, and the design considerations.¹⁸⁴

(b)(3)

⁵¹⁸³N. E. Bradbury to J. Carson Mark and H. M. Agnew, DIR-2177 (SRD) (May 29, 1969), 1 p., A99-019, 91-10. 184 Jane H. Hall to Distribution, Subject: "Memorandum of Understanding - WLPC No. 49," AD-1960 (SRD) (August 21, 1969), p. 2, A99-019, 91-10.

(b)(3)

ARETRAC

II-70

UNCLASSIFIED

UNCLASSIFIED SECRET/BOI

(b)(3)

LA-14066-H

SECARAL ARIAN

U

II-71 ASSIFIED

ARGREWING/ UNCLASSIFIED

--

(b)(3)

BACKARA/BAD

LA-14066-H



~II-72

(b)(3)

[Author's note: The field groups at Los Alamos used the word insertion to refer to the placement of the device in the rack. The canister was then sealed and downhole emplacement operations started. John Hopkins has commented that insertion was usually done early in the day so that emplacement procedures could start early enough that the test device was at least part way down the hole the same day. Because of security concerns, the device was never left "uphole" overnight.²⁰²]

(b)(3)

Containment had been

UNCLASSI

II-73

FIED

(b)(3)

(Ety Kid

²⁰²John C. Hopkins, personal communication (U) (April 5, 2002).

UNCLASSIFIED SERGRETHER

satisfactory, no above-background levels of radioactivity had heen detected,

(b)(3)

(b)(3)

ARCRATICA/



(b)(3)

LA-14066-H

AACRAAMAD (

II-75

UNCLASSIFIED

UNCLASSIFIED ARGRAAM

(b)(3)

II-76

/sterner Alate


LA-14066-H

SA AREVARD

(b)(3)

II-77

secreations of

ļ

LA-14066-H

ΤD

(b)(3)

A catcher container for prompt radiochemical sampling would be mounted in the rack above the canister and retrieved after the device had been detonated. Dry nitrogen purging would be used to maintain a benign environment. Browne noted that temperature and humidity conditions would be monitored from the time of emplacement until shortly before zero time.²¹⁵

(b)(3)

(b)(3)

SEGRENKA

١.

UNCLASSIFIED

÷

II-79

IED



(b)(3)

. For additional information, the reader is referred to the cited reference.²²³ The relevant GMX progress reports give more detailed information.

(b)(3)

²²³"Program Status Weapons Research and Development, October – December 1969 (U)," DIR-2195 (SRD) (no date), p. 19.

SECRET/RI TNCLAS

(b)(3)

11-80

AFRA ARA

LA-14066-H

UNCLASSIFIED

SACARTMAN

(b)(3)

LA-14066-H

AT ARAMAN UNCLASSIFIED

II-81

statet/pp/

(b)(3)

Diagnostics were to be alpha, time interval, radiochemistry, and on the secondary (1) Pinex

(b)(3)

SAGRATICAL

П-82

UNCLASSIFIED

П-83

SECRETARD

(originally an acronym for Pinhole Neutron Experiment)—using the television time integrated method; (2) TRAX (Time Resolved Asymmetry Experiment)—time and space resolved measurement of 14-MeV neutron production in the boost region; and (3) RTA (Radiation Time of Arrival)—radiation time of arrival where observation was made of x-rays through a transparent hole in the case. It was noted that the rack design was very complicated and that the design and fabrication schedule was tight!

(b)(3)

SHARNATURNO

TIN

UNCLASSIFIED

SECTRATION /

ANAR MAN

LA-14066-H

STEED 411

II-84

,

(b)(3)

a. Purpose

(b)(3)

LA-14066-H

/\$4\$4\$4\$/BAP

II-85

UNCLASSIFIED



Sacherry

LA-14066-H

UNCLASSIFIED

II-86

UNCLASSIFIED AVERATION

LA-14066-H

SAARET KO

(b)(3)

UNCLASSIFIED

SEQRE V/BAV

(b)(3)

e. Case Studies

(b)(3)

Augutetynkia

ASSIFIED 21

UNCLASSIFIED



LA-14066-H

ARGHERRAM

П-89

UNCLASSIFIED

SFL BET / RAD

II-90

SAGAMAN

LA-14066-H



SACREA/RW/

LA-14066-H

AAARAANA

II-91

ASSIFIED

UNCLASSIFIED

SECTRATIONAL

b. Initial Considerations

П-92

\$\$\$\$\$\$\$\$\$\$\$ LA-14066-H ASSIFIED UNIT

LA-14066-H

ARCRAATE

(b)(3)

II-93

UNCLASSIFIED

SALARE WRAN

(b)(3)

II-94





(b)(3)

LA-14066-H

SECRETARD

۲ میں ۲۰۰۰ ۲۰۰۱ میں ۲۰۰۰ ۲۰۰۱ میں ۲۰۰۰ II-95

UNCLASSIFIED

SACKEWKD/

II-96

SECRET/BD



Startan 1840/

(b)(3)

Associate Director for Weapons, Duncan MacDougall, was an unhappy man.

(b)(3)

LA-14066-H

/SALAAA/MAN

II-97

UNCLASSIFIED

SAGRATINI/

1. Considerations behind the Model 2

(b)(3)

ÍI-98

SECRETIME

UNCLASSIFIED

(b)(3)

2. Field Tests of the Model 2

LA-14066-H

SALARA MAD

II-99 UNCLASSIFIED

AF CIRCETARIO

II-100

/ste/c/telee to/type

LA-14066-H

ä

ΥĘ

ASSIFIED

UNCLASSIFIED ARCREAMED

LA-14066-H

/\$A/APAta/BAD

II-101

UNCLASSIFIED

ARAREMBAL

II-102

(b)(3)

•

SACARTIRE!



UNCLASSIFIED



----LA-14066-H

SECRETIEN

<u>I</u>I-103

UNCLASSIFIED

\$4,4,4,4,4,4/

(b)(3)

II-104

/space and

ASSIFIED UNCI

(b)(3)

1. Proposal

(b)(3)

LA-14066-H

ARARAA AN

II-105

UNCLASSIFIED

1

UNCLASSIFIED

SECTREATIONS OF

(b)(3)

II-106

SAC ALE WARM



UNCLASSIFIED

SECRETIND

(b)(3)

a service service of the service

LA-14066-H

\$RCREWRA

II-107

UNCLASSIFIED

SACIREMED

SECRET AND



UNCLASSIFIED



LA-14066-H

.

\$#ACBEAT/BP

П-109

UNCLASSIFIED

UNCLASSIFIED

AAAMMINAN

II-112

SECREAMEN



UNCLASSIFIED

CHAPTER III TABLE OF CONTENTS

(b)(3)

1. Livermore: A Small, Lightweight Secondary	Ш-3			
2. Los Alamos: 1959–1964	. Ш-4 [?]			
B. Los Alamos 1965-1968 Design Program Involving a Small Secondary	III-10			
(b)(3)				
2. Early Mk 18 Program	Ш-13			
C. BMP Program and the Mk 18	Ш-15			
1. Concerns With the Primary in Terms of Secondary Design	III-15,			

(b)(3)

E.	Livermore's Shape-to-Fit Program		
	1. Background	····	<u>Ш-89</u>
		(b)(3)	
F.	Looking Ahead		Ш-96

(b)(3)

LA-14066-H

SECRETERS/ UNCLASSIFIED

.

(b)(3)

III-2

Ar Art Area /

DIVER A GUILTED
(b)(3)

1. Livermore: A Small, Lightweight Secondary

(b)(3)

In the mid-1950s the U.S. Military began to push for smaller, lighter nuclear weapon designs. At Livermore, the secondary design team began work on the design of a secondary suitable for this application.

(b)(3)

ASSIFIED

Ш-3



b. Influence

(b)(3)

2. Los Alamos: 1959–1964

(b)(3)

After Hardtack, a moratorium on testing went into effects

(b)(3)

LA-14066-H

ź

UNCLASSIFIED

(b)(3)

LA-14066-H

/states and UNCLASSIFIED

III-5



This type of lightweight device was the design objective of a UK/U.S. study group called JOWOG-3 (Joint Working Group 3).

(b)(3)

/R. Thorn discussed this design concept at the July 16, 1959, meeting of

the TWG.³

The design was again discussed by R. Pollock during the August 20, 1959, TWG meeting. By now the design was more complete.

(b)(3)

⁵"Minutes of the Third Meeting of TWG." TWG-3 (SRD) (July 30, 1959), p. 4, A99-019. 87-8_____

(b)(3)

Ш-б

UNCLASSIF

LA-14066-H

UNCLASSIFIED

(b)(3)

Ш-7

UNCLADELL ARCKERANT!

AFRETARD UNCLASSIFIED

III-8

UNCLASSIFIED SA (

LA-14066-H

-

ED

Ш-9 🗠

(b)(3)



B. Los Alamos 1965-1968 Design Program Involving a Small Secondary

(b)(3)

LA-14066-H

III-10

(b)(3)

LA-14066-H

UNCLASSIFIED

Ш-11

UNCLASSIFIED

III-12

LA-14066-H

(b)(3)

2. Early Mk 18 Program

a. Options for a Warhead for the Minuteman III Missile

In the mid-1960s the Military began to consider a possible Minuteman III missile that would employ a multiple warhead carriage. The warhead for this application was designated Mk 18.

(b)(3)

In a June 1, 1966, memo, Bradbury proposed to do a "pie split" with Livermore. If Livermore would support Los Alamos receiving the Mk 18 assignment, Los Alamos would not oppose the assignment of the Mk 3, the Poseidon warhead, to Livermore.³⁵

In a TWX dated June 23, 1966, AEC Director of Military Application, Delmar Crowson, informed the Laboratories that in order to have different nuclear design approaches for the Mk 3 and Mk 18, he was going to assign the Mk 3 to Livermore and the Mk 18 to Los Alamos.³⁶

(b)(3) ³⁵N. E. Bradbury to Dr. Michael M. May, DIR-2032 (SRD) (June 1, 1966), 2 pp., A99-019, 186-2. ³⁶USAEC Delmar L. Crowson, Wash., D.C. to RUWPQA/USAEC L. P. Gise, Albuq., N. Mex., et. al. (SRD) (June 23, 1966), 3 pp., A99-019, 182-1.

LA-14066-H

SECREMENT UNCLASSIFIED

Ш-13

UNCLASSIFIED

EARA Ŕ

LA-14066-H

UNCLASSIFIED

III-14

(b)(3)

However, there was never a request for weaponization of such a device and the program was dropped.

(b)(3)

C. BMP Program and the Mk 18

1. Concerns With the Primary in Terms of Secondary Design

(b)(3)

LA-14066-H

UNCLASSIFIED

Ш-15

ASSIFIED

:

LA-14066-H

UNCLASSIFIEI

CURENT

(b)(3)

[Author's note: This proposal of Peaslee's would become the BMP program. The BMP program and the test program that grew out of this are included in the following sections. Each section describes a relevant test.

(b)(3)

LA-14066-H

AN AN AN ANA UNCLASSIFIED

III-17



III-18

AAABAA MAAA

UNCLASSIFIED

(b)(3)

LA-14066-H

AFRARIAN

Ш-19

UNCLASSIFIED

(b)(3)

d. Secondary Design

(b)(3)

e. Execution

(b)(3)

`п-20

!



UNCLASSIFIED

UNCLASSIFIED

LA-14066-H

III-21

(b)(3)

III-22

LA-14066-H ASSIFIED 143

f. Secondary Performance

LA-14066-H

AND REPORT Ш-23 UNCLASSIFIED

(b)(3)

UNCLASSIFIED (b)(3)

III-24

SACAAAAAAA UNCLASSIFIED

(b)(3)

Bradbury's Directive g.

(b)(3)

He again expressed this directive in a May 29, 1969, memo to Mark and Agnew.⁷⁹

(b)(3)

Redesign a.

(b)(3)

⁷⁹N. E. Bradbury to J. Carson Mark and H. M. Agnew, DIR-2177 (SRD) (May 29, 1969), 1 p., A99-019, 91-10.

LA-14066-H

(b)(3) BALALABAD UNCLASSIFIED

Ш-25

(b)(3)

b. Preparations and a Last Minute Design Change

(b)(3)

111-26

SPARATAM

UNCLASSIFIED

LA-14066-H

ATTACKA A AND

(b)(3)

III-27

UNCLASSIFIED ر (b)(3)

AAAAAAAAA

LA-14066-H

LÀ-14066-H

ARCRA/MARY

(b)(3)

111-29

(b)(3)

The necessary information for the required performance of the display scopes, interlocks, and timing scopes as well as the dry-run system was provided by Carlton Young in a December 8 letter to the EG&G group in Nevada.⁹⁵

(b)(3)

The fluor was imaged on the television system with a

(b)(3)

⁹⁵Carlton S. Young to Robert Kost/Al Tarr, EG&G, Inc., P.O. Box 295, Mercury, Nevada 89203 (CRD) (December 8, 1969), 5 pp., A99-019, 258-12.

ІП-30

E A KIDA

UNCLASSIFIED

Newtonian telescope; a beam splitter formed images for two television cameras. Proximity focus image intensifiers excluded gamma-produced light before neutron arrival and debris-produced light thereafter. The instrument package was protected by kinking the path from the device as much as possible and placing baffles along the optical path pipe to prevent neutrons and gamma rays from reaching the television package.⁹⁷ The TV Pinex technique had first been used by the Los Alamos group on the Rickey test fired in 1968. TRAX was a time-resolved asymmetry type experiment. In this type of diagnostic an array of detectors viewed the boost region through collimators to give both a time and energy resolved measurement of the 14-MeV neutron production in the boost region. This technique was first used on Door Mist in 1967.³⁴

(b)(3)

³⁸R. W. Peterson to Doug Venable, WPC-O, Subject: "List of J-Division Diagnostics," J-DOT – JRPO-77-4 (SRD) (August 15, 1977), pp. 1–2, A86-049, 1-6.

LA-14066-H

SAAAK MANA UNCLASSIFIED

Ш-31



c. Test and Results

The insertion and emplacement activities took place as outlined by Browne, and no problems were encountered

(b)(3)

ARCRAMKI

UNCLASSIFIED

UNCLASSIFIED

LA-14066-H

(b)(3)

III-33

(b)(3)

AAAAAAAAAA

LA-14066-H

UNCLASSIFIED

III-34



Vander Maat discussed his calculations vs the then available diagnostic data during the January 28, 1970, meeting of the TWG. If he used one type of input in his secondary calculations, he obtained a number similar to the fission but not the fuel yield; if he used a different input, he obtained a number similar to the fuel but not the fission yield. Thus, no matter what his input parameters were, he could not get an output consistent with the observed data. The meeting minutes state his conclusions, "The fuel results indicate that either we do not know how to calculate fusion and fission or we do not know how to interpret the radiochemical results."¹⁰⁴ This problem, as will be noted in the following paragraphs, would continue to bother the design team.

(b)(3)

(b)(3)

LA-14066-H



Ш-35

UNCLASSIFIED

ł

¹⁰⁴ "Theoretical Weapons Group, Minutes of the 135th Meeting," TWG-141 (SRD) (January 28, 1970), p. 4, 18, A99-019, 87-15_

UNCLASSIFIED SACRETICAN

a. Specifications

(b)(3)

SANCARA/A/RACK

LA-14066-H

UNCLASSIFIED

III-36

(b)(3)

\$44AVAPAA/BAA

III-37

UNCLASSIFIED

LA-14066-H



b. Implementation and Instrumentation

(b)(3)

Ш-38

AAAAAAAAAAAAAAAA

UNCLASSIFIED
(b)(3)

LA-14066-H

AN ARE FORM

III-39

UNCLASSIFIED

UNCLASSIFIED

c. Test and Results

(b)(3)

III-40

ARABATARA

la-14066-h UNCLASSIFIED

UNCLASSIFIED

LA-14066-H

ţ

(b)(3)

(b)(3)

A A CARACTA ARABA

III-42

UNCLASSIFIED

(b)(3)

AAAAAAA

LA-14066-H

III-43

(b)(3)

a. Proposal

(b)(3)

I**∏-4**4

AAAAAA/BABA

UNCLASSIFIED

(b)(3)

LA-14066-H

AAAAAAAAAAAA

Ⅲ-45

UNCLASSIFIED

(b)(3)

III-46

AAAAAAAAAAA

LA-14066-H

b. Implementation

(b)(3)

(b)(3)

LA-14066-H

SAABAA/BAH

Ш-47

(b)(3)

́**Ⅲ-4**8

د در

state part

ASSILLO UNCL

c. Results

(b)(3)

LA-14066-H

Ш-49

UNCLASSIFIED

AND UNCLASSIFIED

(b)(3)

SAAMAA MAN

LA-14066-H

Ш-50

-d. Informing the Military

LA-14066-H

State A watch

(b)(3)

Ш-51

UNCLASSIFIED

UNCLASSIFIED

e. Looking Ahead to a Navy Decision

(b)(3)

a. Proposal

(b)(3)

III-52

/states and the states of the

LA-14066-H

LA-14066-H

/SACKAP/KAY

(b)(3)

III-53



III-54





H. W. Kruse, a staff member in J-14, outlined the reaction history detector requirements in a January 19, 1970, memo. Included were specifications not only for the number and types of detectors, but also for their distances from the device, the layout of the collimators, cable requirements, and the types of compensators to be used. This memo was sent to representatives of Edgerton, Germeshausen, and Grier, Inc. (EG&G) and J-7 and formed the basis for the required rack and detector layout designs.¹⁸⁰ These instructions were followed by a January 26 letter from Kruse to Sandoval at EG&G providing information on the necessary data recording scopes, trigger signals, timing requirements, prebase markers, interlock signals, and the dry-run system.¹⁸¹

(b)(3)

¹⁸¹H. W. Kruse to Lee Sandoval (CRD) (January 26, 1970), 7 pp., A99-019, 296-3.

LA-14066-H

NCLASSIFIED

(b)(3)

Ш-55



/standar rank

LA-14066-H

UNCLASSIFIED

III-56

UNCLASSIFIED

; **c.**

Design

(b)(3)

LA-14066-H

5

SAARAAAAAA

Ш-57

UNCLASSIFIED

UNCLASSIFIED

d. Test

(b)(3)

e. Results

(b)(3)

/steal Action and the state of the state of

111-58

UNCLASSIFIED

LA-14066-H

ALAMAAAAA

(b)(3)

III-59

UNCLASSIFIED

MARETARA

a. Initial Work 1970

(b)(3)

(b)(3)

Ш-60



LA-14066-H

LA-14066-H

A CRAFT AND AND A CRAFT

(b)(3)

Ш-61

(b)(3)

Objected to any of the listed tests, it was reported that the listed shots were therefore tentatively approved ²¹¹

(b)(3)

<u>1971</u>

(b)(3)

²¹¹D. P. MacDougall to Distribution, Subject: "Memorandum of Understanding – WLPC No. 61," ADW-7 (SRD) (October 9, 1970), p. 3, A99-019.91-11.

(b)(3)

SA ABER BAL

LA-14066-H

UNCLASSIFIED

 I_{1}^{i}

111-62

LA-14066-H

Steft Pression

UNCLASSIFIED

III-63

UNCLASSIFIED

CSIFTED.

III-64

日日の日日

LA-14066-H

<u>1972</u>

(b)(3)

(b)(3)

\$PACAAF MAAA

Ш-65

71

UNCLASSIFIED

(b)(3)

The attendees at the July 19, 1972, WWG meeting were informed of the proposed FY73 test schedule.

(b)(3)

SEALBANNAN

III-66

LA-14066-H

UNCLASSIFIED ACARA ANARAN

During 1972, changes to the 1971 revised preshot report (TD-2: 71-38) were proposed in TD-2: 72-113.

<u>1973</u>

(b)(3)

LA-14066-H

Ш-67

UNCLASSIFIED Standa/Ann

Ш-68

LA-14066 UNCLASSIFIED

(b)(3)

b. Preparations and Test

(b)(3)

Diagnostics

were to be alpha, time-interval, television-time-integrated Pinex, and radiochemistry.

(b)(3) /The WX-3 and the J-Division work programs were -

(b)(3)

^{229,} Toggle," JOHO-73-8 (SRD) (January 17, 1973), pp. 1-2, B11, Drawer 109, Folder 2 of 3.

(b)(3)

SECRET/RA UNCLASSIFIED

UNCLASSIFIED REAR

Indeed, the J-Division schedule slipped, but not by a full month

(b)(3)

MacDougall's Response_ c.

(b)(3)

Pinex pictures had been obtained. The event had been completely contained; detectors, located 557 ft below the surface, had shown no above-background radiation or excessive pressures in their readings.²³³

(b)(3)

²³⁵Univ. of Calif., Los Alamos Sci. Lab., D. P. MacDougall, Los Alamos, N.M. to BY3/Captain Wayne L. Beech,

(b)(3)

SACABENTRUM UNCLASSIFIED

LA-14066-H

ť

III-70

UNCLASSIFIED

Ш-71

UNCLASSIFIED

ANG BUDGED

LA-14066-H

. (b)(3)



d. Results and Implications

(b)(3)

Good data had been obtained from zero time until $4 \mu s$ when a short noise pattern had occurred. The data were again good from 4 to 8 μs after

(b)(3)

, S.E.C. P.E.V. P.W.

LA-14066-H

UNCLASSIFIED

III-72

which time no further data were collected. Additional information and pictures of the Pinex images are available in the cited reference.²³⁷]

To conclude the data presentation at the April 18 WWG meeting, radiochemist Bryant reported the radchem results,

(b)(3)

LA-14066-H

AT A BELLINGUN UNCLASSIFIED

III-73

2

²³⁷Leslie M. Redman, "LASL Weapons Quarterly (U), for the Period Ending March 31, 1973," Los Alamos Scientific Laboratory report LA-5330-PR (SRD) (June 1973), pp. 60-62.

(b)(3)

ĬII-74

UNCLASSIFIED
(b)(3)

LA-14066-H

sa/ARA/MAA

UNCLASSIFIED

III-75



III-76

AN AMAAAAA UNCLASSIFIED

LA-14066-H

SAAAAAAAA UNCLASSIFIED

(b)(3)

Ш-77

(b)(3)

g. Further Discussion

(b)(3)

a. Proposal

(b)(3)

LA-14066-H

UNCLASSIFIED

III-78



_h__ Deferred

(b)(3)

It was reported that this test had been deferred on the basis that the information was not needed soon.

a. Moving Forward

(b)(3)

LA-14066-H

ARAMANA

UNCLASSIFIED

Ш-79

(b)(3)

b. Los Alamos Announcements

\$ AVARAMINA LA-14066-H UNCLASSIFIED

(b)(3)

in a letter dated October 31, 1972, to Navy/Lockheed personnel, R. B. Olwin updated these groups as to the latest plans at Los Alamos for tests of weapon designs that might be suitable for use in the Navy's Mark 400 warhead

(b)(3)

c. Secondary Design Considerations from the Technical Viewpoint

(b)(3)

LA-14066-H

AFFARMANARAD

UNCLASSIFIED

Ш-81



UNCLASSIFIED

LA-14066-H

<u>111-82</u>

(b)(3)

UNCLASSIFIED

Ш-83



e. <u>Preparations</u>

(b)(3)

LA-14066-H

UNCLASSIFIED

III-84

(b)(3)

III-85

MARCLASSIFIED

<u>f.</u> Implementation

(b)(3)

g. MacDougall Reports

(b)(3)

MacDougall sent the latest revised data for the major diagnostics to Camm on November 20, 1973.

(b)(3)

SACRAA/KA

111-86

UNCLASSIFIED

(b)(3)

Results h.

Y !

(b)(3)

Preliminary results were again discussed during the August 15, 1973, WWG meeting. These included a presentation by Kruse of the J-14 data. The TRAX experiment had yielded gamma and neutron signals with good time resolution. Kruse presented graphs showing detector currents for gammas and neutrons. Berzins reported on the J-12 Pinex results. He included a computer . , analysis of the Pinex data.

STATISTIC UNCLASSIFIED

Ш-87

(b)(3)

h. Results

١.

· •

1

(b)(3)

Preliminary results were again discussed during the August 15, 1973, WWG meeting. These included a presentation by Kruse of the J-14 data. The TRAX experiment had yielded gamma and neutron signals with good time resolution. Kruse presented graphs showing detector currents for gammas and neutrons. Berzins reported on the J-12 Pinex results. He included a computer analysis of the Pinex data

(b)(3)

LA-14066-H

SALARAARA

Ш-87

UBULASSINED

(b)(3)

As discussed in Chapter II, the postshot report on the primary was released on March 14, 1974.

i. Considerations

(b)(3)

STATESTA

UNCLASSIFIED

LA-14066-H

ІП-88

(b)(3)

1. Background

a. A Warhead for the Trident Missile

In the late 1960s and early 1970s time period, the Navy began to plan for an advanced strategic missile system to be carried on their new submarine called the Trident. The Livermore group had traditionally been given the assignment for the warheads for the Navy's strategic systems. Livermore had supplied the W47 for use on the Navy's Polaris A1 and A2 missiles. Livermore had supplied the W58 for use on the Navy's Polaris A3 missile. Livermore had supplied the Navy's Poseidon missile. The Livermore, Lockheed, and Navy connection was a close one.

(b)(3)

LA-14066-H

UNULADAL L

Ш-89

When the Navy planners began to consider the development of a new, longer-range missile, the Livermore design team began to consider designs for a suitable warhead that would fit into the reentry bodies that this missile would carry. Based on their assignments in the past, the Livermore group probably had every expectation of achieving the assignment for the new warhead.

(b)(3)

c. Los Alamos Viewpoint

(b)(3)

UNCLASSIFIED

LA-14066-H

Ш-90



a. Objective

(b)(3)

[Author's note: The Mk 500 was the Navy's proposed RB that could maneuver to evade destruction.]

(b)(3)

SILIED

III-91

(b)(3)

LA-14066-H ir mi

III-92

(b)(3)

(h)(3)

d. Test Implementation

e. Results

LA-14066-H

ANDARAMARA UNCLASSIFIED

Ш-93

UNCLASSIFIED

III-94

LA-14066-H

ALAURE FRANK

UNCLASSIFIED

f. Significance

(b)(3)

SEAMENTICAL UNCLASSIFIED

111-95



F. Looking Ahead

When the Los Alamos weapons groups received, in the late spring of 1973, the Phase 3 award for the XW76, the secondary-design team had a series of previous tests that they could use as a design base. Although their 2-D code development was somewhat lacking, there were 1-D codes that had been developed, and their use compared with experimental data from relevant tests.

(b)(3)

AFVANA

LA-14066-H

UNCLASSIFIED

III-96



CHAPTER IV TABLE OF CONTENTS

5, ENOMALERING, AND YULNERADILIT I	1V
·····	IV-3
(b)(3)	
	- N/
(b)(3)	
	-TV-3
atibility	IV-
•	R/
(h)(3)	
ans to Provide Minimum Helium in the Boost Gas	IV-
	IV-(
mputer Codes	IV-
1	IV-
	TV /
•••••••••••••••••••••••••••••••••••••••	
ity Tests	
ity 10868	
	(b)(3) (b)(3) (b)(3) (b)(3) ms to Provide Minimum Helium in the Boost Gas inputer Codes

LIST OF FIGURES

(b)(3)

LIST OF TABLES

LA-14066-H

ARARAA MARA UNCLASSIFIED



This Page Intentionally Left Blank

AAAAAA **/\$**#

IV-2

UNCLASSIFIED

UNCLASSIFIED

CHAPTER IV. MATERIALS, ENGINEERING, AND VULNERABILITY

A. Materials

1. Considerations

(b)(3)

It appears that the main goal for the development of this alloy at that time was for use in improving corrosion resistance in fuel cores in the early nuclear reactor program.¹

(b)(3)

¹Donald J. Sandstrom, personal communication (U) (January 16, 2002).

LA-14066-H

SECRATIBOL

ICLASSIFIED

UNCLASSIFIED AT A ANA A MED

temperature environmental tests will be started as soon as the metal parts become available, probably in January."⁷

(b)(3)

'Group GMX-3 Progress Report, November 16 through December 15, 1963," GMX-3-3505 (SRD) (no date), p. 4, A86-016, 28-14.

IV-4

REFER

LA-14066-H

These

ţ

UNCLASSIFIED

(b)(3)

LA-14066-H

\$10 P F F GAAD ICLASSIFIED



LA-14066-H

UNCLASSIFIED

SALARA AVRAL

IV-6



(Additional information on the complex TX61 program is included in Chapter V.)

(b)(3)

UNCLASSIFIED

IV-7



This work is reported in the 1967-1968 GMX-3 progress reports.

(b)(3)

On March 13, 1972, Charles L. Peterson from W-7 wrote M. J. Davis at Sandia Corporation. Peterson informed him that help was needed in examining the potential materials problem that might occur in the W66 and W69

(b)(3)

12-8

UNCLASSIFIED

UNCLASSIFIED

LA-14066-H

\$\$\$

(b)(3)

UNCLASSIFIED

(b)(3)

UNCLASSIFIED LA-14066-H

LASSIFIED

Ę

(b)(3)

e.

Additional Development Programs at Los Alamos

(b)(3)

LA-14066-H

/stere been to be the state

UNCLASSIFIED

(b)(3)

A APAPADA UNCLASSIFIED

IV-12

(b)(3)

LA-14066-H

UNCLASSIFIED


TV-14

BEREFERRE

LA-14066-H

UNCLASSIFIED



f. Preparation

À

(b)(3)

g. Behavior

(b)(3)

LA-14066-H

ABARTARD /W

 $\langle \cdot \rangle$

UNCLASSIFIED

(b)(3)

UNCLASSIFIED LA-14066-H



/The following paragraphs will give the reader some history on the early

program.

(b)(3)

The GMX-2 members began both a literature search and an experimental program designed to address the Agnew question.

(b)(3)

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

IV-18





UNCLASSIFIED ATRAA A

(b)(3)

This sample was being analyzed to determine whether the manufacturer's normal procedures would yield a sufficiently pure material.⁹¹ The Quarterly Status report from the Laboratory for the period ending September 1971 summarizes the previous work on the high-hydrogen materials. The report indicates that various materials containing a high weight-percent of hydrogen in combination with low-Z atoms had been, in the previous two years, exposed to gamma rays from a ⁶⁰Co source)

(b)(3)

The amount of H₂- or H-containing gaseous products evolved per Mrad of energy absorbed had been measured.

(b)(3)

In terms of the most stable organic compounds studied, the report indicates that LiH was only exceeded in terms of radiation stability by NH4Cl. But the report also notes, "Because of interest in the hydrogen-rich materials for other weapon applications, properties of the compacted materials such as tensile strength, dimensional stability, pressing characteristics, and compatibility with other components are being investigated."92

(b)(3)

(b)(3) p. 22, A86-016, 21-4.

⁹¹"GMX-2 Monthly Progress Report, August 11 to September 10, 1971," GMX-2-MR-71-9 (SRD) (no date), p. 28, A86-016, 21-3.

92"Quarterly Status Report on Weapons Research and Development (U), for the Period Ending September 30, 1971," Los Alamos Scientific Laboratory report LA-4820-PR (SRD) (no date), p. 39.

IV-20[™]

NCLASSIFIED

(b)(3)

Ventron-Alfa had received the Los Alamos order, including the material specifications; their bid had been sent to LASL on April 7. A group from Los Alamos had visited the plant.

(b)(3)

Personnel at Y-12 would then recrystallize the Allied Signal production and press the material into billets. The material was then machined to the required specifications for each piece.]

The May 11 to June 10, 1972, GMX-2 monthly report indicates that the Ventron Corporation planned to have the hundred pound order prepared by June 9; they were going to ship a 100-gram sample to Los Alamos for testing. At Los Alamos, samples (using the Livermore supplied material) had been prepared and given to DuBois and Baytos for the thermal conductivity determinations. The results of these determinations were reported.⁹⁶

(b)(3)

It was reported in the June 11 to July 10, 1972, GMX-2 report that the Ventron workers had to rework the "entire lot." However, following suggestions made by the LASL staff to slurry the material in dry ethanol, the resulting production appeared to meet the required specifications. ' A 100-gram sample of this production was then shipped from Ventron to Los Alamos. Initial analysis indicated that the sample was acceptable; however, the spectrochemical analyses had yet to be completed.⁹⁷

(b)(3)

²⁰GMX-2 Monthly Progress Report, March 11, 1972 to April 10, 1972," GMX-2-MR-72-4 (SRD) (no date), p. 17, A86-016, 21-10.

⁹⁶"GMX-2 Monthly Progress Report, May 11, 1972 to June 10, 1972," GMX-2-MR-72-6 (SRD) (no date), pp. 26–27, A86-016, 21-12.

⁹⁷"GMX-2 Monthly Progress Report, June 11 to July 10, 1972," GMX-2-MR-72-7 (SRD) (no date), pp. 15–16, <u>A86-016, 21-13</u>.

LA-14066-H

SAABAABAL EINELASSIFIED



It was hoped that these inclusions could be identified and a method developed for preventing their inclusion into the material.¹⁰¹ (Larson has remarked that a follow-up visit to the Ventron plant indicated that material from an overhead walkway was falling into the processing vats. This problem was apparently corrected.)

(b)(3)

4. PBX 9501

(b)(3)

(b)(3)

¹⁰¹"WX-2 Monthly Progress Report, December 11, 1972 to January 10, 1973," WX-2-MR-73-1 (SRD) (no date), pp. 13–14, A86-016, 271-4.

(b)(3)

RETURA

UNCLASSIFIED

LA-14066-H

LA-14066-H

AAAAAAA UNCLASSIFIED

IV-23

(b)(3)

UNCLASSIFIED

(b)(3)

IV-24

UNCLASSIFIED

LA-14066-H

SAGRA/MAD UNCLASSIFIED

(b)(3)

IV-26

1

ARACARICANRA

(b)(3)

UNCLASSIFIED

(b)(3)

During the January 4, 1967, meeting of the WWG, R. G. Shreffler, Alternate Division Leader of W-Division, reviewed the vulnerability program. He stated that he felt it essential that the Laboratory continue its efforts in vulnerability studies

(b)(3)

¹²⁷"Group WX-3 Progress Report (U), March 16 through April 15, 1973," WX-3-75-3 (SRD) (no date), pp. 9–10, ______A86-016, 275-3.

1

(b)(3)

LA-14066-H

SECRETINE (UNCLASSIFIED

IV-27

ŝ

CLASSIFIED (b)(3) AFRACEARA LA-14066-H IV-28 NCLASSIFIED

ŢŢ

LASSIFIED

LA-14066-H

SACAPETIEN UNCLASSIFIED

IV-29

(b)(3)

(b)(3)



LA-14066-H





SAMABLE VIELASSIFIED TA-14066-H

LA-14066-H

SSIFIED

(b)(3)

(b)(3)

7. Pu Alloys

(b)(3)

IV-34

·••••

AACAAAAAA

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

SECRETION UNCLASSIFIED

IV-35

(b)(3)

ĽA-14066-H **ICLASSIFIED**

(b)(3)

d. Continued Efforts: Sample Preparation

(b)(3)

LA-14066-H

ARAAAAAA

UNCLASSIFIED

UNCLASSIFIED

/SECONE LASSIFIED

IV-38



LA-14066-H

AAAAAAAAAAAA UNCLASSIFIED

UNCLASSIFIED

IV-40

ARCARTIN UNCLASSIFIED

(b)(3)

RJEA/BAA UNCLASSIFIED

LA-14066-H

(b)(3)

LA-14066-H CLASSIFIED

(b)(3)

LA-14066-H

AAAAAA LASSIFIED

UNCLASSIFIED

IV-44

AN AN AND AND UNCLÀSSIFIED

LA-14066-H

(b)(3)

(b)(3)

LA-14066-H

FIED



AAAAAA/BAA

LA-14066-H

UNCLASSIFIED

(b)(3)

LA-14066-H

SSIFIED

(b)(3)

AAA Arvaria LA-14066-H ICLASSIFIED

UNCLASSIFIED

UNCLASSIFIED

IV-49
(b)(3)

RAPANT. STOP UNCLASSIFIED

LA-14066-Н

IV-50

(b)(3)

LA-14066-H

UNCLASSIFIED

IV-51

(b)(3)

Arrend LASSIFIED

LA-14066-H

fV-52

Questions

(b)(3)

²⁶⁶Siegfried S. Hecker and Joseph C. Martz, "Plutonium Aging: From Mystery to Enigma," Ageing Studies and Lifetime Extension of Materials, edited by L. G. Mallinson (Kluwer Academic Publishers/Plenum Publishers, New York, New York, 2001), pp. 23–52.

LA-14066-H

/standard for the UNCLASSIFIED



B. Engineering

1. Tools

(b)(3)

In December 1968, the Engineering Department sent C. G. Nottrott, Acting Area Manager of LAAO, a memo that stated, "Attached are the Secret preliminary proposal, ENG-147, and unclassified project directive and criteria drawings for the proposed Numerical Control Machining Facility at TA-21."²⁶⁹

(b)(3)

The attendees decided that the Director would attempt to obtain authority from LAAO and ALOO to order a Numerical Control machine. A 44-week delivery schedule had been estimated. It had originally been planned to place this machine in a modified space at DP-West (TA-21). The Engineering Department representative at this meeting proposed that a Special Fabrication and Assembly Facility be designed and built as part of DP-West

(b)(3)

The CMB-11 monthly report dated June 20, 1969, indicated that a new assembly facility was to be built east of Building 5 at DP site,

(b)(3)

The LASL shops had offered a modified Excello Model 751 mill that could hopefully be modified for use in a glovebox enclosure. (b)(3)

(b)(3)

LA-14066-H

ICLASSIFIED

²⁶⁷D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 79, November 29, 1971," ADW-168 (SRD) (November 30, 1971), p. 1, A99-019, 91-11.

²⁶⁸ TX Committee, Minutes of the 140th Meeting, October 13, 1972," TXC-140 (SRD) (October 13, 1972), pp. 5–6, B11, Drawer 52, Folder 3 of 4.

²⁶⁹Engineering Department to C. G. Nottrott, Acting Area Manager, LAAO, Subject: "Numerical Control Machining Facility, TA-21," ENG-699 (SRD) (December 10, 1968), 1 p., A99-019, 120-30.



It was reported that the existing gauging equipment was being upgraded for use until the new equipment was available

(b)(3)

The CMB-11 report dated July 20, 1969, noted that the shops department personnel had modified and gloveboxed a modified tracer-controlled Excello boring machine. This unit was to be delivered to CMB-11 by August. It was stated, "This machine will provide a new and much needed machining capability for shapes and accuracy. SD will make pot chucks and templates on their tape controlled machines."²⁷²

By the end of August, the Model 751 Excello Tracer Mill had been installed in a glovebox in Building 5 at DP Site. Final testing of the rotary contour inspecting gauge was in progress.

(b)(3)

It appeared that this type of inspection

might be adapted for use in a glovebox."

In November 1969, LASL staff met with the Heald Machine Tool design and engineering staff to firm up the design criteria for the numerical control turning machine that had been ordered by LASL. (Heald had, in September, accepted the order from Los Alamos.²⁷⁵ A two-axis, numerically controlled machine for use in a glovebox enclosure had been specified.) The LASL staff learned that the projected delivery date from Heald had slipped to September 1970.²⁷⁶

In December 1969, the CMB-11 staff reported that for measurement of the thickness of metals (specifically unalloyed Pu and 1 w/o Ga-Pu) they were evaluating the use of an eddycurrent-type gauge manufactured by Laser Electronics and Scientific Corp.²⁷⁷

(b)(3)

The machined parts were within 0.5 mil of the part

specification."278

(b)(3)

²⁷²⁴ Group CMB-11 Monthly Report," CMB-11-9521 (SRD) (July 20, 1969), p. 12, A89-068, 42-3.

(b)(3)

^{27*}"Group CMB-11 Monthly Report," CMB-11-9557 (SRD) (October 20, 1969), p. 5, A89-068, 34-1.7
²⁷⁵"Group CMB-11 Monthly Report," CMB-11-9537 (SRD) (September 20, 1969), p. 12, A89-068, 42-6.
²⁷⁶"Group CMB-11 Monthly Report," CMB-11-9563 (SRD) (December 2, 1969), p. 9, A89-068, 34-3.
²⁷⁷"Group CMB-11 Monthly Report," CMB-11-9568 (SRD) (December 20, 1969), p. 5, A89-068, 34-5.
²⁷⁸"Program Status Weapons Research and Development, January - March 1970 (U)," Los Alamos Scientific Laboratory report DIR-2203 (SRD) (no date), p. 36.

LA-14066-H

BRACHERINA UNCLASSIFIED

ASSIFIED

However, it was realized, as this

operation was in progress, that the group had a need for more sophisticated inspection capabilities

(b)(3)

The CMB-11 monthly report dated August 20, 1970, stated that the assembly and installation of a 3-axis measuring machine manufactured by Brown and Sharpe had been completed. Necessary probes, accessories, and a precision optical rotary table had been ordered. It was reported, "Methods of adapting the machine to inspection of Pu parts (with emphasis on W-Division weapons components) without committing the machine to full glove box enclosure are under study."²⁸¹

The monthly report from CMB-11 dated November 20, 1970, noted that the group had initiated the procurement of two computer-controlled, three-axis measuring gauges. One, to be located in a glovebox, would be used for measuring Pu parts.²⁸²

Personnel from CMB-11 visited the Heald plant at Worchester, Massachusetts, on January 25–28, 1971, in order to perform a factory checkout before the 2-axis numerically controlled turning machine, that had been ordered by Los Alamos, was shipped. The Los Alamos group was very disappointed. They reported that during their visit, none of the five test parts had been cut to the required dimensional specifications. The CMB-11 representatives stated, "As a matter of fact, the thread cutting feature had not been tested and never did operate under automatic control during the inspection tests." The Los Alamos group added, "None of the LASL provided tapes were usable due to insufficient information supplied to us for manual tape preparation." It was clear that the machine was going to require significant modification. Moreover, additional problems were found that had not been covered in the original specifications for the machine.²⁸³

The Laboratory's Quarterly Status report for the period ending March 1972, reported, "Two remote-XYZ-operated Coordinate Measuring Machines purchased from Bendix have been installed by factory representatives. All phases of the factory checkout specifications were repeated after installation and were acceptable." It was noted, however, that the height of the

(b)(3)

UNCLASSIFIED

LA-14066-H

²⁸²"Group CMB-11 Monthly Report," CMB-11-9672 (SRD) (August 20, 1970), p. 10, A89-068, 40-3. ²⁸²"Group CMB-11 Monthly Report," CMB-11-9711 (SRD) (November 20, 1970), p. 6, A89-068, 40-7. ²⁸³"Group CMB-11 Monthly Report," CMB-11-9723 (SRD) (February 20, 1971), pp. 7–8, A89-016, 175-2.

support frame for the rotary table restricted Z-axis motion and thus the rotary tables would require some modification by Bendix personnel. A computer program for the PDP-8 computer used for data processing had been written to gauge hemispheres by rotating them through <u>360°</u> and measuring the radius at any number of locations between the equator and pole.

(b)(3)

-support fixtures have been made." The initial effort had indicated the need for better temperature control of the plutonium parts.²⁸⁵

An April 1972, report noted that at Los Alamos the program to update the machines, gaging, and pit assembling facilities for Pu components was nearing completion "with the incorporation of tracer- and numerical-controlled turning machines, rotary and three-axis gages, and a controlled atmosphere room for making assemblies."²⁸⁶

The WX-3 progress report for August 16–September 15, 1972, reported, "The Heald NC three-axis vertical lathe is now equipped with a precision bracket that can be locked either to hold an air motor in a vertical position or, by virtue of a well designed swivel joint, to hold the motor in a horizontal position

(b)(3)

⁴⁰³"Quarterly Weapon Research and Development Status Report, April 1–June 30, 1972," CMB-1893 (SRD) (July 14, 1972), p. 14, A89-056, 24-9.

²⁸⁶"Plutonium Research Programs, FY 1972," Plutonium Research Coordinating Committee, U.S. Atomic Energy Commission report SMA-470-746 (SRD) (April 1972), p. 18, B11, Drawer 47, Folder 1 of 1.

LA-14066-H

JNULASSIFIED

(b)(3)

(b)(3)

UNCLASSIFIED

LA-14066-H

IV-58

3. Reservoir Designs to Provide Minimum Helium in the Boost Gas

In a March 1969 memo, primary designer R. Canada outlined the problems that were the result of the formation of ³He from the decay of the tritium used in the primary's boost gas.

(b)(3)

The yield of a boosted primary is degraded as tritium is converted

to ³He both by the loss of the source of 14-MeV neutrons and also by the decrease of the preboost multiplication rate caused by the high cross-section for neutron capture which is characteristic of ³He." He went on to add, "In a conventional boosted single-stage device the tritium produced by ³He appears too late in the bomb's explosion to contribute to the yield, and the temperature does not get high enough to produce significant ³He + D fusion."²⁹³

(b)(3)

²⁹³R. Canada to Distribution, Subject: "³He in Weapons," W-4-2518 (SRD) (March 10, 1969), 5 pp., A99-019, 199-13.

(b)(3)

'LA-14066-H

ASSIFIED

IV-59

L

UNCLASSIFIED SA GAAAMOO

IV-60

ARCKET AND

LA-14066-H

TIMOT A COLUTED

(b)(3)

LA-14066-H

SSIFIED

IV-62

LINCLASSIFIED LA-14066-H

(b)(3)

ASSIFIED

4. Detonators

a. Test Firing Data

(b)(3)

Historically, various types of detonators have been used, depending upon the type of primary, in the NTS test program. If one type of detonator had been used and it was decided to change to another type of detonator in a similar shot, it was of course necessary to understand any change in the behavior of the new detonator. One way to test any change was to fire the different types of detonators in field tests at Los Alamos.

The GMX-3 progress report for March 16 through April 15, 1969, noted that a request had been given to GMX-8 for a test fire that would compare the 1E30 detonator in a PBX 9407 pellet with the MC-1991 detonator. On April 18, GMX-8 personnel fired the shot. [Author's note: The GMX-8 firing pads were at Kappa site, TA-36.] The GMX-3 progress report stated, "The trace from the 1E30 is not identical with that from the MC-1991; it now remains for us to determine the difference in wave shape and to assess its effect on the system."³¹⁶

Apparently, GMX-8 personnel fired yet another shot. The GMX-3 progress report for May 16 through June 15, 1969, reported that the trace shapes were interchangeable.

(b)(3)

³¹⁶⁶Group GMX-3 Progress Report (U), March 16 through April 15, 1969," GMX-3-7818 (SRD) (no date), p. 10, A86-016. 32-16.

LA-14066-H

NCLASSIFIED

(b)(3)



4. Detonators

a. Test Firing Data

(b)(3)

Historically, various types of detonators have been used, depending upon the type of primary, in the NTS test program. If one type of detonator had been used and it was decided to change to another type of detonator in a similar shot, it was of course necessary to understand any change in the behavior of the new detonator. One way to test any change was to fire the different types of detonators in field tests at Los Alamos.

The GMX-3 progress report for March 16 through April 15, 1969, noted that a request had been given to GMX-8 for a test fire that would compare the 1E30 detonator in a PBX 9407 pellet with the MC-1991 detonator. On April 18, GMX-8 personnel fired the shot. [Author's note: The GMX-8 firing pads were at Kappa site, TA-36.] The GMX-3 progress report stated, "The trace from the 1E30 is not identical with that from the MC-1991; it now remains for us to determine the difference in wave shape and to assess its effect on the system."³¹⁶

Apparently, GMX-8 personnel fired yet another shot. The GMX-3 progress report for May 16 through June 15, 1969, reported that the trace shapes were interchangeable.

(b)(3)

¹⁰⁰Group GMX-3 Progress Report (U), March 16 through April 15, 1969," GMX-3-7818 (SRD) (no date), p. 10, A86-016. 32-16.

(b)(3)

IV-63

LASSIFIED

LA-14066-H



Test Fire Data for NTS

B. Pruit Ginsberg, who came to the Laboratory in 1970 to work in GMX-7, recalls that his group leader, Robert L. Spaulding, was a very particular person; his attitude was transmitted to every member of his group. Before each NTS test, several activities took place. Group members of GMX-7 would perform tests to make sure that their firing circuits and detonators were performing properly. These were the confirmation tests. Next Spaulding would send to the staff at PHERMEX a very complete layout of the firing and detonation circuits and all the required specifications. The PHERMEX group could then use this information in setting up any hydro shot that was to be completed for the forthcoming NTS test. Finally, a shot timing memo would be sent out from GMX-7 for use in the NTS event. ³¹⁹

(b)(3)

Included are the types of firing units, firing cables, signal cables, and the firing voltage, as well as the type of detonators and the detonator lot used in the firings. Firing data are reported, one for the shot and one for a backup shot.]

b. Early Work Applicable to the 1E33 Detonator Development

(b)(3)

Instead, the detonator group would develop a new type. [Author's note: It will be noted that the 1E30 was an important evolution from earlier detonators in the fact that it was much smaller. Its development, which Ginsberg recalls as being meticulously done, was an important precursor program for the 1E33.]

(b)(3)

³¹⁹B. Pruit Ginsberg, personal communication (SRD) (January 29, 2003).

IV-64

(b)(3)

LA-14066-H

INCLASSIFI

1

ICLASSIFIED

Taylor in his history has noted that the "new detonator cable developed by WX-7 was most beneficial."³²⁴

Today the 1E33 detonator is still in use in the W76. Ginsberg has reported that these detonators show no signs of deterioration with age.³²⁵

5. Engineering Computer Codes

The WX-3 progress report for November 16 through December 15, 1972, stated in regard to the Mk 400 program, "The SABOR-DRASTIC computer code is now operating correctly for the combined RV/WH models and results were obtained for one of the support schemes."

(b)(3)

By early 1973, the engineers had begun to consider using, for lateral support, a light foam over the entire length of the WH. The results from the SABOR-DRASTIC code runs for this type of mounting were encouraging.³²⁷

6. Assembly

(b)(3)

³²⁴John W. Taylor, "The W76 Program: An X-ray View (U)," M-2 TM253 (SRD) (January 8, 1976), p. 70. ³²⁵B. Pruit Ginsberg, personal communication (SRD) (January 29, 2003).

(b)(3)

³²⁷"Group WX-3 Progress Report (U), January 16 through February 15, 1973," WX-3-73-97 (SRD) (no date), p. 15 A86-016, -275-1.....

(b)(3)

LA-14066-H

SECRETIED UNCLASSIFIED



C. Vulnerability

1. Considerations

As the USSR began to develop missiles that carried nuclear weapons, military planners in the United States became concerned that these types of weapons could be used as defense weapons against incoming nuclear-armed missiles from the United States. The question then arose as to how to "harden" the U.S. reentry vehicles and warheads to minimize the impact of this type of Soviet defense.

In addition, it became technically possible in the United States to have one missile carry more than one warhead. As these warheads were released and detonated over a target(s), and if the offensive warheads were detonated too close together during a similar time period, the radiation released from one would affect the others. Again, there was the question of how best to deploy these types of warheads and how to "harden" each warhead from the effects of the others (fratricide).

In response to these problems, scientists in the U.S. weapon complex developed special materials and engineering features designed to minimize the damage (both from radiation and from the shock and heat produced by the interaction of radiation with materials) to a nuclear warhead (b)(3)

It was necessary to test these designs and materials to see if they met the design objectives. The tests included field-type tests and tests at NTS. In addition, computer codes were developed, based on experimental data, to predict the behavior of components under adverse conditions.

Several types of field tests were employed. In one type of test, shocks were sent into the special materials to study their behavior. Other tests measured the effects of high temperature and similar adverse environments. In another type of test, radiation from a radioactive source, an accelerator, critical assembly, or reactor was used to expose the device to neutrons or x-rays. The type and amount of radiation that could be delivered was dependent upon the irradiating source. These field tests were never able to duplicate an actual exposure environment during deployment.

(b)(3)

LA-14066-H

CLASSIFIED

The most complex and expensive tests employed were the NTS-type tests. These tests were used to mock-up, in so far as possible, the actual conditions that a warhead might be subjected to. In the following section these vulnerability tests will be briefly described in terms of the specific tests that were relevant to the pre-Phase 3 development of the XW76.

2. NTS Vulnerability Tests

a. Description

NTS vulnerability tests were "effects" tests where the output (neutrons/x-rays) from a detonated nuclear device was used to determine how various weapons, weapon materials, and engineering features would respond should they encounter a hostile environment. This environment might result from the nearby detonation of a nuclear ABM sent by the targeted country or from close detonation of warheads from a U.S. missile in a MIRVed type of deployment. The hostile environment might be encountered at high altitude or near the target.

In this type of NTS test, the sponsoring agency specified what type and level of radiation was required for their experiments. Participating agencies could also specify what exposure would be most useful in their experiments.

Many of the vulnerability tests were sponsored by the Department of Defense/Defense Atomic Support Agency (DOD/DASA). These tests usually had, as noted in Chapter I, double names. Some were sponsored by a weapon laboratory. The sponsoring agency's project managers decided (based on what they wanted in terms of radiation output) on a suitable device. The laboratory (Los Alamos or Livermore) that had designed this device was then asked to provide the device and to assume responsibility for emplacement and detonation of the device. In general, this device was usually a design that had already undergone a previous test(s) where the output radiation had already been determined. If a tested device were not available, a preliminary test to determine specific output might take place at NTS. In addition to supplying and detonating the device, the scientists at the specified laboratory were also responsible for the device diagnostics. In addition, certain diagnostics, for example seismic yield, were done on a routine basis by outside agencies.

Once the nuclear device with its known output had been specified, suitable stations at various distances from the device were set up in the facility in which the experiment was to take place. (Because of the layout required, vulnerability tests were usually done in tunnels at NTS.) The necessary shielding was installed. Instruments to measure the flux of the radiation falling on the samples to be tested were designed, built, and installed. The pieces of equipment or samples for which exposure data were desired were inserted at the specified locations. Special closure assemblies, used to close off the affected region and in theory, prevent bomb debris from spreading into the main tunnel/environment, were designed and installed.

The test configuration with its tunnels. test stations, access holes, and other required facilities could be extremely complex.

(b)(3) The zero room housed the detonated device, firing stand, firing and diagnostics equipment, and front-end closures. (b)(3)

Another line of slight extended from the zero room into an alcove designated "M." The line of sight for the major experimental areas was constructed in a main

LA-14066-H

UNCLASSIFIED MARIA/

drift 925 ft in length. A further 30-foot extension consisting of an arched tunnel housed alcove L. As shown in Figure IV-4, several shafts were used to access the various areas.³³³

John Hopkins, former Test-Division leader, has commented that each layout for a vulnerability test was different. However, all were major efforts in terms of construction and instrumentation.

(b)(3)

IV-68

LA-14066-H

LADSIFIED

LASSIFIED

b. Test List

(b)(3)

The test layout was designed with three experimental stations in the exposure pipe. An aluminum flux screen was used to cut off, at these stations, x-rays below 10 keV. The latter two stations also included the use of a polyethylene filter to further reduce the flux.

Not only were the various engineering/test groups at Los Alamos responsible for providing the nuclear device and diagnostics, but personnel from J-14 and W-7 along with personnel from EG&G were also responsible for x-ray effects measurements.³³⁷

(b)(3)

IV-69

A COTTAT

LA-14066-H

UNCLASSIFIED FT/RIQ

As noted in Table IV-1, the Los Alamos exposure experiments that would be the most relevant to the XW76 program were the detonator experiments.

(b)(3)

IV-70

7/84

LA-14066-H

UNCLASSIFIED

MARA UNCLASSIFIED

In another Los Alamos experimental set-up, ³/₄-inch-diameter cylinders, made up of disks of various materials with different thickness layers, were exposed. It was reported that the data obtained in this experiment would allow for the determination of damage thresholds. Another experiment was designed to investigate the thermal limits in various material interfaces. These interfaces included various cross sections of radiation case materials, high-Z loaded plastics, and cylinders of HE.

A listing and description of additional experiments is available in the cited references.³⁴⁷ Just after the test had been completed, the GMX engineers reported that good data had been obtained; however, the temperatures had been lower than expected.³⁴⁸

(b)(3)

LA-14066-H

/SAGREDURA UNCLASSIFIED

³⁴⁷John H. McQueen to Cdr. D. D. Swift, TC/DASA, Subject: "Final Report of LASL Data from Experiments Aboard the Hudson Seal Event," J-DO Tech (SRD) (April 21, 1969), 19 pp., A99-019, 265-13. "Program Status Weapons Research and Development, July – September 1968, Part 2 of Two," Los Alamos Scientific Laboratory report DIR-2142 (SRD) (no date), pp. 25-26.

³⁴⁸"Group GMX-3 Progress Report, September 16 through October 15, 1968," GMX-3-7455 (SRD) (no date), p. 15, A86-016, 32-10. "W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), pp. 41-42, A86-016, 242-7.

(b)(3)

ANAMA ANAM LA-14066-H UNCLASSIFIED



(b)(3)

UNCLASSIFIED

LA-14066-H

IV-74

ASSIFIED

The results from the material experiments were noted to have been generally successful. Additional information is available in the cited reference.³⁶¹

(b)(3)

LA-14066-H

:

AFAFEDARD UNCLASSIFIED

UNCLASSIFIED LA-14066-H

LASSIFIED AAAAA

ТА-14066-Н

ALCONDANCE UNCLASSIFIED

1

14

LA-14066-H

(b)(3)

In addition to the LASL experiments, plans were made for the extensive use of the test facility by other organizations

(b)(3)

Preparations for the test began at Los Alamos almost two years before the test would actually take place.

(b)(3)

UNCLASSIFIED

LA-14066-H

SŢ UNCLASSIFIED

(b)(3)

(b)(3)

³⁹⁴ W-Division Quarterly Status Report, April 1, 1970 through June 30, 1970," W-2264 (SRD) (July 15, 1970), pp. 17-<u>24, A86-016, 242-13.</u>

۱

(b)(3)

IV-80

ep/

NCLASSIFIED

ТА-14066-Н

LA-14066-H

(b)(3)

TV-81

SSIFIED

(b)(3)

SALABAAAAA

TA-14066-H

UNCLASSIFIED

LA-14066-H

UNCLASSIFIED

(b)(3)



During the summer of 1971, the Los Alamos group began working on their own setup for a full vehicle experiment

(b)(3)

In the spring of 1972 the engineers completed output calculations in order to verify the time history of the flux out of the exposed device and the exposure level at the 1,900-ft station.⁴³⁰

(b)(3)

Before the test, the Los Alamos field test groups had installed a trailer containing sufficient equipment to support approximately seventy channels of measurement instrumentation. After the test, it was reported that all the LASL instrumentation had recorded data.⁴³²

(b)(3)

"*Leslie M. Redman and Cecil C. Carnes, "Quarterly Status Report on Weapons Research and Development (U), for the Period Ending March 31, 1972," Los Alamos Scientific Laboratory report LA-4965-PR (SRD) (June 1972), ______ p. 71.

(b)(3)

⁴³²Luella M. Button, "Quarterly Summary of Field Testing and Instrumentation Development (U), for the Period Ending September 30, 1972," Los Alamos Laboratory report LA-5082-PR (SRD) (October 1972), p. 8.

IV-84

UNCLASSIFIED

LA-14066-H
(b)(3)

In the summer of 1970, the Los Alamos engineers in GMX-3 and GMX-7 began outlining plans for future exposure experiments. These included a hot x-ray effects test.⁴³⁸

(b)(3)

⁴³⁸ "Group GMX-3 Progress Report (U), July 16 through August 15, 1970," GMX-3-9023 (SRD) (no date), p. 17, A86-016, <u>33-16.</u>

(b)(3)

LA-14066-H

CLASSIFIED

IV-85

UNCLASSIFIED

(b)(3)

F ASSIFIED

LA-14066-H

IV-86

ARAND

In the spring of 1973, the Los Alamos groups installed their experiments in the number-2 test chamber and completed the installation of all diagnostic cables.⁴⁴⁵ A 32-channel multiplex system was designed and built to collect 30 channels of thermocouple data.⁴⁴⁶

After the detonation of the device, the multiplex unit collecting data from the thermocouples operated until the experimental wiring was destroyed after 100 milliseconds. The thermocouple data indicated that the temperatures measured at the outer surface of the HE had been higher than predicted. Nevertheless, a PBX 9501 pellet embedded in the mock HE system showed no evidence of decomposition.⁴⁴⁷ (Later, it was reported that perhaps the actual temperatures had been lower than those measured.⁴⁴⁸)

(b)(3)

⁴⁴⁵Luella M. Button, "Quarterly Summary of Field Testing and Instrumentation Development (U), for the Period Ending March 31, 1973," Los Alamos Scientific Laboratory report LA-5268-PR (SRD) (May 1973), p. 1.
⁴⁴⁶Luella M. Button, "Quarter Summary of Field Testing and Instrumentation Development (U), for the Period Ending June 30, 1973," Los Alamos Scientific report LA-5350-PR (SRD) (July 1973), p. 1.

⁴⁴⁷"Group WX-3 Progress Report (U), July 16 through August 15, 1973," WX-3-73-439 (SRD) (no date), p. 4, A86-016, 275-7.

⁴⁴⁸ Group WX-3 Progress Report (U), September 16 through October 15, 1973," WX-3-73-539 (SRD) (no date), pp. 4. 6. A86-016, 275-9.

(b)(3)

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

IV-88

LA-14066-H

segremen UNULABOLTIED

LA-14066-H

/\$AARET/BAD

(b)(3)

IV-89

UNCLASSIFIED



3. Calculations

The relevant literature of the late 1960s and early 1970s time period indicates that a very active program was carried on at Los Alamos to calculate the effects of x-rays and neutrons on the Los Alamos weapon designs. These calculations were then compared with the experimental results obtained in the NTS vulnerability tests.

It was reported that the most useful way of expressing the neutron vulnerability of a nuclear weapon was through use of an "F-number." These were customarily expressed in terms of the average number of reactions per kg of material per unit neutron fluence on the exterior of the carrying vehicle. The literature of the period extensively reports on the calculations of F-numbers.

(b)(3)

(b)(3)

ASSIFIED

LA-14066-H

IV-90[™]

CHAPTER V TABLE OF CONTENTS

CHAP	TER	V. WEAPON PROGRAMS AND CHANGES IN THE STOCKPILE:	
1	19651	МАҮ 1973	V-3
	A. Ph	ase 3 Programs at Los Alamos	V-3
	1.	Assignments	V-3
	2.	Weapon Programs	V-4
	3.	Conclusions Concerning the Phase 3 Program	V-48
	B. Ad	lvanced Development and Pre-Phase 3 Programs	V-49
	· 1.	1965	V-49
	2.	1966	V-53
	3.	1967	V-57
	4.	1968	V-60
	5.	1969	V-71
	6.	1970	V-82
	7.	1971	V-91
	8.	1972 and First Quarter of 1973	V-100
	9.	Conclusions	V-112
(C. Tre	ends	V-113
	1.	Change in Composition	V-113
	2.	New Weapon Systems	V-116
	3.	Decreased Levels of Funding	V-120
	4.	Tighter Test Requirements/Cost Increases	V-121
	5	Result	V_121
			4 - 1 - 1

LIST OF FIGURES

(b)(3)

Figure V-5.	Number of Weapons (by Type) in the U.S. Stockpile as a Function of Year.	V-114
Figure V-6.	Number of Builds and Number of Retirements of	
	U.S. Nuclear Weapons as a Function of Year	V-115
Figure V-7.	Total Yield of Nuclear Weapons in the U.S. Stockpile as a	
	Function of Weapon Type and Year	V-117
Figure V-8.	Use of Reactor Products in U.S. Nuclear Weapons by Force Structure	V-118
Figure V-9.	Use of Highly Enriched Uranium in the	
	U.S. Nuclear Weapons Stockpile by Force Structure	V-119

LA-14066-H

SECRETATO UNCLASSIFIED

V-1

This Page Intentionally Left Blank

AFCARATION UNCLASSIFIED

LA-14066-H

CHAPTER V. WEAPON PROGRAMS AND CHANGES IN THE STOCKPILE: 1965–MAY 1973

A. Phase 3 Programs at Los Alamos

1. Assignments

a. Phase 3 Programs Entering the Stockpile

During the period 1965–May 1973, the Los Alamos Scientific Laboratory weapon development and design group members were responsible for several Phase 3 weapon programs. [The Phase sequence for weapon programs is noted in Chapter I.] For those weapons that entered the stockpile, the specific weapon program, the date of the Phase 3 award, and the date of the attainment of Phase 6 are as follows:¹

Weapon Program	Phase 3	Phase 6				
B61 Mod 0	June 1963	January 1967				
B61 Mod 1		February 1969				
B61 Mod 2	August 1971	June 1975				
B61 Mod 3	March 1972	October 1979				
W66*	January 1968	October 1974				
W69	January 1967	February 1972				
W72	May 1969	September 1970				
* Never deployed to the field ²						

b. Canceled Phase 3 Assignments

In addition to those weapon programs assigned to Los Alamos that were in or went into Phase 3 during the 1965 to 1973 period and entered the stockpile, several Phase 3 programs were assigned to Los Alamos but were later canceled. These were the XW64, XW67, XW73, and XW74.³

c. Discussion

The Phase 3 weapon programs under development at Los Alamos from 1965 through May 1973 will be discussed in the following sections. Each section will cover a specific weapon.

LA-14066-H

k*Æ™Æ*Æ ASSIFIED

¹"FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), pp. 25–26, 301, 309, 326.

²⁴'FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), p. 301.

³Betty L. Perkins, "Why Nougat? (U)," Los Alamos National Laboratory report LA-12950-H (SRD) (November 1, 1995), p. A-3.

UNCLASSIFIED cret ra

The B61 Mod 0, 2, and 5 bombs

2. Weapon Programs

a. B61

While the Phase 3 assignment for the B61 Mod 0 was received by the Los Alamos Scientific Laboratory in 1963, the Los Alamos weapon groups were going to find that they would have an extended development program for this weapon. The B61 would go through many models and deployment objectives over a period of many years. Several Mods of this bomb, in modified designs from the ones that were first developed, are still in the U.S. nuclear weapon stockpile.

(b)(3)

would be carried by U.S. Navy A-6E and F/A-18A/C/D aircraft. These bombs would also be carried by U.S. Air Force F-16 A/B/C/D and F-111 D/E/F aircraft. The later Mod 3, 4, and 10 variations would be carried by U.S. Air Force F-16 A/B/C/D and F-111 D/E/F aircraft as well as F117-A aircraft. The Mod 3, 4, and 10 bombs would also be carried on NATO F-16 A/B aircraft and on Tornado aircraft. A recent version (Mod 7) is carried on the Air Force B-2A and B52-H. As the different Mods were introduced into the stockpile, many additional safety features would be included. The early Mod 0-Mod 2 designs discussed in this section have been retired from the stockpile or converted.⁴

(b)(3)

The B61 Mod 0 and Mod 1 weighed 715 lb and had a diameter of 13.3 inches and a length of 141.6 inches. The B61 Mod 0 first entered the stockpile in January 1967/

(b)(3)

⁴"FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), pp. 25, 29.

ACREATING

NCLASSIFIED

LA-14066-H

(b)(3)

Because most of the development work on the Mod 3 occurred after the award of the XW76's Phase 3, the history of this effort again will not be included in this section.

(b)(3)

LA-T4066-H

SECRETICA UNCLASSIFIED



₩6

A TAK LA-14066-H CLASSIFIED

UNCLASSIFIED Septented

LA-14066-H

SHOREFIED UNCLASSIFIED

(b)(3)

ALTREPART

UNCLASSIFIED

LA-14066-H

(b)(3)

V-8



UNCLASSIFIED

V-10

AFCREY BO UNCLASSIFIED

LA-14066-H

(b)(3)

LA-14066-H

AFCREMENT (UNCLASSIFIED



However, there were numerous production problems in these early units. On April 26, 1967, USAEC area manager, L. P. Gise, sent a TWX to Bradbury and to J. A. Hornbeck, president of Sandia Corporation. He also sent a copy to DMA Director, D. L. Crowson. Gise stated, "The Mk 61 bomb program at this point in time (seven months after Phase 5 FPU [First Production Unit] and four months after FPU/WR) has not yet attained any semblance of a production item, and it is substantially behind both WR and "Type" UU schedules." Gise noted, "In view of this support material status, I can only conclude that we have not yet attained a consistently producible design." Gise reported that selected aspects of the nuclear system were under review. Gise then stated, "Towards this end, and subject to your comments, I intend to propose to DMA [Division of Military Application] that Mk 61 final assembly be immediately suspended." Gise noted that resumption of the final assembly operations would be contingent upon clearing up the major design/production problems. But he added, "I would hope that you would not interpret this stop production proposal as an invitation to initiate major redesign of all the troublesome components, but rather limit design or production changes to those definitely necessary to make the Mark 61 producible on a production line basis."³⁷

(b)(3)

³⁷USAEC L. P. Gise, Albuquerque, N.M. to RUWTHRA/N.E. Bradbury, Dir. LASL, Los Alamos, N.M.; ZEN/J. A. Hornbeck, Pres., Sandia Corp., Albuq., N.M. (SRD) (April 26, 1967), 3 pp., A99-019, 214-6.

(b)(3)

LA-14066-H

ASSIFIED

UNCLASSIFIED SEACHERNED ((b)(3) Autorian UNCLASSIFIED V-13 14000-11



ARCRIE AM UNCLASSIFIED

LA-14000-H



UNCLASSIFIED \$RATE AND

UNCLASSIFIED LA-14066-H



In the late 1960s there was increasing concern over the safety and security of nuclear weapons.

(b)(3)

As previously noted, after production of the Mod U and Mod 1, other Mods of the B61 were planned. At the January 14, 1972, meeting of the TX Committee it was reported that when the B61 went "back into production the next time," there might be problems because some of the hardware companies had gone out of business or did not want the AEC contracts

(b)(3)

LA-14066-H

LASSIFIED



b. XW64

In May 1964, the Military Characteristics for nuclear warheads for the Lance Missile system were issued. This missile was to be a surface-to-surface missile for use by the Army. Both Los Alamos and Livermore submitted proposals for this warhead. As it would turn out, both Laboratories would initially receive the authorization to proceed. The XW64 was the nomenclature given to the Los Alamos design; XW63 was the nomenclature given to the Livermore design.

Los Alamos received the authorization to proceed with the development of the XW64 on July 20, 1964, (b)(3)

However, the initial authorization was soon withdrawn. Attempts were made by the Los Alamos management in 1966 to have the program reactivated.⁷⁷ But neither the XW63 nor the XW64 was ever produced.

The final warhead, that provided a nuclear capability for the Army's surface-launched guided missile (MGM-52C) known as Lance, was the W70. The Phase 3 assignment for the W70 warhead was given to Livermore in April 1969

(b)(3)

c. W66

The W66 was a two-stage thermonuclear warhead designed for use on the Sprint missile. This missile was a short-range, low-altitude, quick reaction intercept missile. Therefore, the missile had high launching and maneuvering accelerations and a very short reaction time between launching and the firing of the warhead.

(b)(3)

The Sprint missile/warhead was part of the Safeguard weapon system.

The AEC laboratories originally received the Phase 3 authorization for the development of the warhead for the Sprint missile system in September 1965.

(b)(3)

⁷⁷N. E. Bradbury, Los Alamos Scientific Lab., Los Alamos, N.M. to Brig. Gen. Delmar L. Crowson, DMA, USAEC, Wash., D. C., DIR-2017 (CRD) (February 14, 1966), p. 1, A99-019, 217-7.

(b)(3)

V-18

NET ASSIFIE

LA-14066-H

UNCLASSIFIED AFT/BF/VBD

LA-14000-H

UNCLASSIFIED

V-19



7SPQ UNCLASSIE LA-14000-m

1

UNCLASSIFIED SEARAY MA (b)(3) ARCMETARIA UNCLASSIFIED **V-21** A-14000-M



UNCLASSIFIED ANDRADAA

A-14000-F1

Andreaver UNCLASSIFIED

UNCLASSIFIED

AFARATION

LA-14000-11

UNCLASSIFIED

ľ

V-24

V-25

UNCLASSIFIED

(b)(3)

UNCLASSIFIED

UNCLASSIFIED

(b)(3)

The required engineering and environmental test program was completed. The drawings of / the weaponized design were released.

The Laboratory's final quarter report for 1972 states, "The SLA-LASL Interim Development Report for the W66 is complete, and will be reviewed by the Design Review and Acceptance Group next quarter. We drafted production specifications for the W66 primary assembly and reviewed them with personnel from the Burlington AEC Plant." It was reported that time had been spent at Bendix. Kansas City. incorporating improved operations to increase the ouality of the W66 tiles.

(b)(3)

The W66 entered the stockpile in October 19/4. However, it was never deployed to the field. All units were retired in August 1986.¹²²

d. XW67

The XW67 was originally designated as the warhead for the Mk 17 RV to be used on the Minuteman II and Poseidon C3. (b)(3)

The warhead, given to Los Alamos, went into Phase 3 in the spring of 1966. The first production unit was scheduled for October 1968. The Military Characters were approved on June 28, 1966.¹²³

(b)(3)

¹²² FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), pp. 299-301.

¹²³"LASL Program for Fiscal Years 1968-1969," Los Alamos Scientific Laboratory report DIR-2081 (SRD) (May 24, 1967), p. 17. R. G. Shreffler to Distribution, Subject: "LASL XW-67 Program," AW-1225 (SRD) (July 12, 1966), 10 pp., A99-019, 227-2. Brigadier General Delmar L. Crowson, Director of Military Application to L. P. Gise, Manager Albuquerque Operations, Subject: "Mk-17 Military Characteristics," (SRD) (July 14, 1966), 1 p., A99-019, 227-2.

LA-14066-H

NCLASSIFIED

UNCLASSIFIED FARELT/RUA

The W-Division quarterly status report for January 1 through March 31. 1967. indicated that the nit system for the XW67 warbead was in the design phase.

(b)(3)

The W-Division quarterly report for January through March 1967 reported that environmental tests were being planned. (b)(3)

It was noted that the development of the XW67 was felt to be

The W-Division quarterly report for April 1 through June 30, 1967, notes that development work was continuing. Extensive structural analysis on the design was being performed.¹²⁹ During the summer of 1967, the engineers found that the XW67 prototype would need to be reengineered; the proposed design did not meet the load requirements.

(b)(3)

¹²⁵ W-Division Quarterly Status Report, January 1, through March 31, 1967, Part 2 of Two," W-2020 (SRD) (April 14, 1967), pp. 15, 28-30, A86-016, 242-1.

¹²⁹ W-Division Quarterly Status Report, April 1, 1967 through June 30, 1967, Part 2 of Two," W-2035 (SRD) (July 17, 1967), pp. 2–3, 23, A86-016, 242-2.

V-28

SECREARD

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED KREATARAD

Highball was to have been the full-scale test.¹³³ However, it never took place. A TWX from the AEC dated December 18, 1967 stated, "The W67 program is hereby cancelled. Please terminate all effort on this program."¹³⁴

The W-Division status report for October 1, 1967, through December 31, 1967, also states that the Mk 17 reentry vehicle system/Mk 67 warhead had been canceled. It was noted, "Probably most of the instrumentation can be utilized in other operations, but the majority of the warhead is to be scrapped."¹³⁵

Harold Agnew, in a 1972 letter to Camm, summarized the reasons for the cancellation of the XW67. He noted, "Both the Air Force (Minuteman) and the Navy (Poseidon) became more and more concerned about the ABM defenses that the Soviets might be deploying in the future, and they both feared that a single RB [re-entry body] might be vulnerable to these defenses, whereas they felt that a cluster of smaller RB's (MIRV's) could defeat them. At the same time, both services decided that they could not afford to support two programs. This coupled with fears concerning first strike capability implications caused the Mk 17/W67 to be cancelled in December, 1967."¹³⁶

(b)(3)

133"LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 1967), p. 17.

¹³⁴USAEC, W. Lee Hancock, Albuquerque, N.M. to RUWTHRA/N. E. Bradbury, LASL, Los Alamos, N.M. (SRD) (December 18, 1967), 1 p., A99-109, 217-14.

¹³⁵ "W-Division Quarterly Status Report, October 1, 1967 through December 31, 1967, Part 2 of Two," W-2084 (SRD) (January 15, 1968), p. 23, A86-016, 242-4.

¹³⁶H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), p. 4, B11, Drawer 56, Folder 1 of 4.

LA-14066-H

LADDITIED



(b)(3)

e. <u>W69</u>

(b)(3)

warhead entered the stockpile in 1972

The W-Division quarterly status report for January 1 through March 31, 1967, indicates that the XW69 warhead was now in Phase 3. By March, a preliminary design had been released for warhead development scheduling. (b)(3)

The next quarterly report from W-Division (April 1, 1967 through June 30, 1967) indicates that the development effort on the XW69 was continuing

(b)(3)

FORFINKE

V-30

LA-14066-H

JNCLASSIFIED

This retired
(b)(3)

The status report for weapons research for April–June 1968 indicates that a mockup assembly for the first flight test had been sent to Sandia. Other assemblies had been sent to Sandia for various engineering and environmental tests. When several tests had indicated some problems, engineering modifications had been made. The forward case design had been released.¹⁴⁶

(b)(3)

Laboratory report DIR 2133 (SRD) (no date), p. 14.

LA-14066-H

UNULADOIDIED

SPCRET/RD

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

Shipper to the production

LA-14066-H

UNCLASSIFI

V-33

(b)(3)



In the early part of 1969, the steel case cover for the W69 was reengineered to provide a more robust design.¹⁵⁸ Engineering tests continued. During 1970, in the program to develop the forward support system, environmental covers were tested. Additional testing to better understand the fatigue characteristics was recommended.¹⁵⁹

In 1971, engineering development work continued. During the first quarter of 1971, assembly procedures, drawings, and inspection documents were reviewed. It was reported that no difficulties were anticipated in meeting the Phase 5 date. A meeting to discuss W69 preproduction activities was held at SLA on February 2, 1971. Personnel from the Burlington and Pantex AEC_Plants, AEC-ALO, ALA, and LASL participated.¹⁶⁹

(b)(3)

Long-term storage tests were also recommended.¹⁶¹

During the summer of 1971, final preparations for Phase 5 continued.¹⁶² The Laboratory's third-quarter report indicates that LASL personnel had visited the Burlington Plant to observe the assembly of the first production unit. Although there had been a variety of typical start-up problems, these were in the process of solution.¹⁶³ The Laboratory's fourth quarter report notes that there continued to be small assembly problems at Burlington that were being solved. The LASL representatives had observed disassembly and reassembly of a Type 5 warhead at Pantex.¹⁶⁴

Phase 6 was achieved in February 1972.¹⁶⁵

(b)(3)

After the Phase 6 status of the W69 was achieved, there continued to be activities related to this weapon. Several of these will be noted in the following paragraphs.

¹⁶¹"Quarterly Status Report on Weapons Research and Development for the Period Ending June 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4732-MS (SRD) (August 1971), p. 40.

¹⁶⁵ "FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), p. 309.

LA-14066-H

(b)(3) REALBAD UNCLASSIFIED

¹⁵⁸ W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 27, A86-016, 242-9.

¹⁵⁹ W-Division Quarterly Status Report, October 1, 1970 through December 31, 1970," W-2301 (SRD) (January 15, 1971), p. 71, A86-016, 242-15.

¹⁶⁰ W-Division Quarterly Status Report, January 1, 1971 through March 31, 1971," W-2318 (SRD) (April 15, 1971), p. 18, A86-016, 242-16. "Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4680-MS (SRD) (May 1971), p. 37.

¹⁶² W-Division Quarterly Status Report, July 1, 1971 through September 30, 1971," W-2343 (SRD) (October 15, 1971), p. 62, A86-016, 242-18.

¹⁶³"Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4820-PR (SRD) (December 1971), pp. 62–63.

¹⁶⁴"Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4880-PR (SRD) (March 1972), p. 42.



SSIFIED

f. W72

The W72 was a modified W54 designed for use in the Walleye missile. The Walleye missile was an air-to-surface, TV-guided glide bomb.¹⁷⁵ It was developed for use on Air Force F-4D aircraft.¹⁷⁶

The Phase 3 assignment was made in May 1969. The Laboratory's program status report for April–June 1969 notes that the BA72 program to develop a nuclear warhead for the Walleye using W54 nuclear systems was to be mainly a Sandia activity₄

(b)(3)

The lot showing

the best performance was chosen for use in the Walleye's warhead. Other tests related to the W72 development took place at Sandia and at Burlington.¹⁷⁸

Phase 6 for the W72 was achieved in September 1970. May 1971 was the scheduled Initial Operational Capability date.¹⁷⁹ All units had been retired by September 1979.¹⁸⁰

The LASL weapons quarterly for the third quarter of 1972 suggested that a convertible capsule design that allowed for conversion from a conventional HE to a nuclear warhead might be used in the extended-range Walleye.¹⁸¹ This suggestion was never implemented.

(b)(3)

⁴⁷⁵ Program Status Weapons Research and Development July- September 1969 (U)," Los Alamos Scientific Laboratory report DIR-2187 (SRD) (no date), p. 61.

¹⁷⁶ "FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), p. 325.

(b)(3)

¹⁷⁵ Program Status Weapons Research and Development July– September 1969 (U)," Los Alamos Scientific Laboratory report DIR-2187 (SRD) (no date), p. 61.

¹⁷⁹"Weapons Program Study and Development Report," Headquarters Field command Defense Atomic Support Agency, Sandia Base, Albuquerque, New Mexico report FC10700038 (SRD) (October 1, 1970), p. 25.

¹⁸⁰"FY 1994 Annual Weapons Program Report," DOE Albuquerque Operations Office report (SRD) (October 1, 1994), p. 326.

¹⁸¹Leslie M. Redman, Cecil C. Carnes, Jr., "LASL Weapons Quarterly for the Period Ending September 30, 1972
(U)," Los Alamos Scientific Laboratory report LA-5139-PR (SRD) (January 1973), p. 72.

LA-14066-H

BANNE UNCLASSIFIED



g. XW73

The Condor was a proposed Navy missile. It was to be a TV-guided, rocket-powered, air-tosurface missile. The warhead for this missile was designated the BA73. Initially, two versions of the warhead were to be provided.

(b)(3)

The initial plans for the warhead were to use a slightly modified W69.¹⁶² The Condor Development Authorization notice for the nuclear warhead section is dated July 15, 1969.¹⁸³ However, Giller, in a November 24, 1969, TWX to the Laboratory, noted that the Initial Operational Capability (IOC) date for the Condor warhead had been delayed. The Laboratory was instructed to study all the possible alternatives in the design and development of nuclear systems for the Condor.¹⁸⁴

(b)(3)

A letter dated July 13, 1970, to the Chairman of the AEC from the DDR&E (Director, Defense Research & Engineering) states, "Recent program reviews have resulted in a decision to delay a commitment to production of the Condor missile system until the completion of system engineering development and operational evaluation. This action will also delay the Navy support for the nuclear warhead development program until approximately September 1972 when further Condor program decisions can be expected." The letter also notes, "It may be prudent for the AEC to stop all nuclear Condor development activity until a Navy commitment to production is made." However, it did appear that some developmental activities might have to continue if the Condor warhead were to become operational as early as January 1975.¹⁸⁷

(b)(3)

(b)(3)

LA-14066-H

INCLASSIFIED

¹⁸⁷USAEC, Edward B. Giller, Wash., D.C. to AN3, USAEC, H. C. Donnelly, Albuquerque, N.M. et. al., (CRD) (July 31, 1970), 2 pp., A99-019, 39-7.

∀-38

¹⁸²"Program Status Weapons Research and Development July– September 1969 (U)," Los Alamos Scientific Laboratory report DIR-2187 (SRD) (no date), p. 61. "W-Division Quarterly Status Report, July 1, 1969 through September 30, 1969," W-2217 (SRD) (October 15, 1969), p. 27, A86-016, 242-11.

¹⁸³H. C. Donnelly, Manager, Albuquerque Operations Office, AEC to N. E. Bradbury, Director, LASL and J. A. Hornbeck, President SLA, Subject: "Condor Development Authorization," (CRD) (July 15, 1969), 2 pp., A99-019, 198-11.

¹⁸⁴USAEC, Edward Giller, Wash., D.C. to USAEC, H. C. Donnelly (SRD) (November 24, 1969), 3 pp., A99-019, 2<u>18-18.</u>

UNCLASSIFIED AMARKET/BAD

The AEC and the Navy then apparently decided to suspend specific warhead development until the Condor missile was farther along in its development. It was felt that not to do so might result in a "less than optimum nuclear warhead" to interface with the missile. In October 1970, it was reported that the W73 was in a suspended Phase 3. Future development efforts were uncertain.¹⁸⁹

(b)(3)

¹⁸⁹ Weapons Program Study and Development Report," Headquarters Field Command, Defense Atomic Support Agency, Sandia Base, Albuquerque, New Mexico report FC10700038 (SRD) (October 1, 1970), p. 26._____

(b)(3)

LA-14066-H

AFAMENARI UNCLASSIFIED

(b)(3)

The Advanced Planning Document for the W73 was issued in October 1971.

(b)(3)

The third quarter report for 1972 from the Laboratory states, "The insertable-capsule (IC) concept has recently interested the AEC and DoD for several convertible warhead applications. The convertible WH uses a conventional HE warhead, which, if necessary can be converted to a nuclear warhead by inserting a fissionable core. ...Some of the more interesting applications are for Harpoon, the Modular Guided Glide Bomb, Condor, the Mk 48 and Mk 46 torpedoes, and the Mk 84 bomb." It was also noted, "LASL is working on a proposed program to field a nuclear test of a device representative of the IC, convertible-warhead concept before June 30, 1973."¹⁹⁶

Despite the initial warhead development program discussed briefly in the previous paragraphs, the warhead was never produced.

h. XW74

۱. ·

The XW74 was an Army-Navy-proposed 155-mm projectile. The project was canceled in June 1973 at the end of the Phase 3, which had been awarded to Los Alamos.

The Laboratory status report for weapons for the period January-March 1969 notes that a Phase 2 feasibility study for a new 155-mm nuclear round for the Army was nearing completion. Two designs had been proposed.

(b)(3)

(b)(3)

¹⁵⁰Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), p. 72.



V-40°



The Laboratory's status report for that period gives detailed information on the material problems that would have to be solved should the design be a successful one.¹⁹⁹-The Phase 3 for the W74 warhead was initiated on October 28, 1969.²⁰⁰ The Laboratory was assigned engineering development responsibility on February 24, 1970.²⁰¹

The Laboratory status report for January-March 1970 reports, "Authorization to proceed with a Phase 3 development program for a 155-mm AFAP [Artillery Fired Atomic Projectile] was received from DMA on February 24. The list of approved Military Characteristics and the Army Stockpile to Target Sequence were received in early March. The Army nomenclature for the AFAP is XM-517. AEC nomenclature for the nuclear warhead is W74.

(b)(3)

The January 1 through March 31, 1970, W-Division report indicates that engineering design and field test efforts were continuing for this project.

(b)(3)

During the spring of 1970, design and procurement of components for the local hydrodynamic shots was reported as proceeding on schedule. Special high-pressure materials were undergoing development.²⁰⁴

During the summer of 1970 various specific pit designs were studied for use in the XW74.

(b)(3)

'A structural test plan was outlined. Stress analysis was being performed using the

(b)(3)

Program Status Weapons Research and Development October – December 1969 (U)," Los Alamos Scientific Laboratory report DIR-2195 (SRD) (no date), p. 65.

²⁰⁰ "Nuclear Technology and Analysis Report (U)," Field Command Defense Nuclear Agency, Kirtland Air Force Base, New Mexico 87115 report HQDNA-185M, (SRD) (August 1, 1972), p. 32, B11, Drawer 57, Folder 1 of 2.
²⁰¹ USAEC, Edward B. Giller, Wash., D.C. to C13, N. E. Bradbury, LASL, Los Alamos, N.M., BW3, M. M. May, LRL. Livermore. Calif. (CRD) (February 24, 1970), 1 p., A99-019, 198-12.

(b)(3)

²⁰⁴ W-Division Quarterly Status Report, April 1, 1970 through June 30, 1970," W-2264 (SRD) (July 15, 1970), p. 40, A86-016, 242-13.

LA-14066-H

(b)(3) UNCLASSIFIED

SAAS II code

(b)(3)

During the first quarter of 1971 engineering design work continued. The status report from W-Division for January 1 to March 31. 1971 notes that the examination of alternate case materials was continuing.²⁰⁷ (b)(3) The Laboratory s status report ending March 1971 indicates that two

concepts relevant to nonviolent disablement had been investigated theoretically. The report also notes, "The AEC production schedule for the W74 has slipped because of funding problems. The date of the Phase 5 is now December 1974 and for the Phase 6 is March 1975."²⁰⁹ The Laboratory's status report ending June 1971 reports that weapon denial schemes were continuing to be investigated.

(b)(3)

^{20/1}*W-Division Quarterly Status Report, January 1, 1971 through March 31, 1971," W-2318 (SRD) (April 15, 1971), pp. 19-20, A86-016, 242-16

(b)(3)

²⁰""Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4680-MS (SRD) (May 1971), pp. 38–39.

(b)(3)

V-42

LA-14066-H

ICLASSIFIED

(b)(3)

Engineering design and test activities had also continued during the summer.

(b)(3)

In early November the design/engineering teams had begun the next iteration towards producing a weaponized device.

(b)(3)

ŚĘĊŖĘŊĄŖ LASSIFIED

LA-14066-H

-43



The Model 12 is discussed in the cited reference.²³¹

Many new materials were also developed for the W74's Model 12, including PBX 9501 and various alloys. Additional information is available in the cited reference.²²²

(b)(3)

(Additional

LA-14066-H

LASSIFIED

information on the development of this explosive is reported in Chapter IV. As previously reported, the W76 warhead would use PBX 9501 in the primary's HE system.)

(b)(3)

²²¹R. Canada to WPRC, Subject: "Notes from March 2, 1972, WPRC Meeting," (SRD) (March 14, 1972), pp. 1–2, B11, Drawer 53, Folder 1 of 2.

²²²"Weapons Working Group, Minutes of the 243rd Meeting," WWG-243 (SRD) (August 23, 1972), pp. 8–13, B11, Drawer 53, Folder 1 of 2

(b)(3)



design of the XW74.

However, these tests were very useful in guiding the

(b)(3)

¹²⁹R. K. Osborne and M. T. Thieme, "Theoretical Design of Implosion Weapons, 1959-1980 (U)," Defense Research Review. UCRL 53880-4-2 (SRD) (July 1992). p. 72.

(b)(3)

LA-14000-H

SAABET/KA UNCLASSIFIED

(b)(3)

(. The question of what was the most suitable design to use in the "revised" XW74 was discussed during the September 25, 1972, WLPC meeting

(b)(3)

The device is shown in Figure V-4. Additional information is available in the cited

(b)(3)

SHARFMRD (

LA-14066-H

UNCLASSIFIED

V-46

~

Ľ

LASSIFIED

references.249

(b)(3)

! However, the Initial firing tests had been fairly conservative as to the structural design, and therefore it was not known what the exact modifications would have to be.²⁴²

(b)(3)

²⁴² Military Applications Planning Committee, Minutes of the 17th Meeting, December 15, 1972," TDW-48 (SRD) (December 15, 1972), p. 1, B11, Drawer 53, Folder 1 of 2.

LA-14066-H

ACRATICAD UNCLASSIFIED

JNCLASSIFIED SACKAYARA

(b)(3)

It was to be the final proof test of the W74.²⁴⁰ This test never took place. The XW74 program was cancelled in June 1973.

3. **Conclusions Concerning the Phase 3 Program**

Development Problems a.

The brief summaries of the Los Alamos Phase 3 programs reported in the previous sections of this chapter demonstrate that even after a Phase 3 award had been made, the development program did not necessarily go smoothly. There were many tests where the test results were not the predicted results. In several cases, a large number of tests were required before an adequate design was available for the stockpile.

The Phase 3-to-stockpile sequence was not an easy, predictable one.

Ь. **Primary Considerations**

It is not surprising that the Los Alamos Laboratory during this period did not receive a Phase 3 assignment for a strategic missile system that used MIRVed warheads.

(b)(3)

Focus of Work c.

The reader will also have noted, in the discussion of the Phase 3 weapon programs at Los Alamos, that in the early 1970s the Phase 3 work was becoming less and less of the total work load. Except for the continuing work on the B61, either (1) the weapons had successfully gone to the stockpile or would soon go to the stockpile, or (2) the Phase 3 work had been terminated or soon would be terminated.

(b)(3)

²⁴⁶B. Heil, W-9 to WPRC Members, Subject: "FY 1973 NTS Tests," W-9-1351 (SRD) (February 23, 1972), p. 22, B11, Drawer 51, Folder 3 of 3. AACKET/MAN TINCLASSIFIED

LA-14066-H

The completion/cancellation of the Phase 3 programs without receiving new Phase 3 programs would have meant that a work reassignment would have been required for many Laboratory employees. A large number of persons were employed either directly or indirectly in a Phase 3 program. Not only were the primary/secondary design groups and NTS test groups involved, but also the field test groups, the engineering development and design groups, and the materials science groups. There were in addition many persons at the Laboratory who served to track the efforts and coordinate the work with the staffs at the weapon facilities (such as Sandia, Rocky Flats, Pantex, and Oak Ridge) as well as with the Military.

d. Status Symbols

In a sense, receiving an assignment of a Phase 3 program was a status symbol. It indicated that that laboratory was making a valuable contribution to maintaining for the United States a modern, nuclear weapon stockpile.

This status symbol was important politically. It was also important for maintaining high morale among those involved in the weapons program.

B. Advanced Development and Pre-Phase 3 Programs

During this 1965-early 1973 period, in addition to Phase 3 programs assigned to Los Alamos, the members of the various weapon groups were also working on programs that were in the Phase 1 or 2 stage. The members of these groups were also involved in a large number of advanced development projects. Many of the programs/projects were canceled before they reached a Phase 3 designation. However, as a historical background for the XW76 program, it is important to understand these programs.

The following discussion of advanced development and pre-Phase 3 programs will focus on projects under consideration in a particular year. [Author's note: These projects, especially those in advanced development, have been difficult for this author to track because the programs often changed names, changed objectives etc. The discussion presented in the following sections is by no means complete.]

The Mk 18 and Mk 400 programs are, however, not included. These were specific precursor programs to the XW76. Because of their importance in the development of the XW76, they will be described in detail in the following chapter. The small primary development program, previously described in Chapter II, will also not be included.

1. 1965

(b)(3)

LA-14066-H

ASSIFIED

(b)(3)

Sprint

(b)(3)

Thumbelina

(b)(3)

Mk 17 Warhead and the Increased Penetrability Program

(b)(3)

Other Applications for High-Yield Weapons

Other applications for these high-yield types of weapon designs included use in the Advanced Manned Strategic Aircraft [AMSA], the ICBM-X, and the Poseidon warhead. None of these systems had, at that time, been approved.²⁵⁴

(b)(3)

XF/1/RP

254 ... LASL Program for Fiscal Years 1966-1967," DIR-1980 (SRD) (May 25, 1965), p. 40.

LA-14066-H

UNCLASSIFIED

ł

V-50

(b)(3)

Short Range Attack Missile (SRAM)

(b)(3)

Improved Pershing

(b)(3)

Follow-On Lance

After the award of the Phase 3 for the Lance missile warhead (W70) to Livermore, the Military requested, as a follow-on, a study of a warhead for the Lance and a warhead for an ABM that had low requirements of reactor products

(b)(3)

LA-14000-H

ANAMENMA UNCLASSIFIED

(b)(3)

d. Single-Stage Weapons Primary

(b)(3)

11

SADM [Special Atomic Demolition Munition]

(b)(3)

e. Projectiles 175-<u>mm Shell</u>

(b)(3)

This

LA-14066-H

nuclear device was to be a close ballistic match to the conventional high-explosive round.---

8-in. Shell

٢

(b)(3)

(b)(3)

TYKI

264"LASL Program for Fiscal Years 1966-1967," DIR-1980 (SRD) (May 25, 1965), p. 42.



Two tests were planned for the coming year.260 /

1966 2.

Я.

Sprint_

(b)(3)

The LASL yearly program report notes, "...LASL feels that considerable further time and 1" work are required to understand the nuclear effects of the possible Sprint warheads sufficiently to make an optimum choice for the Sprint missile. Since the Sprint IOC [Initial Operational Capability] has been extended to October 1971, and since a warhead can be developed in three years, it would seem logical to use the next year to evaluate the nuclear effects in detail."268

(b)(3)

It w. noted, "A Ready Item of this system can be available upon 2-3 months notification of atmospheric testing."271

266"LASL Program for Fiscal Years 1966-1967." DIR-1980 (SRD) (May 25, 1965), np. 40-42.

(b)(3)

268"LASL Program for Fiscal Years 1967-1968," DIR-2029 (SRD) (no date), pp. 45-46.

(b)(3)

MORETIRI

271" LASL Program for Fiscal Years 1967-1968," DIR-2029 (SRD) (no date), p. 30.

LA-14066-H

UNCLASSIFIED

(b)(3)

<u>Thumbelina</u>

(b)(3)

<u>Mk 17</u>

As previously reported, the warhead for the Mk 17 RV (XW67) went into Phase 3 during the spring of 1966.

(b)(3)

AMSA, ICBM-X

(b)(3)

- -----

Maneuvering Reentry Vehicle (MRV)

A study dated December 28, 1966, was released by LASL and SCSL titled "Technical Data Package for MRV Study." This study presented information on possible nuclear warheads and related design guidance for use by the USAF/BSD [Ballistic Systems Division of the Air Force Systems Command] and their contractors for the Maneuvering Reentry Vehicle Study Program. Eight options ranging in yield from a 10-kt yield to a 2-Mt yield were presented.²⁷⁶

<u>Scampi</u>

(b)(3)

Short Kange Attack Missile (SRAM)

The Los Alamos weapon groups were continuing, during this period, to consider designs suitable for use in a Short-Range Attack Missile

(b)(3)

""" Technical Data Package for MRV Study," W-1-E-12050 (SRD) (December 28, 1966), 20 pp., A99-019, 187-6.

(b)(3)

secretien

V-54

LA-14066-H

INCLASSIFIED

)

It was noted that a full-scale SRAM test would be scheduled for FY 1967 if the Phase 3 were authorized.²⁷⁸ (b)(3)

Improved Pershing

In addition, the Laboratory weapon-development staff suggested that a warhead similar in design to the one developed for the SRAM might be used as the warhead in the Improved Pershing, a system designed to fill the proposed Quick Reaction Alert role in Europe.

(b)(3)

Alternate Lance

The U.S. Army's Lance missile was a surface-launched guided missile. It was reported that the Laboratory was continuing work on an alternate Lance proposal

(b)(3)

Halberd

Poseidon

(b)(3)

(b)(3)

Primary

(b)(3)

²⁷⁸ LASL Program for Fiscal Years 1967-1968," DIR-2029 (SRD) (no date), p. 32.

(b)(3)

LA-14066-H

VARE NCLASSIFIED

(b)(3)

Improved Small Atomic Demolition Munition (SADM)

(b)(3)

e. Projectiles

175-mm and 8-in. Projectiles

(b)(3)

It appeared that a redesigned 8-in. shell with the same ballistic characteristics as the conventional 8-in. shell or a rocket-assisted 8-in. shell could also be developed.

There had been a lack of a specific interest by the Army for either the 175-mm or 8-in. shell projectiles. The Laboratory's yearly report noted that the Los Alamos groups had reduced their effort on these types of designs to "essentially zero."²⁸⁷

155-mm Projectile

(b)(3)

""LASL Program for Fiscal Years 1967-1968," DIR-2029 (SRD) (no date), p. 33.

0C- V

(b)(3)

SP(RET/RU

LA-14066-H

UNCLASSIFIED

3. 1967

a.

(b)(3)

Spitfire (Spartan)

By 1967, Spitfire had evolved into a program called Spartan. A report dated February 7-8, 1967, reported, "The present LASL SPITFIRE Test Program is directed toward developing a warhead for the SPARTAN Missile. To shorten the warhead development time scale, the tests are being performed in as near a weaponized configuration as possible."²⁹⁰ (This program would later become part of the Safeguard program described in a following section.)

(b)(3)

A memo from Jane Hall noted that during the October 25, 1967, WLPC meeting Bradbury and Agnew had reported on the October 23, 1967, Spartan meeting that had been held in Washington. The two had reported that General Giller (Assistant General Manager for Military Application) had indicated that he expected LASL and LRL to propose a pie-split in the Spartan program.

(b)(3)

²⁹⁰"Minutes of the First Meeting Spartan Ad Hoc Interface Working Group," (SRD) (February 7-8, 1967), p. 13, A99-019, 227-18.

(b)(3)

ASSIFIED

V-57

LA-14066-H

ARGEFTIND

A Sandia paper dated November 16,1967, noted that a follow-on Spartan program was also being considered.

(b)(3)

The Sandia paper stated, "This Follow-on Spartan will supplement, but not replace the Spartan."²⁹⁵

It appears that by December 1967, the management at the Laboratory was beginning to consider the fact that the LASL would probably be assigned the Sprint program and Livermore would receive the Spartan.

Sea-Based Ballistic Missile Intercept System

Also under consideration at that time was the design of a warhead for the Sea-Based Ballistic Missile Intercept System (SABMIS).²⁹⁷

Scampi

(b)(3)

""Follow-On Spartan," Sandia report RS 5624/52 (SRD) (November 16, 1967), p. 1, A99-019, 227-18.

(b)(3)

²⁹⁷"LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 21, 1967), p. 19.

(b)(3)

SFICKI/BAD

"""LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 21, 1967), pp. 19-20.

V-58

LA-14066-H

UNCLASSIFIED

JNCLASSIFIED

This effort will be discussed in the final

chapter of this report. 302

(b)(3)

<u>Walleye</u>

There was a Phase 2 study of a warhead for the Walleve glide bomb.

(b)(3)

<u>SAM-D</u>

A Phase 2 study for a primarily mobile field Army air defensive system was expected. This was to be called the SAM-D missile.³⁰³

e. Projectiles and Earth-Penetration Weapons

175-mm and 8-in. Shells

(b)(3)

However, it was reported, "...lack of specific interest by the Army has reduced this activity to essentially zero. A hardware program which uses 8-in. projectiles in earth penetrating weapons is being done in collaboration with the Sandia Corp."³⁰⁴

<u>155 mm</u>

There continued to be interest in the 155-mm projectile program.³⁰⁵

(b)(3)

After evaluation of the data, it appeared that further engineering

improvements were needed."

³⁰²"LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 21, 1967), pp. 20-21.

³⁰³"LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 21, 1967), p. 21.

³⁰⁴"LASL Program for Fiscal Years 1968-1969," DIR-2081 (SRD) (May 21, 1967), p. 23.

³⁰⁵ W-Division Quarterly Status Report, April 1, 1967 through June 30, 1967, Part 2 of Two," W-2035 (SRD) (July 17, 1967), pp. 1-15, A86-016, 242-2.__

(b)(3)

³⁰⁷ "W-Division Quarterly Status Report, October 1, 1967 through December 31, 1967, Part 2 of Two," W-2084 (SRD) (January 15, 1968), pp. 17–20, A86-016, 242-4.

LA-14066-H



(b)(3)

Bayonet

Reported in the third quarter report for 1967 from W-Division was a proposed program called Bayonet. The report states, "Bayonet is a program to determine the feasibility of using a nuclear warhead in an air-dropped weapon which can be exploded after the weapon has impacted and penetrated the ground."/The warhead compartment had a diameter of approximately 6.5 inches. The report from W-Division also indicates that the first Bayonet warhead had been drop-tested at Tonopah.³⁰⁹

The fourth-quarter report from W-Division notes that the second complete Bayonet assembly had, in October 1967, been drop-tested at the Tonopah Test Range. The missile had not penetrated properly, and the warhead had been damaged. It was stated that Sandia Corporation would correct the missile deficiencies before additional drop tests were made. However, based on the available data, it appeared that if the missile survived, the warhead would survive and would be able to function after impact.³¹⁰

f. Nonnuclear Kill

(b)(3)

4. 1968

(b)(3)

The detense program aimed at early destruction of an incoming missile carrying a nuclear warhead was now called Spartan. The Spartan missile was to be deployed on a trajectory that intercepted the enemy's incoming missile above the atmosphere

(b)(3)

³¹⁰"W-Division Quarterly Status Report, October 1, 1967 through December 31, 1967, Part 2 of Two," W-2084 (SRD) (January 15, 1968), pp. 13–15, A86-016, 242-4.

(b)(3)

ET/R/D

LA-14066-H

UNCLASSIFIED

The Spartan and Sprint missiles together formed what was by this time period being called the Sentinel program. The Spartan was the device that was to be the initial high-altitude defense for destroying the incoming missile. The Sprint was to be the low-altitude backup device.

SECREMEN

ICLASSIFIED

The latter two tests are

In January 1968, the responsibility for continued Phase 3 development of the Sprint warhead was transferred from Livermore to Los Alamos. This warhead was designated XW66. The primary responsibility for development of the Spartan warhead was assigned to Livermore. This warhead was designated XW71. However, the Los Alamos Laboratory was given a backup role for Spartan. The assignment memo from Assistant General Manager for Military Application Brig. Gen. Giller stated, "Los Alamos Scientific Laboratory/Sandia Corporation-Sandia Laboratory are to continue development of the SPITFIRE device as a backup warhead with warhead characteristics and schedules compatible with the SPARTAN program."³¹²

(b)(3)

included in the following paragraphs.]

(b)(3)

³¹²Brigadier General Edward B. Giller, USAF, Assistant General Manager for Military Application, Headquarter, to Those listed below, Subject: "Laboratory Assignments for Development of Sentinel Warheads," (CRD) (January 22, 1968), 2 pp., A99-019, 1980-9. "LASL Program for Fiscal Years 1969-1970," DIR-2143 (SRD) (October 1, 1968), p. 12. ____

(b)(3)

SFAAREA/BAD/

V-61

INCLASSIFIED

LA-14066-H



been obtained from this experiment."

It was noted that very favorable results had

(b)(3)

Advanced Spartan

An Advanced Spartan was also being considered in 1968. The third-quarter W-Division status report notes, "The first meeting of the Phase one AEC/DOD Advanced ABM Coordinating Group was held at AEC/DMA.

(b)(3)

³¹⁸"W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), pp. 39–42, A86-016, 242-6.

(b)(3)

SECRET/BADI

V-62

١

LA-14066-H

NULASSIFIED

ASSIFIED

Safeguard: Follow-On Sprint

The second quarter report from W-Division notes that a joint SLA/LASL document was being prepared that would summarize the Follow-On Sprint programs at the two Laboratories.³²⁸ The W-Division quarterly report for July 1 through September 30, 1968, announced that a LASL/SLA program for the Follow-On Sprint was being studied. Design layouts were being prepared for two warhead proposals for the Upstage II interceptor that was a Follow-On Sprint variation.

(b)(3)

³²⁸"W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 11, A86-016, 242-6.

(b)(3)

secret to

UNCLASSIFILD

LA-14066-H

V-63

SEARCE MAD

The final quarter report from W-Division indicates that calculations had been completed on the hardness of preliminary primary designs for the Follow-On Sprint.³³/

(b)(3)

However, at the same time, there were apparently signs that the Air Force might be losing interest in the project.³³⁵

SABMIS (Sea-Based Ballistic Missile Intercept System)

During the spring a meeting was held between representatives from the Navy and the AEC laboratories to review the SABMIS program and the Phase 1 document that had been prepared. While a Phase 2 status for the SABMIS was turned down, further system studies were authorized and were reported as underway. It was noted that a Phase 2 would again be requested the next year. It was reported, "Meanwhile communications will continue between NRDL (Naval Radiological Defense Laboratory) and the AEC laboratories,"³³⁶

AIM (Air Interceptor Missile)

(b)(3)

³³⁰ "W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), p. 11, A86-016, 242-8.

(b)(3)

³³³⁴ Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 8.

³³⁶"Program Status Weapons Research and Development, April-June 1968, Part 2 of Two," DIR-2133 (SRD) (no date), p. 8. "W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 13, A86-016, 242-6.

SACREMAN

LA-14066-H

NCLASSIFIED

UNCLASSIFIED MARCAVAN

Condor

Condor was a television-guided, rocket-boosted glide bomb that was sponsored by the Navy.

The first quarter of 1968 report from W-Division notes that LASL and the Sandia Corporation were cooperating in a Condor Phase 2 study.³³⁹ The second quarter report states that the Phase 2 feasibility study meeting had been held on May 14, 1968. Phase 3 manpower and material cost estimates on each of the submitted proposals had been sent to ALO.³⁴⁰

As the reader will recall, Phase 3 for the Navy's Condor was given to Los Alamos in 1969. This warhead carried the nomenclature XW73.

SCAM (Subsonic Cruise Attack Missile

The SRAM (Short Range Attack Missile) had gone into Phase 3 on January 1967 as the XW69. However, a Subsonic Cruise Attack Missile (SCAM) was also desired. There were discussions among the LASL personnel concerning suitable warhead proposals for the SCAM. The first quarter 1968 report from W-Division states, "A data package containing parametric material relative to weight and length as a function of yield has been submitted to AFWL [Air Force Weapons Laboratory] for transmission to their contractors. All data presented were based on present-day technology."³⁴¹

The third quarter 1968 report from W-Division notes that the contractor's four-month study effort on the SCAM feasibility had ended July 31, 1968, with the submission of final reports. It had been concluded that it would be feasible to provide an approximately 800-mile range missile using present state-of-the-art technology,

(b)(3) 'It was reported that LASL was continuing liaison with the Aeronautical Systems Division to provide inputs and guidance as requested.³⁴²

(b)(3)

³³⁹ W-Division Quarterly Status Report, January 1, 1968 through March 31, 1968, Part 2 of Two," W-2104 (SRD) (April 18, 1968), p. 4, A86-016, 242-5.

³⁴⁰ W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 11, A86-016, 242-6.

³⁴¹"W-Division Quarterly Status Report, January 1, 1968 through March 31, 1968, Part 2 of Two," W-2104 (SRD) (April 15, 1968), p. 7, A86-016, 242-5.

³⁴² W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), p. 17, A86-016, 242-7.

LA-14066-H

SHARFIMA UNCLASSIFIED

SCAD (Subsonic Cruise Armed Decoy)

(b)(3)

Meetings were being held in preparation for the Phase 2 meeting.³⁴³ It was reported that drafts of the General Requirements for SCAD and the Phase 1 data package had been received from the Air Force Weapons Laboratory.³⁴⁴

MRV (Maneuvering Reentry Vehicle) and ARV (Advanced MRV)

The second quarter W-Division report notes, "Aerospace Corporation briefed the weapons laboratories on the status of the MRV program and requested preliminary warhead designs."³⁴⁵

A joint SLA/LASL data package was prepared for the Maneuvering Reentry Vehicle (MRV) program, (b)(3)

By the end of the year, the program had become the Advanced MRV (ARV). The working group for the ARV had been organized in a meeting on October 3, 1968; the first meeting of this working group had been held on December 2, 1968.

(b)(3) However, despite this interest in the program, it was noted that AKV prototype development contracts were going to be delayed a year or more.³⁴⁷

MARS

There was also consideration of a system called MARS. This was a system to be mounted on armored vehicles.

(b)(3)

BDM (Bomber Defense Missile) and DPM (Dual-Purpose Missile)

Also in this long "want" list from the Military was the Air Force interest in a Bomber Defense Missile (BDM) and a Dual-Purpose Missile (DPM). The Dual-Purpose Missile was to be a ramjet-propelled missile with a velocity of Mach 4 to 4.5 and a range of 300 miles.³⁴⁹

³⁴⁵"W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 13, A86-016, 242-6.

(b)(3)

³⁴⁷ "W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), p. 12, A86-016, 242-8. "Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 8.

(b)(3)

³⁴⁹"Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 8.

V-66



LA-14066-H

UNULASSICIED

³⁴³"Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 9.

³⁴⁴ "W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), pp. 16–17, A86-016, 242-8.
ICLASSIFIED

(b)(3)

/ Los Alamos supplied the Air Force Weapons Laboratory with a list of 20 possible warhead candidates for these applications.³⁵⁰

By the fourth quarter of 1968, meetings had been held with representatives of the Office of Research Analysis and AFWI.

(b)(3)

LAR (Low-Angle Reentry Vehicle)

The first quarter 1968 report from W-Division notes that a technical data package on a LAR (Low-Angle Reentry Vehicle) had been submitted to AFWL for submission to Air Force contractors:

(b)(3)

During the spring, meetings were held with DOD agencies and contractors to discuss the warhead proposals contained in the LASL Phase 1 data-package.³⁵³

(b)(3)

I ne final quarter report for the year from W-Division states, "A DoD contractor has completed the evaluation and pre-design study of the Low Angle Reentry Vehicle concept.....

(b)(3)

Primaries

(b)(3)

³⁵⁰ W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), p. 12, A86-016, 242-7.

(b)(3)

W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 11, A86-016, 242-6.

(b)(3)

APARELIAN INCLASSIFICU

LA-14066-H

ADM (Atomic Demolition Munition) and ADAM (Advanced Atomic Demolition Munition There was interest in an Atomic Demolition Munition (ADM) for the Army.

(b)(3)

The second quarter 1968 report from W-Division notes that the second meeting concerning a possible ADM had been held on May 21–23.³⁵⁹ LASL proposals were submitted in response to the requirements that had been outlined at the first meeting.

(b)(3)

A nonnuclear self-destruct

was to be included.)³⁰⁰ The third quarter 1968 report from W-Division notes that the ADM proposals were being updated.³⁶¹

During the latter part of 1968, the Army-approved requirements for an Advanced Atomic Demolition Munition (ADAM) were received at Los Alamos.³⁶²

(b)(3)

The final quarter 1968 report from W-Division states that a Phase 2-type data package was being prepared.³⁶⁴

<u>Walleye</u>

The first quarter report from W-Division states, "LASL is currently working with ALOO to re-do the Walleye Phase 2 cost data study prepared in January 1967. DMA is requesting detailed laboratory manpower and material cost estimates for these proposals and on all future Phase 2 studies."³⁶⁵ [The reader will recall that the Walleye was an electro-optical guided glide bomb. It was designed for use by the Air Force's F-4D aircraft.]

It was felt that a Phase 3 development program for the Walleye might be authorized in FY 1969.³⁶⁶ [The reader will recall that the Phase 3 was assigned in May 1969. The warhead would be given the nomenclature XW72.]

(b)(3)

³⁵⁹"W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), p. 10, A86-016, 242-6.

³⁶⁰ Program Status Weapons Research and Development, April-June 1968, Part 2 of Two," DIR-2133 (SRD) (no date), p. 7.

³⁶¹"W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), p. 11, A86-016, 242-7.

³⁶²"Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), n. 8.

(b)(3)

³⁶⁴ W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), p. 11, A86-016, 242-8.

³⁶⁵ "W-Division Quarterly Status Report, January 1, 1968 through March 31, 1968, Part 2 of Two," W-2104 (SRD) (April 15, 1968), p. 8, A86-016, 242-5.

SECRET/BAX

³⁶⁶"LASL Program for Fiscal Years 1969-1970," DIR-2143 (SRD) (October 1, 1968), p. 14.

V-68

LA-14066-H

UNCLASSIFIED

Sparrow

There were also discussions with the Air Force Weapons Laboratory on providing a warhead for what was called an Advanced Sparrow missile.³⁶⁷

d. Projectiles and Earth-Penetrating Weapons

Bayonet

The program called Bayonet continued in 1968. The concept at this time was for this device to include a shaped charge of HE that was designed to detonate when the missile was a few feet from the ground. The ensuing jet was supposed to penetrate the earth, and in theory, enable an easier entry for the nose of the missile. The problem with the concept was how to penetrate hard rock.³⁶⁸

The second quarter W-Division report indicates that further work and redesign of the system had taken place in the reporting period of April 1–June 30, 1968. The report notes, "Except for the additional tests of the redesigned internal ballistic system, the planned Bayonet feasibility program has been completed."³⁶⁹

(b)(3)

LA-14066-H

(b)(3)

SECRETIRD UNCLASSIFIED

³⁶⁷"Program Status Weapons Research and Development, April-June 1968, Part 2 of Two," DIR-2133 (SRD) (no date), p. 8. "W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), p. 11, A86-016, 242-7.

³⁶⁸ Program Status Weapons Research and Development, July-September 1968, Part 2 of Two," DIR-2142 (SRD) (no date), p. 9.

³⁶⁹"W-Division Quarterly Status Report, April 1, 1968 through June 30, 1968, Part 2 of Two," W-2128 (SRD) (July 15, 1968), pp. 14–15, A86-016, 242-6.



155-mm

(b)(3)

8-inch

An in-house meeting was held in Albuquerque in November for the purpose of considering a new 8-inch artillery-fired atomic projectile (AFAP). The Phase 2 meeting was held in December. It was reported that it had been concluded that a new nuclear warhead was feasible.³⁷⁵

e. Nonnuclear Kill

(b)(3)

^{3/3}"Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 9. "W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), p. 16, A86-016, 242-8.

> (b)(3) SECTEDITION UNULASSIFIED

LA-14066-H

Steel Rods

There was also interest by the Air Force in a technique of nonnuclear kill of reentry vehicles using high-velocity (4,000–6,000 m/s) 50-g steel rods. These would be accelerated using a nuclear detonation. It was hoped that retired W59 warheads could be modified for use in the program.³⁷⁷

The year-end quarterly report from the Laboratory notes that two W59 warheads were being modified for tests.³⁷⁸ The year-end quarterly report from W-Division states, "The extent of LASL participation in this program is to review the test and packaging procedures for nuclear safety hazards and post-test assessment of warhead damage."³⁷⁹

5. 1969

Backup Spartan-

In a letter dated January 28, 1969, from the AEC, the Laboratory was informed, "This office has been informally advised by DMA that the LASL Spartan backup device has been deleted from the STS program." The letter added, "Accordingly, it is requested that you examine your STS support requirements in order to gain an early appreciation of those items that might be affected by this change."³⁸⁰

(b)(3)

(b)(3)

(b)(3)

V-71 NCLASSIFICD

³⁷⁷"Program Status Weapons Research and Development, April-June 1968, Part 2 of Two," DIR-2133 (SRD) (no date), p. 8. "W-Division Quarterly Status Report, July 1, 1968 through September 30, 1968, Part 2 of Two," W-2145 (SRD) (October 15, 1968), p. 12, A86-016, 242-7.

³⁷⁸"Program Status Weapons Research and Development, October-December 1968, Part 2 of Two," DIR-2156 (SRD) (no date), p. 8.

³⁷⁹ W-Division Quarterly Status Report, October 1, 1968 through December 31, 1968, Part 2 of Two," W-2164 (SRD) (January 15, 1969), pp. 11–12, A86-016, 242-8.

³⁸⁰Robert E. Miller to W. D. Smith, Jr. et. al., Subject: "STS Program Adjustment," (CRD) (January 28, 1969), 1 p., A99-019, 218-4._____



(b)(3)

Improved Spartan

(b)(3)

The LASL warhead proposals for an Improved Spartan program were outlined in detail in an April 21, 1969, paper.³⁸⁵

(b)(3)

However, the W-Division report for April-June notes that it appeared that the Phase 2 study was going to be delayed.³⁸⁶

The Laboratory's quarterly report of July-September 1969 indicates that while the final Phase 3 Spartan missile warhead development for the Safeguard program had gone to Livermore as the W71, the Los Alamos teams were continuing to work on the development of an Improved Spartan-a follow-on program to the Spartan.

It was expected that (b)(3) the Phase I data package for the Improved Spartan would be forwarded to the DDR&E (Director, Defense Research and Engineering) around October 1,³⁸⁷ The W-Division report for July-September 1969 indicates that in July the Safeguard System Command had published a

(b)(3)

385"LASL Warhead Proposals for Improved Spartan Program," W-9-390 (SRD) (April 21, 1969), 5 pp., A99-019, 218-4.

³⁸⁶"W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 21, A86-016, 242-10. "Program Status Weapons Research and Development, April -- June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 11.

³⁸⁷"Program Status Weapons Research and Development, July - September 1969 (U)," DIR-2187 (SRD) (no date), pp. 46-47, 61.

V-72

LANGLASSIFIED

schedule for the development of a warhead section for the Improved Spartan. The Phase 1 report was to be completed by September 1969 and the Phase 2 by January 1970. However, it appeared, that this schedule was slipping.³⁸⁸

(b)(3)

cited reference.³⁹¹

⁷Additional information is available in the

SABMIS (Sea-Based Ballistic Missile Intercept System)

A January 9,1969, TWX from the Assistant General Manager for Military Application, Edward B. Giller, reported that the DOD was studying the SABMIS concept in order to add depth to the defenses of the continental U.S. In addition, SABMIS had the advantage that it would be a mobile system that would be available for the defense of an overseas area; nuclear weapons would not have to be deployed ashore.

(b)(3)

³⁸⁸ W-Division Quarterly Status Report, July 1, 1969 through September 30, 1969," W-2217 (SRD) (October 15, 1969), p. 14, <u>A86-016</u>, 242-11.

(b)(3)

³⁹¹ "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969," W-2198 (SRD) (July 15, 1969), pp. 50-58, A86-016, 242-10.

LA-14066-H

(b)(3) HABA VIKI ULASSIFILD

UNCLASSIFIED ARAAAAA

(b)(3)

Attack Missile (LAPAM)

There was also interest early in 1969 in a Low Altitude Penetration Attack Missile (LAPAM). It was to be a supersonic, long-range standoff missile targeted for interceptor airfields and ABM tracking radars. The nuclear weapon payload was to include a bomb, elected on the first target, and a warhead, designed for the second target.

(b)(3) [It was reported that the Los Alamos/Sandia weapon groups had prepared a pre-Phase 1 data package.³⁹⁴

Mark 19 and Mk 12A (Warhead for Minuteman).

(b)(3)

The second quarter report noted that a data package, covering the two options, had been sent to several Air Force contractors.³⁹⁵

(b)(3)

The primary had yet to be selected. 390

(b)(3)

The planning information domment for the Mk 19 was released in October 1969.

(b)(3)

³⁹⁴ "Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), pp. 19–20, A86-016, 242-9.

³⁹⁵"Program Status Weapons Research and Development, April – June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), pp. 21–22, A86-016, 242-10.

³⁹⁶"Program Status Weapons Research and Development, July - September 1969 (U)," DIR-2187 (SRD) (no date), p. 45. ___

(b)(3)

SPARATURA UNCLASSIFIED

(b)(3)

The final W-Division quarterly report for 1969 states that the Phase 1 data-package for the Mk 19 had been reviewed on October 16, 1969, during a meeting at the Air Force Weapons Laboratory. It appeared that the program would go to Phase 2. The Air Force Scientific Advisory Board had received a review of the LASL program

(b)(3)

[Author's note: The Mk 19 program would be the precursor to the XW78.]

New Full Fusing Option (FUFO) Bomb

(b)(3)

In a Phase I study, the weapon laboratories had outlined the possibilities for this type of bomb in terms of yield, size, and weight.⁴⁰¹

(b)(3)

LORAH (Long Range Area Homer)

During the first quarter of 1969, it was reported that preliminary studies had been conducted by the Advanced Ballistic Missile Defense Agency and its contractors on future missile defense systems. A system-concept known as Long-Range Area Homer (LORAH) had been selected for more detailed study.⁴⁰³

(b)(3)

⁴⁰¹"W-Division Quarterly Status Report, October 1, 1969 through December 31, 1969," W-2235 (SRD) (January 15, 1970), p. 42, A86-016, 242-12.

(b)(3)

ICLASSIFIED

V-75

⁴⁰³"Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 12. "W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 22, A86-016, 242-9.

During the second quarter, Los Alamos personnel supplied basic warhead data to attendees at a joint AEC/DOD meeting.⁴⁰⁴ The second quarter 1969 W-Division report states for the LORAH system, "Initial design parameters of the proposed missile delivery system indicate that a small nuclear yield will meet the target kill requirements if the small miss distance ($\approx 30-50$ ft versus the 50–100 ft previously stated as a design goal) can be provided by the guidance/homer system now under study for the LORAH system."⁴⁰⁵

- The fourth quarter W-Division report for 1969 notes that in a three-day meeting held during the first part of November, the studies on LORAH, completed by three different contractors, had been presented to the Advanced Ballistic Missile Defense Agency. The results of the three studies had been almost identical. The interceptor would carry 4 to 6 vehicles to an intercept point after which an interceptor-borne sensor would identify targets. As targets were identified, a homing sensor would lock on and would "zero in on the target." The high accuracy of hitting the target would permit the use of very low-yield warheads. It was noted, "The LORAH concept is under study for the 1980s to operate in conjunction with the SAFEGUARD defense system."⁴⁰⁶

Bomber Defense Missile, Dual-Purpose Missile

During January–March 1969, discussions were continued with personnel from the Air Force Weapons Laboratory and their contractors on warhead proposals for the Bomber Defense Missile (BDM) and the Dual Purpose Missile (DPM). It was noted that yields of 1- to 10-kt were of interest for the BDM; yields of 5- to 200-kt were of interest for the DPM.⁴⁰⁷

During the second quarter of 1969, a joint LASL/SLA information document (as input for a study of possible warheads for the DPM) was submitted to the AFWL.

(b)(3)

The Laboratory status report for July-September 1969 notes that the final report on the DPM had been completed by the Office of Research Analysis of the U.S. Air Force. The W-Division quarterly report notes, "LASL has received volumes 1, 2, 3, 4, 6 and 9 of this extensive report. This report recommends further investigation on air-breathing propulsion, missile guidance, and low radar cross sections."

(b)(3)

(b)(3)

LA-14066-H

LASSIFIED

⁴⁰⁴ "Program Status Weapons Research and Development, April – June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 10.

⁴⁰⁵"W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 19, A86-016, 242-10.

⁴⁰⁶ W-Division Quarterly Status Report, October 1, 1969 through December 31, 1969," W-2235 (SRD) (January 15, 1970), p. 37, A86-016, 242-12.

⁴⁰⁷"Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 19, A86-016, 242-9_

UNCLASSIFIED

The fourth quarter 1969 report from W-Division states that the Air Force had approved a funded design study for a Short Range Bomber Defense Missile. The Los Alamos weapon groups in support of this study had sent warhead information to the Air Force Weapons Laboratory.⁴¹⁰

SCAD (Subsonic Cruise Attack Decoy)

Early in 1969, preparations continued for an anticipated AEC/DOD Phase 2 meeting on the Subsonic Cruise Attack Decoy (SCAD).

(b)(3)

The July-September-quarterly reports prepared by the Laboratory groups announced that three contractors were making further feasibility studies, to be <u>completed by September 30</u>, 1969, on the SCAD proposal

(b)(3)

It was stated, "Studies have been made to determine whether a narticular combination will give a significant range advantage to the SCAD.

(b)(3)

ARV (Advanced MRV)

It was announced that the Air Force had awarded identical ARV contracts effective May 1, 1969 to two contractors. These were for the purpose of evaluating flight control system concepts for Terminal Evasion/Accuracy and for Terminal-Evasion-only MRVs. [The ARV program had been extended to include studies of a Simple Terminal Evasion (STE) vehicle as well as the Terminal Evasion/Accuracy vehicle.]

(b)(3)

⁴¹⁰ W-Division Quarterly Status Report, October 1, 1969 through December 31, 1969," W-2235 (SRD) (January 15, 1970), p. 40, A86-016, 242-12.

(b)(3)

LA-14066-H



The third quarter report from Los Alamos notes that in the planning considerations, the Panther A primary had been used in the baseline design for the Terminal Evasion/Accuracy MRV. Two technical exchange meetings had been held in August.⁴¹⁵

The final quarter report for 1969 from W-Division indicates that, in support of the ARV program, personnel from LASL had attended the Technical Direction meetings with the Air Force and their two contractors. The present ARV study program, with the primary goal of evaluating flight control systems, was scheduled for completion in February 1970.⁴¹⁶

ASMA (Advanced Surface Missile)

During the second quarter of 1969, it was reported that an Advanced Surface Missile System (ASMA) was under consideration (b)(3)

Warhead, 417 The nuclear warhead section was to be interchangeable with the HE

Terminal Homing Vehicle

During the second quarter of 1969, it was also announced that the Air Force had started a program to evaluate the possibility of using the Minuteman in an offensive/defensive role. In the Terminal Homing Vehicle concept, the payload of the Minuteman would become a homing vehicle able to "home in on an incoming reentry vehicle." The vehicle would incorporate either a nonnuclear or nuclear kill warhead. The status reports from the Laboratory indicate that a joint LASL/SLA nuclear warhead data package had been prepared.⁴¹⁸

LAR (Low-Angle Entry Vehicle)

In terms of the Low-Angle Entry Vehicle (LAR) it was reported in the W-Division third quarter report that an 18-month contract had been awarded for the conduct of the Feasibility Flight Test Program.⁴¹⁹ The July–September status report from the Laboratory notes, "...we do not expect to have discussions concerning the warhead until the end of the program; that is when the actual flight tests have been conducted."⁴²⁰

⁴¹⁸"Program Status Weapons Research and Development, April – June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 21, A86-016, 242-10.

SEARTHED LASSIFIED

V-78

⁴¹⁵"Program Status Weapons Research and Development, July - September 1969 (U)," DIR-2187 (SRD) (no date), pp. 9–10.

⁴¹⁶ W-Division Quarterly Status Report, October 1, 1969 through December 31, 1969," W-2235 (SRD) (January 15, 1970), p. 39, A86-016, 242-12.

⁴¹⁷"Program Status Weapons Research and Development, April – June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 20, A86-016, 242-10.

⁴¹⁹ "W-Division Quarterly Status Report, July 1, 1969 through September 30, 1969," W-2217 (SRD) (October 15, 1969), p. 15, A86-016, 242-11.

⁴²⁰ Program Status Weapons Research and Development, July - September 1969 (U)," DIR-2187 (SRD) (no date), p. 10.

UNCLASSIFIED

SAM-D (Surface-to-Air Missile-D)

The Army planned to replace the Nike Hercules and Hawk with a Surface-to-Air Missile-D (SAM-D).

(b)(3)

Air Interceptor Missile (AIM)

The final quarter 1969 report from W-Division notes that the Navy had submitted a request for an AEC Phase 2 feasibility study on possible warhead candidates for the Phoenix missile.⁴²² [Additional information on Phoenix is given in the 1970 section.]

(b)(3)

High-Yield Source

(b)(3)

ADAM (Advanced Atomic Demolition Munition)

During the first quarter of 1969, the Laboratory's Phase 2 input for the Advanced Atomic Demolition Munition (ADAM) study was coordinated within the Laboratory. This study was to be submitted early in April.⁴²⁵

(b)(3)

⁴²² "W-Division Quarterly Status Report, October 1, 1969 through December 31, 1969," W-2235 (SRD) (January 15, 1970). n. 38. A86-016. 242-12.

(b)(3)

^{4.25} Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 10. "W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 17, A86-016, 242-9.

LA-14066-H

RCREAMBAU UNCLASSIFIED



The Laboratory quarterly report for July-September 1969 notes that the ADAM Phase 2 feasibility study had been published by the Army. The AEC Impact and Capabilities study had been published by the Albuquerque Operations Office (ALO). 426

Advanced Sparrow

During the first quarter of 1969, there continued to be interest in what was called the Advanced Sparrow missile. The warhead was to weigh about 90 lb and have an 8-inch diameter The warhead was to be a and a length of 17.5 inches (b)(3) direct replacement of the conventional HE warhead.

(b)(3)

The first quarter report from W-Division notes that a joint LASL/SLA input had been submitted to AFWL for a Phase 1 feasibility study for a nuclear capability for the Advanced Sparrow missile.428

Walleye Phase 3

The Phase 3 for the Walleye missile warhead (BA72) was awarded in May 1969. It was noted that development would be mainly a Sandia activity.⁴²⁹

New Implosion System-

(b)(3)

d. Projectiles and Earth-Penetrating Devices

Eight-Inch Artillery-Fired Atomic Projectile (AFAP)

During the first quarter of 1969, it was reported that the Phase 2 feasibility study and the additional Impact and Capability report for a new 8-inch nuclear artillery round was nearing completion/ (b)(3)

426. Program Status Weapons Research and Development, July - September 1969 (U)," DIR-2187 (SRD) (no date), p. 48.

(b)(3)

428. W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 17, A86-016, 242-9.

⁴²⁹ Program Status Weapons Research and Development, April - June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 16.

(b)(3)

SEATAT RA

LA-14066-H

UNCLASSIFICU

The Phase 2 study for the 8-inch AFAP was completed during the second quarter and the applicable reports were issued.433

LASSIFIED

155-mm

During the first quarter of 1969, the Phase 2 feasibility study for a new 155-mm nuclear round for the Army was reported to be nearing completion. The Los Alamos design group had proposed two designs.

(b)(3)

Ine April-June quarterly report announced that the Phase 2, 155-mm study had been completed; the Phase 2 meeting had been held in February 1969. The Phase 2 report was dated May 2, 1969; the Impact and Capabilities Study was dated April 18, 1969.435

(b)(3)

Engineering studies,

gun-tiring tests, and materials-development activities were in progress or planned.⁴³⁸

Bavonet

The first part of 1969 saw completion of the engineering tests for the Bayonet feasibility program.439

⁴³³"Program Status Weapons Research and Development, April – June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), p. 11. "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 19, A86-016, 242-10.

(b)(3)

⁴³³"Program Status Weapons Research and Development, April - June 1969, Part 2 of Two (U)," DIR-2180 (SRD) (no date), pp. 10-11. "W-Division Quarterly Status Report, April 1, 1969 through June 30, 1969, Part 2 of Two," W-2199 (SRD) (July 15, 1969), p. 19, A86-016, 242-10.

(b)(3)

^{a,38}"Program Status Weapons Research and Development, October-December 1969 (U)," DIR-2195 (SRD) (no date),

pp. 65-66. ⁴³⁹"Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 12. "W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 24, A86-016, 242-9.

LA-14066-H



e. Nonnuclear Kill

For the program involving the acceleration of steel as a kill mechanism, the Laboratory's 1969 first-quarter report states that the W59 warheads had been modified and were ready for delivery to the test facility.⁴⁴⁰

A meeting had been held to review and coordinate various phases of the hypervelocityprojectile tests on two modified W59 warheads in Mark-5 reentry vehicles. Two test shots were initially planned. One would use a 50-gram steel rod moving at 20,000 ft/s into the midsection of the W59. The other would use a 50-gram steel rod moving at 20,000 ft/s into the secondary of the W59.⁴⁴¹

6. 1970

(b)(3)

CAFE (C-3 Alternate Front End)

In a TWX dated May 11, 1970, it was noted that there might be a ban on the use of MIRV (Multiple Independent Reentry Vehicle) type deployment. DOD personnel were therefore reviewing capabilities for alternate strategic missile loadings. In particular, the Navy was concerned about a design that could be used as a single RB on Poseidon.⁴⁴²

The W-Division third quarter (July 1, 1970, through September 30, 1970) report announced that a design study for CAFE had been prepared. The CAFE (C-3 Alternate Front End) study was to identify the preferred design for a new reentry body payload for the Poseidon C3 missile. It was reported that two possible modifications to the W67 warhead had been included in this study. (b)(3)

The first CAFE Reentry Interchange Committee Meeting took place on July 15, 1970. An informal meeting was held between LASL/SLA and LMSC (Lockheed Missiles and Space Company) on July 22, 1970. In a submittal to C. E. Grant (through the Navy Plant Representative) at Lockheed Missiles and Space Company dated August 7, 1970, the Los Alamos and Sandia Albuquerque laboratories submitted a paper titled. "LASL/SLA Warhead Data for CAFE Study."

Another meeting on the CAFE program took place on September 18, 1970. During this meeting, a follow-on phase of the CAFE study was requested

(b)(3)

(b)(3)

LA-14000-11

UNCLASSIF.ED

⁴⁴⁰"Program Status Weapons Research and Development, January – March 1969, Part 2 of Two (U)," DIR-2172 (SRD) (no date), p. 10.

⁴⁴¹ W-Division Quarterly Status Report, January 1, 1969 through March 31, 1969, Part 2 of Two," W-2193 (SRD) (April 15, 1969), p. 18, A86-016, 242-9.

⁴⁴²USAEC, Thomas R. Clark, Wash., D.C. to AN1, USAEC, H. C. Donnelly, Albuq., N.M. et. al. (SRD) (May 11, 1970). 6 pp., A99-019, 188-4.

CLASSIFIED

The fourth quarter report from W-Division for 1970 notes that the Poseidon C3 Alternate Front End Study had been completed by Lockheed Missiles and Space Company. The purpose of the study had been to identify the preferred design for a new reentry body payload, called the Mk 1 Prime, for the Poseidon missile in the event that the Mk 3 MIRV system could not be deployed as planned.

(b)(3)

The total warhead weight was 680 lb.446

Improved Spartan

(b)(3)

Phase 6 for the W71 would not be achieved until October 1974. (The weapon would be placed in the inactive stockpile in October 1976.⁴⁴⁷)

In a February 24, 1970, letter to Chairman of the AEC, Glenn T. Seaborg, John S. Foster, Director of Defense Research and Engineering wrote, "The Department of Defense desires to determine the feasibility of a nuclear warhead for the Improved SPARTAN missile warhead section....The Improved SPARTAN missile is a sub-system of the SAFEGUARD ballistic missile defense system and is being designed to provide a long-range intercept capability in the exoatmosphere against ICBM and SLBM reentry vehicles (RVs), long-range intercept capability in the atmosphere against depressed trajectory RVs, a capability to intercept penetration-aided RVs, an endo- and exoatmospheric intercept capability against FOBS [Fractional Orbital Bombardment System], and an intercept capability against satellites (but only within missile performance limitations)." [Author's note: Wow, what an assignment!] Foster noted that the presently planned improved Spartan missile would use the Spartan missile's first and second stages while incorporating a new third stage(

(b)(3)

An effort through Phase 2 was requested for this improved

Spartan missile.448

(b)(3)

(The Laboratory's quarterly report for January-March 1970 states, "Though there are many system parameters as yet unresolved, the AEC has received a formal request to participate in the Phase 2 Feasibility study of warhead designs for the Modified Spartan missile system."

(b)(3)

⁴⁴⁶ "Program Status Weapons Research and Development, October-December 1970 (U)," Los Alamos Scientific Laboratory report LA-4614-MS (SRD) (February 1971), p. 42. "W-Division Quarterly Status Report, October 1, 1970 through December 31, 1970," W-2301 (SRD) (January 15, 1971), p. 66, A86-016, 242-15.
⁴⁴⁷ "FV 1004 Annual Warman Participation of the second status and the sec

447 "FY 1994 Annual Weapons Program Report," DOE Operation Office report (SRD) (October 1, 1994), pp. 318-322.

448 John S. Foster, Jr. to Honorable Glenn T. Seaborg (SRD) (February 24, 1970), 5 pp., A99-019, 186-10.

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED

(b)(3)

The document noted that the improved Spartan would incorporate a larger and more powerful third stage motor. This new motor would allow for greater missile maneuverability and therefore allow a reduction in warhead miss distance. It would also allow for a "loiter" capability. The warhead would need to be multipurpose.

(b)(3)

It was also noted that if the Safeguard system were extended to maximum deployment, it could be extended to include an area defense of the population against a light or

(b)(3)

ELA-14066-H

UNCLASSIFIED

irrational attack, or against accidental launches.

<u>Mk 19</u>

(b)(3)

(b)(3)

LA-14066-H

Juis si 14 6

V-85

ICLASSIFIED

(b)(3)

In a TWX dated May 11, 1970, it was noted that there might be a ban on the use of MIRV. DOD personnel were therefore reviewing capabilities for alternate strategic missile loads. Use of a Mk 19 type device in this application appeared to be a possibility.⁴⁶⁴

In May 1970, the Los Alamos staff released a study of the Mk 19 warhead.⁴⁶⁵ This paper appears to have been prepared as an "information document" for the Military. A letter dated July 24, 1970, from H. M. Agnew, Weapons Division Leader, noted, "The level of effort being expended at the LASL on the Mk 19 program is relatively high; however, the output generally is in the form of study proposals. The more attractive proposals are for systems currently in the very early conceptual stages where many design iterations remain to be undertaken."⁴⁶⁶

(b)(3)

⁴⁶⁴USAEC, Thomas R. Clark, Wash., D.C. to AN1, USAEC, H. C. Donnelly, Albuq., N.M. et. al. (SRD) (May 11, 1970), 6 pp., A99-019, 188-4.

FCRETTRIA

⁴⁶⁵ "Mk 19 Warhead Study," W-9-709 (SRD) (May 5, 1970), 19 pp., A99-019, 182-6.
⁴⁶⁶ H. M. Agnew to Mr. Don B. Shuster, W-2265 (SRD) (July 24, 1970), 1 pp., A99-019, 182-6.

LA-14066-H

UNCLASSIFIED

A TWX dated September 15, 1970, pointed out that the system cost for the Minuteman III and MIRV was 2.2 billion. The projected cost for the Mk 19 was approximately 130 million. Thus, the R&D cost for the Mk 19 was 5% of the total system costs.⁴⁶⁷

(b)(3)

<u>FUFO</u>

During 1970, the DOD guidance placed emphasis upon a new FUFO bomb

(b)(3)

<u>SABMIS</u>

There also continued to be an interest in a Sea-Based Anti-Ballistic Missile Intercept System. It was reported that development schedules for this system were such that tests of the warheads themselves would not be scheduled for FY 1971.

(b)(3)

⁴⁶⁷D. W. Bergen, Los Alamos Scientific Laboratory, Los Alamos, N.M. to C. I. Hudson, University of California, Lawrence Radiation Laboratory (SRD) (September 15, 1970), p. 3, A99-019, 188-4.

(b)(3)

LA-14066-H

AFAREINAN UNCLASSIFIEI

V-87

2

Stepheneter Aread

JNCLASSIFIED

b. Two-Stage, Lower-Yield Weapons

SCAD (Subsonic Cruise Armed Decoy)

Design efforts for the Subsonic Cruise Armed Decoy (SCAD) for use on the B-52 and FB-111 were also continuing in 1970/

(b)(3)

The final quarterly report for 1970 from the Laboratory notes that the SCAD was a subsonic turbojet-powered missile expected to be employed as a standoff missile in [nuclear] armed and unarmed [conventional] versions. It was believed that the unarmed version would be developed first. An updated Phase 1 data package was required before the Phase 2 could be initiated.

(b)(3)

Phoenix,

Phoenix was a Navy air-to-air missile. The Phoenix missile system had been in development for many years; by 1970, flight testing of the missile with a conventional warhead was essentially complete. It was reported, "The Phoenix is the prime armament for the F-14 advanced fighter aircraft in its role of defending the fleet against air threats."⁴⁷⁴

The request for a Phase 2 study to determine the feasibility of providing a nuclear capability in the Phoenix air-to-air missile was received at Los Alamos early in February 1970. The Phase 2 study meeting was scheduled to be held near the end of Max.

(b)(3)

IT was noted, "This commonality of consideration complicates what "Would have otherwise been a straightforward study."

The quarterly report for April–June 1970 from the Laboratory states that basic information on several proposed warhead designs had been provided for the Phoenix Phase 2 study.⁴⁷⁵ The W-Division quarterly report notes, "The proposed warheads included designs [for the Phoenix Phase 2 study] with particular advantages such as minimum use of special nuclear materials, no limited life components, low yield variation with age... Variations in the proposed designs to meet the requirements of other missile systems have also been provided."⁴⁷⁶

(b)(3)

^{*/**}Program Status Weapons Research and Development, January-March 1970 (U)," DIR-2203 (SRD) (no date); p. 53. "W-Division Quarterly Status Report, January 1, 1970 through March 31, 1970," W-2249 (SRD) (April 15, 1970), p. 15, A86-016, 242-12.

⁴⁷⁵"Program Status Weapons Research and Development, April-June 1970 (U)," DIR-2212 (SRD) (no date), p. 58.
 ⁴⁷⁶"W-Division Quarterly Status Report, April 1, 1970 through June 30, 1970," W-2264 (SRD) (July 15, 1970),

STATION CLASSIFIEL

LA-14066-H

p. 34, A86-016, 242-13.



ACE (Advanced Control Experiment)

Another program under consideration in 1970 was called ACE. These letters stood for Advanced Control Experiment. The purpose of the program was to develop and flight test a maneuvering reentry vehicle. It was noted that McDonnell-Douglas and General Electric were working for SAMSO (Space and Missile Systems Organization) on a predesign study to last through November 1970. The mission of the ACE warhead was "assured destruction." At that time, the concept was to use three, 350-lb RVs per Minuteman booster.

(b)(3)

Representatives from Los Alamos hadattended design meetings at McDonnell-Douglas, Huntington Beach, California, and at General Electric, Philadelphia, Pennsylvania. Additional information is available in the cited reference.⁴⁷⁷

ABC (Advanced Ballistic Concept)

The October-December quarterly report from the Laboratory indicates a program called the ABC (Advanced Ballistic Concept).

(b)(3)

In 1970, a joint operation agreement was reached between Los Alamos and SAMSO on the ABC program. From the starting date of June 29, 1970, the program was to have a 26-month study period. The Air Force Systems Command and SAMSO were responsible for program management. The Aerospace Corporation was responsible for the general systems engineering and technical integration. The AEC contractor was the AVCO Corporation.⁴⁷⁹

LAR (Low-Angle Reentry Vehicle)

The Low-Angle Reentry vehicle (LAR) was also during 1970 being considered for development.

LORAH (Long Range Area Homer)

In 1970 the Long Range Area Homer system was also being proposed for use with a nuclear warhead. It was felt that tests of the LORAH warhead would have to include diagnostic measurements of the fast neutron and high-energy x-ray output from the small, lightweight warhead. However, it was reported that development was not far enough along that tests needed to be included in the schedule for FY 1971.⁴⁸¹

⁴⁷⁷J. J. Jacoby, W-4 to R. G. Shreffler, Subject: "ACE Trip Report – First Design Review," W-4-2980 (SRD) (August 26, 1970), 6 pp., A99-019, 307-4.

(b)(3)

LA-14066-H

SPACEFERRE UNCLASSIFIED

^{4/9}"Joint Operating Agreement Between Los Alamos Scientific Laboratory and SAMSO on the ABC Program," (U) (no date), p. 1, Table I, film GAMF-1208. Los Alamos Records Center.

^{(b)(3)} ⁴⁸¹Charles I. Browne to Mr. W. R. Cooper, JOT-104-70 (SRD) (May 15, 1970), p. 4, A99-019.

UNCLASSIFIED SFRAAMAA/

Single Stage

(b)(3)

d. Projectiles and Earth-Penetrating Devices

<u>155-mm</u>

c.

Authorization to proceed with a Phase 3 development program for a 155 mm AFAP was received from DMA on February 24. The list of approved Military Characteristics and the Army Stockpile to Target Sequence were received in early March. The projectile's warhead was given an AEC nomenclature of W74.⁴⁸⁴

8-inch

(b)(3)

The July–September quarterly report from the Laboratory as well as the W-Division quarterly report state that as an addendum to the existing Phase 2 study for the improved 8-inch AFAP, two design proposals for one of the projectile configurations had been submitted

(b)(3)

The designs are shown in the cited

W-Division report.400

(b)(3)

⁴⁸⁴"W-Division Quarterly Status Report, January 1, 1970 through March 31, 1970," W-2249 (SRD) (April 15, 1970), p. 47, A86-016, 242-12.

(b)(3)

⁴⁸⁶"Program Status Weapons Research and Development, July-September 1970 (U)," DIR-2230 (SRD) (no date), p. 25. "W-Division Quarterly Status Report, July 1, 1970 through September 30, 1970," W-2282 (SRD) (October 15, 1970), p. 18, A86-016, 242-14.

V-90

/SECRET/BB UNCLASSIFIED

(b)(3)

Earth Penetrators

The final quarter W-Division report for 1970 indicates that there was an earth-penetrating program called Rumpler. LASL had received authorization in November to participate with Picatinny Arsenal and SLA in a joint program to explore the feasibility of the Rumpler concept. LASL was to be responsible for the study of the nuclear device for inclusion in the penetrator. The final quarterly report for 1970 from the Laboratory states, "The Rumpler program is an attempt to demonstrate the feasibility of penetrating earth to a moderate depth with a missile launched straight down from a mobile recoilless rifle on a portable field mount

(b)(3)

7. 1971

(b)(3)

Modified Spartan

The Laboratory status report for the first quarter of 1971 noted that the Laboratory was . continuing work on a Modified Spartan program.

(b)(3)

It was noted that the Scimitar proposal had been released for inclusion in the AEC advanced planning document and also for the Phase 2 feasibility document.⁴⁸⁹ The W-Division report for the first quarter also notes that detailed studies had been applied to the Scimitar proposal for the Modified Spartan.⁴⁹⁰

(b)(3)

⁴⁸⁹"Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4680-MS (SRD) (May 1971), p. 30.
 ⁴⁹⁰"W-Division Quarterly Status Report, January 1, 1971 through March 31, 1971," W-2318 (SRD) (April 15, 1971), p. 14, A86-016, 242-16.

LA-14066-H

LA LILL

(b)(3)

/It is a step in

LA-14066-H

the development of a warhead for the Modified SPARTAN."""

Hardsite Defense (SPRINT II)

The third quarter report for 1971 from the Laboratory notes that a new component of the Safeguard ABM system, called "Hardsite Defense" (HSD), was to be implemented. It was reported, "HSD consists of an autonomous module for close-in, low-altitude intercept (~10,000 to 30,000 ft) and is based upon three radar/data-processor units located about 10 nautical miles apart. The module will have six or seven firing sites containing about 100 modified Sprint interceptors to defend approximately 21 silos." The new interceptor was to be known as Sprint II: the planned readiness date was July 1977.

(b)(3)

An AEC/DOD Joint Working Group had been set up to provide an alternative that used less tritium. A number of proposals had been suggested,

(b)(3)

⁴⁹³D. P. MacDougall to Major General E. B. Giller, USAF, ADW-158 (SRD) (November 5, 1971), p. 4, A99-019, 258-28.

(b)(3)

SSIFIED

(b)(3)

Included in the list of proposed FY73 shots that were presented during the December 8, 1971, WWG meeting was a test for the HSD

(b)(3)

High-Yield Bomb

In the early months of 1971, the weapon groups conducted design studies for a High-Yield Bomb to meet a Phase 2 sign-off date of early 1972.⁴⁹⁹ The W-Division quarterly report for January 1 through March 31, 1971, notes, "A special group of designers has been assigned to explore methods of fulfilling the requirements for a new High Yield Bomb (HYB). This essentially is renewed effort on the improved Class D FUFO, concentrating on increased lavdown loads and higher vields.

(b)(3)

I ne Laboratory's quarterly report for the period ending September 1971 notes that the nuclear designs for the Phase 2 input for the High-Yield Bomb had been completed.

(b)(3)

(Also included in the Phase 2 proposal was the minimum number of nuclear tests that would be required to complete the development of the selected design.⁵⁰² The W-Division third quarter report states that the designs suggested ranged in diameter from 18 to 22 inches.

(b)(3)

⁴⁹⁹ "Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4680-MS (SRD) (May 1971), pp. 31-32.

(b)(3)

³⁰²"Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4820-PR (SRD) (December 1971), p. 51.

LA-14066-H

UNCLASSIFIED



(b)(3)

By the fourth quarter of 1971, requirements for the High-Yield Bomb had become more specific.

(b)(3)

The bomb was to be delivered by

both tactical and strategic aircraft. Five candidates for the Phase 2 had been selected and the joint SLA-LASL input for the Phase 2 study had been submitted to FC/DNA on December 20.⁵⁰⁴

High-Yield Multiple RV (Mk 19)

The high-yield multiple RV program was also <u>underway in 1971</u>. This program was in reality a continuation of the Mk 19 effort.

(b)(3)

⁵⁰⁴ Quarterly Status Report on Weapons Research and Development for the Period Ending December 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4880-PR (SRD) (March 1972), p. 36.

(b)(3)

للمتغد تغذ

V-94

(b)(3)

High-Yield Warheads

Harold Agnew in a letter to Assistant General Manager for Military Application, Major General Edward B. Giller, stated that the country should consider what would happen if there were to be a ban on the use of MIRV and at the same time there was a ban on testing. Agnew recommended that the country should develop and test the highest possible yield warhead that could be carried as a single [warhead] on the Poseidon. He felt that the same should be done for the Minuteman and, if the Titan were kept in service, a high-yield warhead should also be developed for this application./He noted, "...we won't be carrying out our responsibility to the U.S. if we don't do these nuclear developments."⁵¹⁰

(b)(3)

⁵¹⁰H. M. Agnew to Major General Edward B. Giller, DIR-2250 (SRD) (July 6, 1971), 2 pp., A99-019, 269-3.

LA-14066-H

(b)(3)

AAREIPURAD (

LASSIT

V-95



b. Two-Stage, Lower-Yield Weapons

ABC (Advanced Ballistic Concept)

At this time, the ABC program was in a sense part of the Mk 18 program that will be described in the final chapter of this report.

(b)(3)

These are

discussed in Chapter II.

ACE (Advanced Control Experiment) and SEE (Small Evader Experiment)

There continued to be interest in providing warheads for a maneuvering-type vehicle. One proposed maneuvering vehicle weapon program was called the Advanced Control Experiment (ACE). The stated advantage of this type of weapon was that it would replace the Mk 12 with a greatly enhanced penetration capability. In 1971, it was reported that the first flight test for the missile system was planned for the next year

(b)(3)

A similar program was called SEE (Small Evader Experiment). In the summer of 1971, this program had just been started. It was reported in a memo from W-9's C. M. Gillespie as a 5-month paper study. The goal of the program was to develop the smallest maneuvering RV that was "strategically effective for Poseidon."⁵¹⁵

(b)(3)

He reported,

"Aerospace and SAMSO are interested in the LASL proposal but want to know [if] it will work before giving much consideration to it."

The reason given for the interest in these maneuvering defense systems was to counter a future Soviet terminal ABM threat.⁵¹⁶

(b)(3)

^{310th}Weapons Working Group, Minutes of the 234th Meeting," WWG-234 (SRD) (October 20, 1971), p. 11, A99-019, 92-18.

UNCLASSIFICD

LA-14066-H

(b)(3)

VSM (Very Small Multiple)

Also mentioned in this time period was the Very Small Multiple (VSM) concept where 10 to 14 warheads would be carried on the Minuteman III missile,

(b)(3)

c. Single Stage

ADM (Atomic Demolition Munition) and SADM (Small Atomic Demolition Munition) The first quarter report from the Laboratory states, "A study has been made to determine the amount of stemming necessary for significant containment of a NATO ADM device detonated in a simple hole relatively near the surface (45 ft)."

(b)(3)

The energy moving up the hole was deposited in the walls generating a radial shock that enlarged the hole.⁵¹⁹

(b)(3)

High-Yield Primary

There were several tests related to the development of a high-yield primary.

(b)(3)

^{S10} Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4680-MS (SRD) (May 1971), pp. 33-34.

LA-14066-H

.

LASSIFIED

UNCLASSIFIED AABAA/KAV

(b)(3)

d. Projectiles and Earth-Penetrating Weapons

Rumpler

During 1971, the weapon groups continued work on the Rumpler program. At that time, the program was again described in the Laboratory's first quarter report as follows, "The Rumpler program is an attempt to demonstrate the feasibility of penetrating earth to a moderate depth with a missile launched straight down from a mobile recoiless rifle on a portable field mount

(b)(3)

SAA BED

LA-14066-H

shath was ale

SAA ALALAN

Work on the Rumpler program continued during the second quarter

(b)(3)

Criticality tests were begun in June. It was reported, "Tests will be done with the warhead surrounded by air, water, and in soil mixtures of clay, sand and water with varying density and water content."526 [Author's note: These studies were apparently done to demonstrate that the device would not preinitiate before achieving the desired depth.]

The Laboratory's third quarter report states that the propellant charge for the Rumpler gun had been determined. Three earth-penetration tests had been completed by Sandia personnel.527

The final quarterly reports from the Laboratory and W-Division for 1971 note that all of the interior ballistic tests as well as the propellant setback and functioning tests had been completed. The mechanical nuclear-safing system components for the Rumpler device had also been completed, assembled, and tested. 528

e. Nonuclear Kill

The Laboratory's status report for the period ending March 31, 1971, reports that there had been a revival of interest in nonnuclear kill for a hard-point ABM defense system and for antisatellite applications. The Laboratory weapon groups were investigating the "kind of ietprojectile that is formed by half of an implosion system.

(b)(3)

Apparently there was little continued interest in this program because later progress reports fail to mention the project. [Author's note: John Hopkins recalls that this work continued for several years before it was phased out.⁵³⁰ Use of tuballoy projectiles would of course be studied and developed at Sandia and other facilities.]

⁵²⁶ Quarterly Status Report on Weapons Research and Development for the Period Ending June 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4732-MS (SRD) (August 1971), p. 39. "W-Division Quarterly Status Report, April 1, 1971 through June 30, 1971," W-2327 (SRD) (July 15, 1971), p. 20, A86-016, 242-17.

⁵²⁸"Quarterly Status Report on Weapons Research and Development for the Period Ending December 31, 1971 (U)," Los Alamos Scientific Laboratory report LA-4880-PR (SRD) (March 1972), p. 36. "W-Division Quarterly Status Report, October 1, 1971 through December 31, 1971," W-2358 (SRD) (Januarv 15, 1972). p. 32. A86-016. 242-19.

(b)(3)

UNCLASSIFIED

³³⁰John Hopkins, personal communication (SRD) (March 12, 2003). spielest and

⁵²⁷ Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1971 (U)," Los Alamos Scientific Laboratory report LA-4820-PR (SRD) (December 1971), p. 56. "W-Division Quarterly Status Report, July 1, 1971 through September 30, 1971," W-2343 (SRD) (October 15, 1971), p. 46, A86-016, 242-18.



8. 1972 and First Quarter of 1973

(b)(3)

Modified Spartan

In a February 28, 1972, memo, D. P. MacDougall reported that the AEC's Giller anticipated that a Phase 3 request might be initiated in the FY 1972–1974 time period for the Modified Spartan.⁵³¹

The Phase 2 report for the modified Spartan was published in April 1972

(b)(3)

Site Defense (Sprint II)

The Laboratory's first quarter report for 1972 indicates that the Laboratory weapon groups were continuing to work on the Hardsite Defense program. However, the program was now called Site Defense. Its objective was to defend a portion of the U.S. Minuteman force. The new interceptor for this program, called Sprint II, had a slightly reduced launch dispersion, increased hardness, and decreased miss distance. The missile in its cell, as well as the entire module, was to operate virtually unattended. The planned readiness date continued to be July 1977.⁵³³

In a February 28, 1972, memo, D. P. MacDougall reported that in a January 20, 1972, document, Giller had indicated the he anticipated that a Phase 3 request might be initiated in the FY 1972–1974 time period for a Hardsite Defense System (Sprint II).⁵³⁴

(b)(3)

⁵³¹D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11. Drawer 49. Folder 1 of 5.

(b)(3)

⁵³³Leslie M. Redman and Cecil C. Carnes, "Quarterly Status Report on Weapons Research and Development for the Period Ending March 31, 1972 (U)," Los Alamos Scientific Laboratory report LA-4965-PR (SRD) (June 1972), p. 54.

⁵³⁴D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11. Drawer 49. Folder 1 of 5.

V-100

(b)(3)

SELERETRE

UNCLASSIFIED

(b)(3)

The Laboratory's third quarter report for 1972 notes that the Laboratory had submitted its SD (Site Defense) Phase 2 study. This study included information on (1) minimum tritium and plutonium usage, (2) rapid-deployment capability, (3) maximum ratio of warhead lethal radius to SD-system miss distance, and (4) minimum perturbation of the Sprint II missile

(b)(3)

Details of the design are presented in the cited reference.⁵⁴⁰ The first quarter report of 1973 indicates that the Los Alamos groups were continuing to consider options for the Site Defense system.

(b)(3)

³⁴⁰Leslie M. Redman and Cecil C. Carnes, Jr., "LASL Weapons Quarterly for the Period Ending December 31, 1972 (U)," Los Alamos Scientific Laboratory report LA-5150-PR (SRD) (March 1973). pp. 56-57.

(b)(3)

LASSIFIED

LA-14066-H

ŝ

(b)(3)

High-Yield Bomb (HYB), Formerly FUFO

In a February 28, 1972, memo, D. P. MacDougall reported that Giller in a January 20, 1972, document had indicated that he anticipated that a Phase 3 request might be initiated in the FY 1972–1974 time period for a High-Yield Bomb (HYB) (formerly FUFO).⁵⁴³

The second quarter report from the Laboratory for 1972 includes, as part of the High-Yield Bomb program, what was to become a very important project for the Laboratory. The report states, "The explosive TATB (1,3,5-triamino-2,4,6-trinitrobenzene) was first described in 1888, but was recognized as a very interesting heat-resistant explosive only in the 1950s. At that time the Naval Ordnance Laboratory reported briefly on its properties and presented a reasonable synthesis." The report notes that this material had a great insensitivity to impact and friction and an explosive power superior to that of other heat-resistant explosives,

(b)(3)

The third quarter report for 1972 states, "We have continued experiments to characterize the 90/10 TATB-Kel-F 800 material discussed last quarter, hampered somewhat by our limited stock of material. First deliveries of TATB from Pantex are expected soon to alleviate this shortage."⁵⁴⁵

The third quarter report also states, "The USAF and USN have studied the 'Final Report of the Phase 2 Feasibility Study of the High Yield Bomb,' dated March 21, 1972, ...and agree that it is technically feasible to develop a high-yield bomb that meets the stated requirements, with a minimum 4-yr development time, and that no bombs that meet the requirements of the HYB are currently available or in production. Subsequently, the Services have requested a Phase 3 Development Engineering program for the HYB.

(b)(3)

³⁴⁶D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11. Drawer 49. Folder 1 of 5.

(b)(3)

⁵⁴⁵Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), pp. 53–54.

LA-14066-H

LASSIFIED
(b)(3)

The Laboratory's fourth quarter report for 1972 announced that a new full-fuzing-option (FUFO) bomb program had been requested. (This program took the place of the high-yield bomb program.) The bomb size was limited to a maximum of 18-in. diameter, 145-in. length, and a 2,400-lb weight. It was to have full-fuzing options and improved safety characteristics. It would be for delivery by both strategic and tactical aircraft/

(b)(3)

This new FUFO bomb program was a result of the review of the March 21, 1972, Phase 2 Feasibility Study of the High-Yield Bomb and the decision that certain aspects of the feasibility study should be expanded before a final decision was made on a Phase 3 development. In terms of design considerations, for the LASL proposals it was noted that, for safety reasons, it was desirable that the bomb incorporate use of the new insensitive explosive

(b)(3)

V-103

SSIFIED



By the first quarter of 1973, the High-Yield Bomb was now being called the new FUFO bomb. The Laboratory's first quarter 1973 report again notes that when the DOD staff had reviewed the Phase 2 study, they had indicated that certain aspects of the feasibility study would need to be expanded before a final decision could be made on the Phase 3 development. Three concerns had been cited as requiring additional information. These were (1) improved safety characteristics, (2) economy of nuclear materials, and (3) the ability to field-test new designs under the then existing restrictions at NTS. In response, for the supplemental Phase 2 study, in January 1973, LASL-SLA presented three proposals,

(b)(3)

SCM (Strategic Cruise Missile)

During the later half of 1972, the Navy initiated a system's study for a strategic cruise missile (SCM). It was noted that the interest in the SCM program stemmed from the fact that the SALT agreement did not cover strategic cruise missiles; the Russians claimed that cruise missiles were tactical weapons. It was reported that the program had 6 million dollars of funding for 1972 and 35 million for the next year. The proposed missile was to have a range of at least 2,000 nm and was to use a rocket boost plus turbofan cruise engine for propulsion. The missile was to be capable of being launched from Polaris submarine tubes (one to six missiles per tube), torpedo tubes, and Talos-Terrier missile launchers.

(b)(3)

Data on four existing warheads had also been provided.³⁵¹

Specific information on warhead yields and sizes for the Submarine Launched Cruise Missile was provided in a December 26, 1972, letter from T. A. Sandford and C. M. Gillespie from the LASL Technical Liaison team to Commander Hugo Hardt of the Naval Air Systems Command.⁵⁵²

During the first quarter of 1973, there continued to be support from Los Alamos in relation to the preparation of the Phase 1 study for the submarine-launched cruise missile for the Navy. The baseline requirements for the SCM system had been set; the target date given for operational availability was November 1978. Warhead design proposals were to aim for minimum volume and minimum intrinsic radiation from the missile.⁵⁵³

(b)(3)

⁵⁵¹Leslie M. Redman and Cectl C. Carnes, Jr., "LASL Weapons Quarterly for the Period Ending December 31, 1972 (U)," Los Alamos Scientific Laboratory report LA-5150-PR (SRD) (March 1973), p. 60.

⁵⁵²T. A. Sandford and C. M. Gillespie to Cdr. Hugo Hardt, Subject: "Joint LASL/SLA Preliminary Warhead (WH) Data for the Submarine Launched Cruise Missile (SLCM) (U)," ADWP-1-72-97 (SRD) (December 26, 1972), 4 pp., B11, Drawer 56, Folder 1 of 4.

⁵⁵³Leslie M. Redman, "LASL Weapons Quarterly for the Period Ending March 31, 1973 (U)," Los Alamos Scientific Laboratory report LA-5330-PR (SRD) (June 1973), p. 64.

V-104

LA-14066-H

ASSIFICE

UNCLASSIFIED SALARIE MRIO

High-Yield, Multiple RV (Mk 19, Modified Mk 12) The high-vield, multiple RV program was continued in 1972.

(b)(3)

On August 17, 1972, the Air Force Weapons Laboratory requested a new Phase 1 study to support renewed Air Force interest in a new ballistic RV for the MM (Minuteman) III. Two types of RVs were to be considered in the Phase I study. One type was to be a modified Mk 12 RV with a maximized

(b)(3)

LA-14066-H

UNCLASSIFIED

UNCLASSIFIED ARCARMAN

vield consistent with the Mk 12 envelope,

(b)(3)

Responding to this request, the LASL designers developed a design that they felt was suitable for incorporation into the modified Mk 12;

(b)(3)

A letter dated October 4, 1972, transmitted the formal LASL/SLA Phase-I warhead concept information for the new Minuteman III ballistic reentry vehicle program, (b)(3) (b)(3) It was noted, "Our [LASL] studies on the Mk 19 reentry systems for

Minuteman began in early 1969. Since then we have had continuing efforts on both RV synthesis and warhead design and testing." Additional information is available in the cited reference.⁵⁵⁸

(b)(3)

A new Planning Information Document to reflect this thinking had been issued by the Albuquerque Office (ALO).

(b)(3)

⁵⁵⁸C. M. Gillespie and T. A. Sandford to Col. Charles C. Hyre, Air Force Weapons Laboratory, Kirtland Air Force Base. N.M. 87117, ADWP-72-45 (SRD) (October 4, 1972), 16 pp., B11, Drawer 53, Folder 2 of 2.

secretion MASSIFIED

LA-14066-H

PAR (Prototype Advanced Reentry)

It was also announced in the third quarter 1972 report from the Laboratory that the Aerospace Corporation was considering a possible study for a high-yield prototype advanced reentry (PAR) vehicle. It was reported, "This study will have a much broader scope than a Mk 19 or a Modified Mk 12 study." The interest was in the use of three 400-lb warheads carried on the Minuteman III missile

(b)(3)

<u>Nova</u>

(b)(3)

ALASSIFIED

V-10/

LA-14000-H

(b)(3)

(Additional information is available in Chapter II and in the cited reference.²⁰²)

(b)(3)

ABC (Advanced Ballistic Concept)

(b)(3)

In a

sense, this program was similar (but had a reduced yield requirement) to the lightweight MIRV program discussed in the previous section.

R. B. Olwin in December 4, 1972, wrote to Capt. D. R. Mentzer of the Space & Missile Systems Organizations to update him on the ABC program.

(b)(3)

Ulwin wrote, "We

wish to congratulate the Air Force, SAMO, Aerospace, and Avco on the success of the ABC program through the first UGT and flight tests." Olwin added, "We feel significant advantages are to be gained through weight savings in an ABC-type vehicle by using some of the new technology we are now pursuing. We would be happy to meet with you at your convenience to discuss any of these subjects."⁵⁶⁷

On March 27, 1973, in support of the ABC Program Design Update Task, T. P. Seitz, another member of the LASL Technical Liaison team, sent Capt. D. R. Mentzer a "payload design update." This update showed various options, including several new warhead configurations that the LASL team felt were suitable candidates for the program. Additional information is available in the cited reference.⁵⁶⁸

(b)(3)

INCLASSIFIED

⁵⁶⁵Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), pp. 61-64.

(b)(3)

⁵⁶⁷R. B. Olwin to Capt. D. R. Mentzer, Subject: "Updated Warhead Information Pertaining to the ABC Program," ADWP-72-31 (SRD) (December 4, 1972), 5 pp., B11, Drawer 49, Folder 1 of 5.
⁵⁶⁸T. P. Seitz to Capt. D. R. Mentzer, Subject: "Joint LASL-SLA Payload Update Data for ABC Program," ADWP-1-73-57 (SRD) (March 27, 1973), 7 pp., B11, Drawer 107, Folder 2 of 3.

V-108



The Navy's Mk 400

The third-quarter report for 1972 from the Laboratory formally announced the Mk 400 program.⁵⁶⁹ This program will be described in the following chapter.

SEV (Small Evader Vehicle)

The third-quarter report from the Laboratory also noted a program called the small evader vehicle (SEV) program. It was stated that LASL had supported this program with preliminary warhead data followed by a formal Mk 500 Phase 1 data package. Representatives from LASL had attended working meetings with the staff at McDonnell Douglas Astronautics Company.⁵⁷⁰

[Author's note: The Mk 500 was a Navy sponsored program. It was in a sense similar to the Mk 400, but the Navy wanted the Mk 500's missile to be able to change course on demand in order to attack the designated target.]

SCAD (Subsonic-Cruise Armed Decoy)

SCAD was envisioned to be a USAF turbojet-powered, air-launched, 800-mile-range missile designed to enhance bomber penetrations. The missile was to be carried on the B-52 and FB-1111 (b)(3)

Reflecting the continuing interest in this type of weapon, in a February 28, 1972, memo, D. P. MacDougall reported that Giller had noted that he anticipated that a Phase 3 request might be initiated in the FY 1972–1974 time period for a Subsonic Cruise Armed Decoy (SCAD).⁵⁷¹

In July 1972, the Director of Defense Research and Engineering (DDR&E) requested that the AEC reinitiate the Phase 2 study for the SCAD program. A March 1973 completion date was requested. The Laboratory's third quarter report states, "The original Phase 2 study for SCAD, requested by DDR&E in October 1970, was never officially initiated because the AEC did not receive an approved Phase 1 data package."

The third-quarter report from the Laboratory also notes, "LASL continues to study SCAD WH candidates that will best fit the SCAD missile and meet USAF requirements."⁵⁷²

The Laboratory's first-quarter report for 1973 indicates that for the SCAD Phase 2 input LASL had submitted four major proposals for the design

(b)(3)

⁵⁶⁹Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), p. 68.

⁵⁷⁰ Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), p. 70.

⁵⁷¹D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11, Drawer 49, Folder 1 of 5.

⁵⁷²Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), pp. 70–71.



Pershing II

The Laboratory's third quarter 1972 report indicated that the LASL groups were working with the Army to conduct a Phase 1 study on a nuclear warhead for an improved Pershing system. (The Pershing missile was a two-stage, surface-to-surface ballistic weapon that was capable of engaging targets to a range of 400 miles.)

[Author's note: The reader will recall that the original Pershing 1A missile carried the W50 warhead

(b)(3)

Planned modifications for the

LA-14066-H

Pershing II included improved accuracy, a warhead with the same or improved effectiveness, and a reduced time to fire as well as an increased firing rate.

(b)(3)

LASL was asked to submit proposals for all three types of warheads for the Phase 1 study for the Pershing II.

Preliminary proposals were presented at the second Phase 1 meeting. For additional information on these, the reader is referred to the cited reference. It should be noted, however, that for the earth-penetrating warheads, different types of projectiles were suggested. For the standard fission warheads, both single-stage and two-stage designs were suggested.⁵⁷⁴

The first quarter report for 1973 notes that the Laboratory had submitted its final input for the Pershing II Phase 2 study in March. Included in the eight proposals were (EPW) (including a Rumpler-like design) and airburst fission warheads. A combination warhead was also proposed. It was noted, "The combination warhead merges EPW technology with airburst or ground-burst weapons, or both. This system could be used against a wide range of targets, and its mode of use in each case would be determined by relative requirements for maximizing target kill while minimizing collateral damage."⁵⁷⁵

(b)(3)

There continued to be interest in an ADM. In a February 28, 1972, memo, D. P. MacDougall reported that in a document that he had recently received from Giller, Giller had indicated that he anticipated that a Phase 3 request might be initiated in the FY 1972–1974 time period for an Improved Atomic Demolition Munition.⁵⁷⁶

(b)(3)

⁵⁷⁴Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), pp. 73–74.

⁵⁷⁵Leslie M. Redman, "LASL Weapons Quarterly for the Period Ending March 31, 1973 (U)," Los Alamos Scientific Laboratory report LA-5330-PR (SRD) (June 1973), p. 67.

⁵⁷⁶D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11, Drawer 49, Folder 1 of 5.

UNCLASSIFIED

ASSIFIED ARAA KO

(b)(3)

d. Projectiles and Earth Penetrating Weapons Mark 48 Torpedo

The third quarter report for 1972 from the Laboratory notes that the input proposals for the Phase 1 study for the Mark 48 convertible warhead had been completed and submitted for review; the Phase 1 document had been published in August 1972. [Author's note: A convertible warhead was a warhead that could use a conventional HE warhead, or when required, this warhead could be exchanged for a nuclear warhead, or turned into a nuclear warhead.]

(b)(3)

It was suggested that the insertable-capsule concept could also be used in the Condor, Harpoon, and the extended-range Walleye. The Laboratory's quarterly status report for the period ending September 30, 1972, indicated that it was hoped that a nuclear test of a device representative of the insertable-capsule concept could take place before June 30, 1973.⁵⁷⁸

Harpoon

The Laboratory's fourth quarter report for 1972 indicates that the Laboratory had also been asked to participate in a similar Phase I study for a convertible warhead for the Harpoon, the Navy's new antiship missile. It was reported that the Los Alamos weapon development teams were investigating the feasibility of low-yield convertible warheads for this application.

(b)(3)

Additional information on the proposed design is available in the

(b)(3)

⁵⁷²Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), pp. 72–73.

LA-14066-H

INCLASSIFIED

cited reference 579

(b)(3)

Penetrator

The third quarter report from the Laboratory indicates that Sandia was conducting an extensive study of earth-penetrating weapons. The staff at Los Alamos were reviewing this work.⁵⁸¹

9. Conclusions

a. Work Load

As the reader will no doubt have realized from reading the previous sections, during the 1965–March 1973 time period there were an incredible number of Phase 1 and 2 projects that were requested by the various groups in the Military. As a result, a large amount of Laboratory effort must have gone into the preparation of suitable studies and proposals and attending meetings with the Military and their contractors. (John Hopkins has commented that one reason W-9 was formed was to respond to all the Phase 1 and Phase 2 requests made to the Laboratory.)⁵⁸² As reported in the previous sections, at times full-scale tests were also completed as part of these projects. However, despite all the effort and money spent on these, many of these projects never progressed into Phase 3 programs.

In addition to the Phase 1 and Phase 2 programs, there were other projects undertaken that the Laboratory management believed to be important in terms of weapon development. Some projects appeared to be important enough that the Laboratory management approved tests at NTS related to these programs. Many of the primary development tests described in Chapter II fall into this category.

b. Importance of the Work

There were a number of efforts that, one may conclude, ended as "dead end" projects. However, if testing during this time period had been restricted to Phase 3 programs only, the Laboratory weapon development groups would not have been able to test the concepts that would later be incorporated into the XW76.

Thus, the fact that the test program at that time allowed for testing of advanced systems was important in future weapon development.

(b)(3)

LA-14066-H

SSIFIED

⁵⁷⁹Leslie M. Redman and Cecil C. Carnes, Jr., "LASL Weapons Quarterly for the Period Ending December 31, 1972 (U)," Los Alamos Scientific Laboratory report LA-5150-PR (SRD) (March 1973), p. 61.

⁵⁸¹Leslie M. Redman and Cecil C. Carnes, Jr., "Quarterly Status Report on Weapons Research and Development for the Period Ending September 30, 1972 (U)," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1973), p. 73.

⁵⁸²John Hopkins, personnel communication (U) (November 21, 2002).

UNCLASSIFIED SECRETIANO/

c. Shift in Design, Materials, and Engineering

One of the most interesting aspects of the early 1970s program was the shift in design, materials, and engineering that occurred in almost all the various components that make up a nuclear weapon. The technical foundation for the designs that Los Alamos would contribute to the modern stockpile was quickly being developed during this critical time period.

(b)(3)

Safety was becoming a major concern. In response to these concerns, the HE formulation groups were investigating the use of insensitive explosives; the designers would adapt their primary designs to use these. In addition, the engineers were investigating techniques to reduce the threat from possible diversion of weapons.

(b)(3)

The metallurgists were developing for use in the new pits plutonium alloys that contained minimum amounts of gallium.

These switches are evident in the new designs that the Laboratory was listing in the 1972 Phase 1 and Phase 2 proposals. These switches would be critical in the Laboratory's success in obtaining new Phase 3 programs.

C. Trends

1. Change in Composition

There are some interesting trends that need to be considered in a discussion of the work of the tate 1960s and early 1970s.

One of these trends is the total number of weapons in the stockpile. Figure V-5 shows this number as a function of year and type. While, as shown in Figure V-5, the number of weapons in the stockpile actually increased rapidly from 1958 until 1966, a slow decline began after that period.

Of greater interest are the number of builds and number of retirements as a function of year shown in Figure V-6. As the reader will note, there was a major decline in production after 1960.

CLASSIFIED

(b)(3)

LA-14066-H

Î

ASSIM

LAU CI

JNCLASSIFIED NAME OF THE OWNER (b)(3)

HAVHANNAN STIFTEL

Another figure of interest is the data on yield versus year shown in Figure V-7. As the reader will note, once megaton yield weapons were developed and entered the stockpile, there was a very rapid rise in the total stockpile yield. However, following the early 1960s, there was a continued decrease in yield as the smaller yield systems entered the stockpile.

Another item of interest is the use of special nuclear material.

(b)(3)

2. New Weapon Systems

Another interesting trend is the continuing decline in the number of new weapon systems coming into the stockpile as a function of year. The reader will recall that the first nuclear weapons had been employed in 1945. The Phase 3 for the Mod 0 for the B61 was June 1963.

(b)(3)

During the latter half of the 1960s and early 1970s, despite all the work that had gone into the Phase 1 and Phase 2 projects, as previously noted, many of proposed weapons never reached Phase 3. Moreover, several of the Phase 3 programs were canceled before the weapon system went to the stockpile. After the award of the B61, the W62 (Minuteman's Mk 12 RV) went to Livermore. The Army's sponsored 63, 64, and 65 were never produced. The Army's W66 went to Los Alamos. The Air Force's W67 went to Los Alamos but was canceled. The Navy's W68 for Poseidon went to Livermore. The W69 for the SRAM missile went to Los Alamos. The W70 for the Army's Lance missile went to Livermore as did the W71 for the Army's Spartan missile. Los Alamos received the W72 assignment for the Walleye and the W73 for the Condor. Although the 155-mm projectile was awarded to Los Alamos as the W74 and the 8-in. to Livermore as the W75, both projects were canceled.

(b)(3)

LA-14066-H

ADDITIL

(b)(3)

V-117

UNCLASSIFIED

LA-14066-H **JLASSIFIED** 1 ? ₹ (b)(3) V-118



WHANNAN SSIFIED

V-119

UNCLASSIFIED ARAMMA MAD

Looking ahead in the period 1973-1980 the number of new systems was even more limited. The ABC program and Mk 18 programs (that evolved into the Mk 400 program) were precursors to the W76; the Phase 3 authorization for this strategic warhead was received by LASL in May 1973. The high-yield multiple RV program/Mk 19/Mk 12A was the precursor for the W78 (Phase 3, June 1974), an assignment that would also go to LASL. The High-Yield bomb program would go to Livermore as the XW77 with a Phase 3 of May 1974. (This program would later become the B83.) The Safeguard (Spartan/Sprint) program would be discontinued. (The weapons that had been stockpiled under this program would be retired.) The improved 8-inch artilleryfired projectile with a Phase 3 date of January 1975 would go to Livermore as the W79. The W80 program, the cruise missile project (air launch, sea launch, and advanced cruise missile) would receive a Phase 3 date of June 1976 and would go to Los Alamos.

(b)(3)

The W82, a 155-mm artillery shell, was given to Livermore but was never produced. The W84, the ground-launched cruise missile warhead, would have a Phase 3 of September 1978 and would go to Livermore. The W85 for the Army's Pershing II missile went to LASL (Phase 3 of May 1979), but all warheads would be retired in March 1991 and their components used to build the B61-10

The W86, the design for the (b)(3) Pershing II Earth Penetrator, was cancelled after Phase 3.584 Thus, there were in a sense seven projects in seven years that reached the stockpile.

With few projects being awarded in this time period, it was a tough fight for the Laboratories to procure and complete a Phase 3 assignment.

Decreased Levels of Funding 3.

Edward Giller, Assistant General Manger for Military Application, in a TWX dated September 17, 1970, reported to the Laboratories that the low level of FY 1971 on-continent funding as well as the trend being experienced in the overall level of funding meant that less money would have to be spent in the NTS test program than had been spent in previous years. Giller noted, "We must look both at the need for specific tests and test programs and at the way they are conducted. On the need side, we must candidly question such things as total numbers of tests being conducted on similar Phase 2 designs, on effects, or on high yield devices."585 This was a notice that the NTS test program would need to undergo some changes. Tests would be limited to those considered to be the most important.

Giller continued his warnings to the Laboratories concerning the need to limit spending. In a document dated January 20, 1972, titled "FY-1974 Weapons Program Budget Planning Assumptions," Giller stated, "Any work which does not directly support either present or anticipated future weaponization requirements must be relegated to a lower priority category.

LA-14066-H

⁵⁸⁴ Betty L. Perkins, "Why Nougat? (U)," Los Alamos National Laboratory report LA-12950-H (SRD) (November 1, 1995), pp. A-1-A-3.

⁵⁸⁵ USAEC, Edward B. Giller, Wash., D.C. to BW3, UCLRL, M. M. May, Livermore, Calif., et. al. (OUO) (September 17, 1970), 3 pp., A99-019, 198-12. SECREMENT UNULASSICILD

\$\$\$APPEARE

This does not imply that such work is not important; it simply means that in light of the present and the anticipated restrictions on overall funding for the nuclear weapons program, the primary emphasis must be directed toward support of current weapons program objectives."⁵⁸⁶

4. Tighter Test Requirements/Cost Increases

After the venting of Baneberry (fired on December 18, 1970), the procedures for conducting an underground test underwent major revisions. These revisions were made to ensure that a similar venting episode would not occur again.

In a letter to Major General E. B. Giller, Assistant General Manager for Military Application dated April 9, 1971, D. P. MacDougall, Assistant Director for Weapons at Los Alamos wrote, "...the post-Baneberry procedures are being estimated to increase the cost per test of field construction by at least 20 or 25 percent, and thus the total number of tests is considerably reduced from that which we had been assuming would be feasible."⁵⁸⁷

5. Result

In the early 1970s, there were major changes in the U.S. weapon program. First, because of funding limits and other restrictions, not as much effort could be placed on advanced development. Second, there were a very limited number of Phase 3 projects that were being given to the weapon laboratories. Third, the limitation on testing and money meant that when a Phase 3 award was made, not as many tests and not as detailed a diagnostic program as might have occurred otherwise to develop a weapon would take place. And finally, with a limited number of projects available and a limitation on money for testing, the development teams at each Laboratory were under great pressure to produce.

The development teams had to meet all the stringent requirements placed on that weapon, and at the same time they had to achieve the required weaponized system quickly and with a minimum effort.

It was not foreseen at that time that several of the weapons undergoing development in the 1970s and 1980s time period were going to be in the stockpile for a long, long time.

⁵⁸⁶D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11, Drawer 49, Folder 1 of 5.

⁵⁸⁷D. P. MacDougall to Major General E. B. Giller, USAF, ADW-79 (SRD) (April 9, 1971), p. 2, A99-019, 258-27.

ŞEÇRIZIMRIZ

LA-14066-H

UNCLASSIFIED

This Page Intentionally Left Blank

Augurermen - ASSIND

LA-14066-H



CHAPTER VI TABLE OF CONTENTS

A. Mk 18	VI-3
1. The Question of a Pie Split	VI-3
2. The Ouestion of the Primary	VI-4
3. Uncertainty in the Program	
4. The Question of the Secondary	VI-16
5. Where Do We Go From Here?	VI-17
6. Phase-Out?	VI-21
7. Transformation	VI-23
B. Mark 400 Program	
1. Livermore vs Los Alamos: The Navy's Submarine-Launched	
Ballistic Missile (SLBM) Warheads-W47, W58, and W68	VJ-24
2. Advanced Planning: 1966-August 1969	VI-26
3. Advanced Systems Studies: September 1969-September 1971	VI-27
4. A New Ballistic Missile System	VI-34
5. Approval for a Mk 400	VI-36
6. At Last, a Phase 2.	VI-40
4 1 (0)	
(b)(3)	
(D)(3) 8. Where is the Phase 3?	VI-51
 (b)(3) 8. Where is the Phase 3?	VI-51
 (b)(3) 8. Where is the Phase 3?	
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64
(b)(3) 8. Where is the Phase 3? 9. Continued Hope for a Phase 3? Maybe 10. Finally—A Phase 3 Assignment 11. Significance C. Harold Agnew 1. Proponent	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64
(b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64 VI-65
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64 VI-65 VI-65
(b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64 VI-65 VI-66 VI-66 VI-67
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64 VI-65 VI-65 VI-66 VI-67 VI-67
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-65 VI-65 VI-66 VI-67 VI-67 VI-68
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-65 VI-65 VI-66 VI-67 VI-67 VI-68 VI-68 VI-69
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-64 VI-65 VI-65 VI-66 VI-67 VI-67 VI-67 VI-68 VI-69 VI-69
 (b)(3) 8. Where is the Phase 3?	VI-51 VI-58 VI-61 VI-64 VI-64 VI-64 VI-65 VI-65 VI-65 VI-66 VI-67 VI-67 VI-67 VI-68 VI-69 VI-69 VI-69 VI-69

LA-14066-H

/states

. 7

UNCLASSIFIED

CLASSIFIED UI the second

This Page Intentionally Left Blank

LA-14066-H SSIFIED

CHAPTER VI. THE MARK 18 AND THE MARK 400 PROGRAMS AND THE PUSH FOR A STRATEGIC MISSILE WARHEAD

A. Mk 18

1. The Question of a Pie Split

Harold Agnew has described the origin of the pie split. In the period 1956–1960 both the Livermore and Los Alamos laboratories were responsible for a large number of Phase 3 programs. Agnew has reported, "With this heavy load of Phase 3 work, it was agreed by both LASL and Livermore managements and by DMA that, to the maximum extent feasible, the two laboratories would get together and arrive at recommendations as to which laboratory would do which development job, and then so recommend to DMA. DMA much preferred this procedure to that of making the decisions unilaterally in Washington and then listening to squawks from one or both of the laboratories. ... The division of the Phase 3 jobs was mostly handled by this sort of agreement which was usually referred to as 'pie splitting'."¹

During the first part of 1966 there were two weapon systems that the Military was proposing that both the Livermore and Los Alamos groups very much wanted as Phase 3 assignments in terms of the warhead: the Mk 3 RB's warhead and the Mk 18 RV's warhead. The Mk 3 RB was to be used on the Navy's Poseidon missile; the Mk 18 RV was for the Air Force's Minuteman—the use of multiple RVs on the missile was envisioned in both these applications. [Author's note: The Navy refers to the reentry system not as a reentry vehicle (this is an Air Force designation) but as a reentry body, hence the designation Mk 3 RB.]

The Navy's Poseidon with its Mk 3 RB had a proposed operational date of August 1970; it was anticipated that the Phase 3 assignment for the warhead would be made in June 1966.

(b)(3)

Bradbury really wanted Los Alamos to have the warhead assignment for the Poseidon and for Livermore to have the warhead assignment for the Minuteman. He indicated this desire in a letter dated May 11, 1966, to Livermore director, Michael M. May.³ However, May replied to Bradbury on May 20 and outlined why he believed that technically Livermore was in a much better position to supply the Poseidon/Mk 3's warhead.

(b)(3)

On the other hand, May felt that the Mk 18 (RV system for the Minuteman) was likely to have a later date required for its production,

(b)(3)

May noted, "This is an area where neither Laboratory

TLASSIFIED

¹H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), pp. 2–3, B11, Drawer 56, Folder 1 of 4.

(b)(3)

³N. E. Bradbury to Dr. Michael M. May, DIR-2028 (SRD) (May 11, 1966), 3 pp., A99-019, 198-7.

LA-14066-H

UNCLASSIFIED SECRETURE

seems to have an edge in experience, and hence, if the workload situation warrants it, the warhead might well be assigned to LASL."⁴ In a letter dated June 1, 1966, to May, Bradbury conceded that if the Mk 3's warhead assignment were to be made soon, Livermore would get it. However, Bradbury stated, "If you end up doing the Mark 3, we would certainly insist that we should do the Mark 18."5

In a TWX dated June 23, 1966, Delmar L. Crowson, Director of Military Application, informed the laboratory directors that he had concluded that the designs utilizing the LRL proposals were the most suitable candidates for Phase 3 development of the Mk 3 warhead.

(b)(3)

Because of this, he was requesting that the Phase 3 assignment go to Livermore. (The warhead for the Mk 3 would be designated W68.) However, there were advantages to having different warheads be developed by different AEC laboratories. Crowson stated that he intended to assign the Mk 18 to Los Alamos.°

During the January 4, 1967, meeting of the Los Alamos WWG (Weapons Working Group), J. K. S. Walter reported that money had been budgeted for the Mk 18 program. However, final approval had not been received through the Air Force. Extensive briefings were scheduled to take place during the month. Interestingly, Walter reported that four warheads were being considered.

(b)(3)

The maximum allowed weight was 150 pounds. As many as eight Mk 18 RVs were planned for use on a single Minuteman."

In a memo dated January 16, 1967, the Chief of Staff of the Air Force noted that current plans were to start development of the Mk 18 Reentry System in FY68. The memo requested that a Concept Formulation Package for the Mk 18 aimed at Contract Definition in July 1967 be provided. The memo also noted that the programmed IOC (Initial Operational Capability) of the Mk 18 RV was July 1971. Although the previous 1966 study had been concerned with an Assured Destruction Mission, the new 1967 study was to also include the Damage Limiting aspect. Additional information on the desired inputs is available in the cited reference.8

2. The Ouestion of the Primary

(b)(3)

It was noted, "They needed this information as soon as possible in order to proceed with system's effectiveness studies."

LA-14066-H

⁴Michael M. May to Dr. N. E. Bradbury, BY 66-32 (SRD) (May 20, 1966), pp. 2-4, A99-019, 198-7. N. E. Bradbury to Dr. Michael M. May, DIR-2032 (SRD) (June 1, 1966), p. 1, A99-019, 186-2. ⁶USAEC, Delmar L. Crowson, Wash., D.C. to RUWPQA/USAEC L. P. Gise, Albuq., N.M. et. al. (SRD) (June 23, 1966), 3 pp., A99-019, 182-1.

⁷"Weapons Working Group, Minutes of the 170th Meeting," WWG-170 (SRD) (January 4, 1967), pp. 8-9, A99-019, 92-12.

⁸Chief of Staff, USAF to AFSC (SCL), Subject: "Mk-18 Re-entry System Trade-Off Analysis (U)," (SRD) (January 16, 1967), 4 pp., A99-019, 186-2. SACRETARIA ASSIFIED

In order to meet this request, a meeting was held at Los Alamos on February 20, 1967, where representatives from GMX-3, T-2, W-4, and W-1 were given information concerning what the various warhead designs submitted by Los Alamos to BSD might be

(b)(3)

Personnel from

VI-5

GMX-3 and W-1 were charged with preparing a formal data package.⁹

(b)(3)

On February 27, 1967, the various weapon laboratories were informed that BSD was going to hold a meeting on March 3 at the Norton AFB in California. The topic of discussion was to be the Mk 18's warhead. Included in the discussion would be the Mk 18 technology and development time scales.¹¹

(b)(3)

The design had just been released to W-1.¹²

E. C. Dudziak, R. G. Shreffler, and J. K. S. Walter represented Los Alamos at the March 3 meeting at Norton AFB. During the meeting it was indicated that the Phase 3 decision on the warhead would come between July 1967 and January 1968. The first production unit was scheduled for January 1971

(b)(3)

In their memo reporting on the meeting, the LASL

SECRET/BO UNCLASSIFIED

⁹J. K. S. Walter to Distribution, Subject: "MK 18 Warhead Designs for Systems Studies," W-1-E-12205 (SRD) (February 21, 1967), 3 pp., A99-019, 182-1.

(b)(3)

¹¹BSD, Norton AFB, Calif., to RAUWPWO/Lawrence Radiation Lab., Livermore, Calif. et. al. (SRD) (February 27, 1967), 3 pp., A99-019, 182-1.

¹²"Weapons Working Group, Minutes of the 173rd Meeting," WWG-173 (SRD) (March 1, 1967), pp. 2–3, A99-019, 92-13.



attendees indicated that the Los Alamos team was to supply a partial input for the Mk 18 study by March 10. This input was to be followed by a more complete, Phase 2-like, document.¹³

On March 10, Walter sent a letter to Headquarters, Ballistic Systems Division that outlined the suggested Los Alamos designs for the Mk 18

(b)(3) Walter's letter included schematics giving "Size dimensions of the various options. He noted, "All of the warheads proposed can be developed within the presently anticipated Mark 18 schedule."¹⁴

On March 14, 1967, Brig. General Delmar L. Crowson sent out a copy of a letter, dated February 17, 1967, from the Air Force to the Manager of the Albuquerque Operations Office as well as to the directors of the various AEC laboratories. In this Air Force letter Major General Otto J. Glasser, representing the Air Force, had requested that the Director of Military Application proceed with having the AEC laboratories participate in studies to optimize the Mk 18 RV configuration. He had reported that the Air Force intended to request a formal Phase 2 study as soon as the warhead requirements were better defined.¹⁵

During the March 15, WWG meeting. Shreffler reviewed the Mk 18 program.

(b)(3)

The weapon groups were working on "Phase 2-like" documents. They hoped to have version two ready during the week of April 3.¹⁶

On April 25, 1967, the Los Alamos and Sandia planning team sent out a 29-page report titled "Technical Data Package for Mk 18 Systems Effectiveness Study." This report was similar to the March 10 proposal but included additional details in terms of weaponization

(b)(3)

¹³E. C. Dudziak, R. G. Shreffler, J. K. S. Walter, Subject: "Trip to BSD, Mark 18 Discussion," AW-1255 (SRD) (March 7, 1967), 5 pp., A99-019, 182-1.

¹⁴J. K. S. Walter to Headquarters, Ballistic Systems Division, W-1-E-12237 (SRD) (March 10, 1967), 10 pp., A99-019, 182-1.

¹⁵Brigadier General Delmar L. Crowson to Those Listed Below, Subject: "Mk-18 Reentry Vehicle," (SRD) (March 14, 1967), 1 p.; Otto J. Glasser to Director of Military Application (SRD) (February 17, 1967), 1 p., A99-019, 182-1.

¹⁶"Weapons Working Group, Minutes of the 174th Meeting," (SRD) (March 15, 1967), pp. 7–9, A99-019, .92-13.

(b)(3)

LA-14066-H

ASSILIED

Angrania

On May 16, in a meeting with Major L. H. Laird of BSD, Walter was told that the Phase 2 request for the Mk 18 had been turned down within the DOD pending completion of the Phase 1 package (b)(3)

On June 1, 1967, Walter sent out \overline{y} et another memo to his team. He noted the design characteristics of the reentry vehicle and the fact that seven vehicles per booster had been used in a recent study that had included an analysis of the weight vs yield as a function of design. As part of Walter's memo, the results of the study showing RV weight vs yield were presented in the form of graphs.²¹

(b)(3)

During the WWG meeting of June 7, Walter announced that the Air Force Weapons Laboratory was currently putting together the Mk 18, Phase 1 package. It should be available later in the week.

(b)(3)

The WLPC meeting held on June 13, 1967, was, from a historical point of view, a very interesting one. It was called to discuss possible primaries for the Mk 18. During the meeting, it was announced that the Phase 1 package in draft form was available.

(b)(3)

²¹J. K. S. Walter to Distribution, Subject: "Mk 18 Design Study," W-1-E-12492 (SRD) (June 1, 1967), 11 pp., A99-019, 182-1.

(b)(3)

LA-14066-H



(b)(3)

On June 19, R. G. Shreffler, Alternate W-Division Leader, sent out a memo to upper management,

(b)(3)

LA-14066-H

SSIFIE

During the July 6, 1967, WLPC meeting, it had been announced that the Mk 18 program had slipped about 9 months; a Phase 2 date was not predictable.³⁰

(b)(3)

(b)(3)

In order to take advantage of your experience in this field as rapidly as possible, I suspect that we will want to have a good many conversations with various members of your staff familiar with the various problems you have encountered. We will try to be as little trouble to you as possible, but I hope we may count on your assistance, and the purpose of this letter is to request this!"³⁴ In a letter dated August 23, 1967, May agreed to assist. He noted, "A potentially mutually advantageous diagnostic program regarding these primaries would be their flash X-ray analysis on your superior Phermex facility."³⁵

(b)(3)

[Author's note: LX-09 was a Livermore developed formulation. The Livermore scientists felt, at that time, that this explosive was stable at relatively high temperatures.] A. Popolato from GMX-3 announced to the HWG attendees that the Los Alamos explosive formulation staff had also developed an explosive that performed as well as LX-09 at high temperature. This explosive was going to be known as PBX 9408.

(b)(3)

³⁴N. E. Bradbury to Dr. Michael M. May, DIR-2091 (SRD) (August 17, 1967), 1 p., A99-019, 182-1. ³⁵Michael M. May to Dr. N. E. Bradbury, Director, BY #67-55 (SRD) (August 23, 1967), 1 p., A99-019, 273-2.

LA-14066-H

UNCLASSIFIED

Popolato, during his presentation, compared the performance of PBX 9404, LX-09, and PBX 9408.³⁶

On August 29, 1967, there was a meeting at SAMSO (Space and Missile Systems Organization) in San Bernardino, California. The purpose of the meeting was to discuss the technical effort that would be required over the next few months to support the Mk 18 program.

(b)(3)

SAMSO personnel had requested the latest warhead design-information in order that Aerospace personnel, assisted by Sandia, could proceed with their reentry vehicle studies. Walter, who attended as the LASL representative, informed the attendees that LASL did not have a firm Mk 18 warhead-proposal.

(b)(3)

Walter, in his report of the meeting, indicated that he had learned that a Phase 2 request was once again being submitted. It was possible that a Phase 2 meeting might be held in December or January. There was to be a meeting with all the various Military groups involved as well as representatives from the AEC laboratories on September 7, 1967. The discussion was to focus on the threat specification.³⁷

During the August 30, 1967, WWG meeting, Thorn reported that the Mk 18 program had been delayed in the hopes of improving the CEP (circular error probability).

(b)(3)

On September 5, 1967, Bradbury in a TWX to Brig. Gen. Edward B. Giller at the DMA stated, "...we are not presently sure as to what we should be doing on the Mark 18 whose production date seems to be indefinitely postponed."³⁹

(b)(3)

Additional information on the Walter request is available in the cited reference.⁴⁰ [The author has discussed the development of the W68 in LA-13755-H (SRD).]

(b)(3)

³⁹N. E. Bradbury, Los Alamos Scientific Lab., Los Alamos, N.M. to Brig. Gen. Edward B. Giller, DMA, Wash., D.C., DIR-2094 (CRD) (September 5, 1967), p. 3, A99-019, 182-1.
 ⁴⁰J. K. S. Walter, Jr. to Dr. C. A. McDonald, W-1-E-12819 (SRD) (September 13, 1967), 3 pp., A99-019, 182-1.

VI-10

SECRETATION (

LA-14066-H

UNCLASSIFIED

³⁶F. W. Kramer to Distribution, Subject: "Thirtieth Meeting of the Hydrodynamics Working Group," GMX-3-6461 (SRD) (August 30, 1967), 8 pp., A99-019, 76-19.

³⁷J. K. S. Walter, Jr. to J. J. Wechsler, Subject: "Trip Report; Mk 18 Meeting; August 29, 1967; SAMSO; San Bernardino, California," W-1-E-12800 (SRD) (August 31, 1967), 3 pp., A99-019, 182-1.

(b)(3)

At the WLPC meeting of November 9, 1967, Deal announced that the Los Alamos group had seen a draft of a Phase 2 letter on the Mk 18.

(b)(3)

opinion of the members present."44

The group agreed that this reflected the

(b)(3)

⁴⁴"Hydrodynamics Working Group, Minutes of the 32^{ad} Meeting, November 9, 1967," HWG-32 (SRD) (November 9, 1967), pp. 7–8, A99-019, 76-19.

LA-14066-H

ARCHARMA UNULASSIFIED



She also reported that the Hydrodynamic Committee had been assigned the job (b)(3) of renaming the primary system. 45

(b)(3)

Uncertainty in the Program 3.

On December 1, 1967, another Mk 18 meeting was held at SAMSO, Norton AFB. The LASL attendees reporting back to LASL noted that there was "some foundation for the rumors concerning cancellation or delay of the Mk 18 program." A decision was to be made early in January or before. If the cancellation notice did not come, the LASL group had agreed to supply, by December 15, an external configuration drawing of the "first choice" LASL candidate. Using this information, the SAMSO and Aerospace groups would then make their best detailed estimate of the overall design early in January.⁴⁸ In a letter dated December 29. 1967, the LASL external configuration drawing was supplied to SAMSO. (b)(3) An external configuration drawing of

(b)(3) a reservour/valve assembly was also supplied. The location for the zipper was indicated.49

It should be noted that at that time there continued to be discussions that indicated that the program objectives were not clear. An example of this uncertainty is the information reported during the January 31, 1968, meeting of the WWG. The minutes of the meeting state, "About the MK 18 as formulated, the systems analysts say everything from worthless to very good." The minutes also indicate that it appeared, at least to the LASL group, that the Military planners could not even agree on what system they wanted or what the objectives were.⁵⁰

With the Mk 18 program in a sort of limbo, Agnew proposed setting up a group similar to the SPO (Special Projects Office), which the Navy, Lockheed, and LRL had formed and which had been used successfully for the Polaris/Poseidon programs. Agnew noted, "The group is to examine the problem objectively and collectively decide what they believe the payload for the Minuteman III should be. From this the MK-18 should evolve." Agnew then stated, "We must be as imaginative, wild, 'can do anythingish' as possible. The FPU [First Production Unit] date, if

⁴⁵Jane H. Hall to Distribution, Subject: "Memorandum of Understanding - WLPC No. 21," AD-1814 (SRD) (November 13, 1967). 2 pp., A99-019, 182-2.

(b)(3)

⁴⁸G. A. Thompson and J. K. S. Walter to J. J. Wechsler, Subject: "Trip Report, Mk 18 Meeting at SAMSO, Norton AFB, December 1, 1967," W-1-E-13049 (SRD) (December 11, 1967), 2 pp., A99-019, 182-2.

⁴⁹J. K. S. Walter to Headquarters, SAMSO, W-1-E-13095 (SRD) (December 29, 1967), 2 pp., A99-019, 182-2.

⁵⁰"Weapons Working Group, Minutes of the 188th Meeting," (SRD) (January 31, 1968), p. 9, A99-019, 92-15.

VI-12

ARCHARTOR TINCLASSIFIED

ever, is a long way off. We can do anything if it takes forever." [Author's note: This proposal was apparently in some measure an attempt to discourage Livermore from trying to "grab" the program.] Agnew reported that Giller had stated that he expected LASL to do the Mk 18.⁵¹

Agnew won his committee and a Mk 18 SPO-like group was indeed formed. The first meeting was arranged for February 13, 1968, at the Aerospace Corporation. By the time the group was set up, the mission/candidates for the Mk 18 program had become much more diverse than had been the case in the fall of 1967. Additional information is available in the cited reference.⁵²

(b)(3)

The first Mk 18 "SPO" meeting was held as scheduled. The attendance list shows a large number of people present from SAMSO, FC/DASA (Field Command, Defense Atomic Support Agency), AEC/ALOO, AFWL, LASL, Sandia, TRW (Thompson, Ramo & Woolridge), DMA, and Aerospace. The LASL report of the meeting notes, "Fortunately, the subject of Livermore participation did not come up." The LASL report also states, "The first meeting was unfortunately too large to achieve an across-the-table atmosphere." The next meeting was to be at LASL on March 26, 1968.⁵⁴ (This meeting was actually held on March 29 at LASL. Among those in attendance was a representative from LRL as well as representatives from Sandia, ALO, FC/DASA, DMA, AFWL, SAMSO, Aerospace and TRW.⁵⁵) The group became known as the Mk 18 Technical Planning Group.

During the February 21, 1968, meeting of the WWG, in response to the fact that the Mk 18 program had become a very general one, Bradbury commented that the Laboratory better pursue "all versions of the Mk 18."⁵⁶

(b)(3)

⁵⁴J. K. S. Walter, W-9 to H. M. Agnew, W-DO, Subject: "First Mk 18 'SPO' Meeting; Aerospace Corporation; February 13, 1968," W-9-14 (SRD) (February 16, 1968), 4 pp., A99-019, 182-2.
 ⁵⁵J. K. S. Walter, W-9 to Distribution, Subject: "Mk 18 Technical Planning Group Meeting March 29, 1968," W-9-48 (SRD) (March 26, 1968), 3 pp., A99-019, 182-2.

⁵⁶ Weapons Working Group, Minutes of the 189th Meeting," (SRD) (February 21, 1968), p. 14, A99-019, 92-15.

LA-14066-H

SEABETARD

UNCLASSIFIED

⁵¹H. M. Agnew to Distribution, Subject: "Meeting at AFWL," W-2082 (CRD) (January 8, 1968), 2 pp., A99-019, 182-2.

⁵²J. K. S. Walter to Distribution, Subject: "Trip Report; Reentry Systems Advisory Group Meeting, January 18; Mk 18 Meeting at Aerospace Corporation, January 19, 1968," W-1-E-13191 (SRD) (January 30, 1968), 3 pp., A99-019, 182-2.

SACREAM

There must have continued to be concern at LASL that Livermore was going to "take" the Mk 18 program. To settle the issue, in a TWX dated February 27, 1968, Giller stated, "I reconfirm the decision contained in the above references that LASL is assigned responsibility for the Mk-18 warhead program." Giller added, "The assignment is not contingent upon an assumed set of system/warhead characteristics but applies regardless of warhead size or yield."⁵⁷

(b)(3)

Additional information is available in Chapter II.]

(b)(3)

Osborne indicated that he wanted to discuss with Sack certain issues connected with this design during a scheduled March 21, 22 meeting at Livermore.

(b)(3)

[Author's note: The reader will recall that Sack was the principal primary designer for the W62 and W68.]

In an April 4, 1968, memo to the Division of Military Application, R. E. Schreiber requested that the AEC contract for the delivery to Los Alamos of a CDC 6600 system. Schreiber noted, "The Mark-18 in particular will require a very significant computing effort, larger than previously pictured." He noted that the design effort on the Mk 18 warhead was being hampered by a shortage of computing capacity.⁶¹

(b)(3)

Case parts had been ordered for

local shots and two NTS shots.

(b)(3)

⁵⁷USAEC, Edward B. Giller, Wash., D.C. to RUWTHRB/USAEC L. P. Gise, Albuq., N.M. et. al. (SRD) (February 27, 1968), 2 pp., A99-019, 182-2.

(b)(3)

^{o1}R. E. Schreiber to Brig. General E. B. Giller and D. B. Anthony (SRD) (April 4, 1968), 7 pp., A99-019, 202-10.

VI-14

Artering

UNCLASSIFIED

Stephent And

(b)(3)

The Mk 18 Technical Planning Group met at the Aerospace Corporation in San Bernardino_ in an informal meeting on April 18 and 19, 1968.]

(b)(3) /J. K. S. Walter brought a copy of the design nome for comments. Additional information on other topics of discussion is available in the cited reference.⁶³

(b)(3)

In their trip reports covering their recent May 1968 trips to Aerospace and Sandia, Walter and Thompson included some interesting information. They noted that during their meetings they had received indications that the status of the Mk 18 was still indefinite, a delay of some form appeared to be imminent.

(b)(3)

⁶⁹J. K. S. Walter, W-9 to Distribution, Subject: "Trip Report, Mk 18 Meeting, Aerospace Corporation, San Bernardino, April 18 and 19, 1968," W-9-91 (SRD) (May 1, 1968), 4 pp., A99-019, 182-2.

(b)(3)

NCLASSIFIED

VI-15

(b)(3)

By July, it appeared that a decision might be made soon on moving ahead on the Mk 18. Walter had once again attended a meeting held on June 28, 1968, at Aerospace Corporation. Walter stated in his trip report, "Because of very recent indications that the Mk 18 may move ahead on the schedule outlined in the Concept Formulation Package (Contract Definition in FY 1969 and IOC [Initial Operational Capability] in December 1972), a meeting was called by SAMSO in order to provide an AEC briefing to the Minuteman people in SAMSO and TRW.²⁰

(b)(3)

4. The Question of the Secondary

In an August 9, 1968, memo, secondary designer, A. T. Peaslee, Jr. (T-2), reported on the problems the secondary designers were having in the design of the Mk 18 secondary.

(b)(3)

felander verse

NCLASSIFIED
(b)(3)

5. Where Do We Go From Here?

The fifth meeting of the Mk 18 Technical Planning Group was held on October 4. 1968.

(b)(3)

There is the possibility that some other number will be assigned to the reentry system."⁷⁴ [Author's note: There is no information, that the author could find in the Los Alamos files, as to why after the Secretary of the Air Force briefing on July 11 the expected approval was not forthcoming.]

(b)(3)

⁷⁴J. K. S. Walter, W-9 to Distribution, Subject: "Trip Report; Mk 18 Technical Planning Group; AFWL; October 4, 1968," W-9-251 (SRD) (October 8, 1968), 3 pp., A99-019, 182-3.

SIFIED

LA-14066-H

-

VI-17

(b)(3)

Walter then closed his letter by

stating, "It is difficult, and perhaps unwise, to make the laboratory development program responsive in detail to the excursions encountered in the Mk 18 program."⁷⁵

(b)(3)

In mid-November 1968, a Mk 18 concept briefing was given to the Air Staff and Air Force Systems Command by representatives from the Space and Missile Systems Organization (SAMSO). (b)(3)

Formulation Plan. If events moved smoothly, an Initial Operational Capability in the last quarter of 1973 appeared possible.⁷⁹

The sixth Technical Planning Group meeting for the Mk 18 was held on December 3, 1968.

(b)(3)

⁷⁵J. K. S. Walter to Headquarters, Space & Missile Systems Organization (SMQ), Subject: "Mk 18," W-9-268 (SRD) (November 5, 1968), 4 pp., A99-019, 182-3.

(b)(3)

⁷⁹"Weapon Development Status Report," Headquarters Field Command Defense Atomic Support Agency, Sandia Base, Albuquerque, New Mexico report FC12680581 (SRD) (December 19, 1968), p. 19, A99-019, 161-1.

LASSIFIED

LA-14066-H

Additional information is

(b)(3)

available in the cited reference.³⁰

(b)(3)

[Additional information is available in Chapter II.]

(b)(3)

A document titled "Mk 18 Description, Planning Information" was circulated in February' 1969. This was described as an update to the December 6, 1967, planning document. The warhead had the dual roles of Assured Destruction and Damage Limiting Missions. Included in the document was information on the proposed warhead.

(b)(3)

reference.82

Additional information is available in the cited

(b)(3)

Three RVs were to be used per booster.

(b)(3)

⁸⁰J. K. S. Walter to Aerospace Corporation, Subject: "Mk 18," W-9-291 (SRD) (December 16, 1968), 11 pp., A99-019, 182-3.

(b)(3)

³² Mark 18 Description, Planning Information," U.S. Atomic Energy Commission, Albuquerque Operations Office document, (SRD) (February 1969), 26 pp., A99-019, 182-3.

LA-14066-H

SELABANING UNCLASSIFIED

UNCLASSIFIED SPACERE PARIS

(b)(3)

To further add complexity as to what design would be used in the Mk 18 warhead, there continued to be interest in the use of eleven RVs per booster.

(b)(3)

One can see that the Military at this time might have been somewhat disappointed in the Los Alamos program for the Mk 18.

(b)(3)

SECRET/RD UNCLASSIFIED

LA-14066-H

UNCLASSIFIED

6. Phase-Out?

On August 15, 1969, representatives from LASL and SLA presented a briefing on the Mk 18, Mk 19, and Mk 12A warheads at SAC (Strategic Air Command) Headquarters. In his trip report, H. C. Hoyt stated, "SAC believes that there is a rapidly increasing need for the capability to kill hard targets.1 (b)(3)

Hoyt also noted, "They don't believe that the Mk 18 gives any significant improvement over the Mk 12 for softer targets, and so they are not interested in the Mk 18." The SAC people also appeared to be uninterested in a warhead replacement for the Mk 12A

(b)(3)

He recommended that the Mk 19 effort be given priority over the

Mk 18 effort.88

(b)(3)

⁸⁸ H. C. Hoyt, W-DO to Distribution, Subject: "SAC Missile W/H Requirements," AW-1328 (SRD) (August 20, 1969), 2 pp., A99-019, 182-4.

(b)(3)

LA-14066-H

CIRCUMP (UNCLASSIFIED



(b)(3)

In a TWX dated September 15, 1970, Bergen noted that the Mk 18 and Mk 19 had been studied in detail for use on Minuteman III.

(b)(3)

The meeting minutes report, "This is probably the last MK 18 shot to be fired."⁹⁹ for this long delay may not only have been technical but also political; Los Alamos had failed to obtain a Phase 3 assignment for the Mk 18. (b)(3)

In a July 27, 1971, memo, W-9's C. M. Gillespie reported that the Mk 18 program was now the ABC program (the Advanced Ballistic Concept program) (b)(3) The program's objective was to provide an RV of 200 pounds or less for MIRV application to either Minuteman III or Poseidon (although the Navy did not appear interested).

(b)(3)

Gillespie in his memo states, "The program is

(b)(3)

⁹⁸"Weapons Working Group, Minutes of the 230th Meeting," (SRD) (March 17, 1971), p. 9, A99-019, 92-18.

VI-22

SECTEMBER UNCLASSIFIED

in a flight hardware development phase with two flight tests scheduled as well as underground vulnerability tests."

(b)(3)

7. Transformation

LA-14066-H

SSILLEU

ĩ

UNCLASSIFIED SEARCA MARA

B. Mark 400 Program

1. Livermore vs Los Alamos: The Navy's Submarine-Launched Ballistic Missile (SLBM) Warheads—W47, W58, and W68

Navy planners and others realized that a ballistic missile outfitted with a nuclear warhead that could be launched from a nuclear submarine presented a weapon system that could be highly effective as a deterrent. This concept was placed into practice with the development of the first SLBM nuclear warhead: designated the W47. This warhead was designed for use with the Navy's Polaris A-1 and the Polaris A-2. It first entered the stockpile in June 1960. There were several modifications made in the warhead during its stockpile lifetime. The next warhead placed into deployment on a submarine-launched ballistic missile was the W58. This was used in the Mark 2 reentry system fitted on the Polaris A-3; this missile could carry three warheads. The weapon development groups at Livermore were responsible for these early warheads. The Lockheed Missile and Space Company (LMSC) carried the major responsibility for the carrier of the warhead(s). A detailed account of the development of the W47 and W58 is given in LA-13755-H (SRD). During the development of these systems, a close relationship was established between Livermore, Lockheed, and the Navy.

The early Livermore–Lockheed–Navy association was a close one. However, the administration at Los Alamos very much wanted to be included in this group. One way to get a foot in the door was to have Los Alamos people present at planning meetings. [Author's note: It should be noted that as an exception Ralph Williamson from T-2 participated in the group known as the Polaris Ad Hoc Group for Long Range Research and Development. He appears to have attended the meeting held on April 29–30, 1965, at the Naval Research facilities in Washington.¹⁰⁶ Although documentation is lacking, he probably attended similar meetings held at other times.]

In the fall of 1965, Al Bridges from Kaman Nuclear visited Los Alamos. At that time a request was made to him to have representatives from Los Alamos participate in the Navy's planning meetings. [Author's note: The administration at Los Alamos must have known that the Navy was interested in developing a new ballistic missile/submarine system to be called Poseidon.] On December 6, 1965, the Chief of Naval Operations requested that the AEC laboratories be authorized to cooperate with the Navy in joint conceptual studies leading to the determination of optimum parameters for various components of the Poseidon missile. This request in turn led to the inclusion of Los Alamos in a memo from Brigadier General Delmar L. Crowson, Director of Military Application. Crowson requested that the Sandia Corporation, the Los Alamos Scientific Laboratory, and the Lawrence Radiation Laboratory participate with the Navy in these requested conceptual studies.¹⁰⁷

VI-24

/stacktering

UNCLASSIFIED

¹⁰⁶Chairman, Polaris Ad Hoc Group for Long Range Research and Development (PLRRD) to Distribution, Subject: "PLRRD Meeting 29-30 April 1965," SPOO110/RHY:md 3900 (U) (March 31, 1965), 1 p., Ralph Williamson papers, A86-049, 1-1.

¹⁰⁷Brigadier General Delmar L. Crowson to L. P. Gise, Subject: "Joint AEC/Navy Conceptual Studies for Optimization of Poseidon Missile System," (SRD) (December 22, 1965), 1 p. and enclosure, A99-019, 217-15.

Thus, the Los Alamos group was able to have representation during the initial planning stages of what would become the W68 warhead for the Poseidon C3 missile. In a memo to L. P. Gise, Manager of the AEC Albuquerque Operations, from Crowson dated February 3, 1966, it was noted that a formal Phase 2 study for the Poseidon C3 missile warhead was not necessary. However, Crowson felt that until the formal authorization to proceed with Phase 3 development was received from the DOD, the AEC should document its early efforts by preparing an "in-house" Phase 2 study.¹⁰⁸ The "in-house" Phase 2 study was published in May 1966. Included in this document were the candidates put forth by LASL and SCSL (Sandia_____Corporation, Sandia Laboratory).

(b)(3)

JIn contrast, in the Livermore section of the "in-house" Phase 2 study, the Livermore group cited their extensive test and proposed test program relevant to the warhead's design.¹⁰⁹

As previously noted, on June 1, 1966, Bradbury wrote Livermore director, Michael May, "Our basic disagreement seems to reside in the question of the Mark 3 Poseidon warhead: We would both like to do it. In actual fact, if the assignment is made within the next few weeks, it would seem highly probable that it would go to LRL.

(b)(3)

However, as previouSIy reported, Bradbury also wrote, "If you end up doing the Mark 3, we would certainly insist that we should do the Mark 18."¹¹⁰ In a TWX dated June 23, 1966, Delmar Crowson (at that time Director of Military Application) stated in terms of the warhead for the Mark 3, "...I have concluded that the designs utilizing the LRL proposals are the most suitable candidates for Phase 3 development. Therefore, I request that AL [Albuquerque Operations Office] assign the development responsibility for the warhead for the MK-3 R/V to LRL/SCLL."¹¹¹

The first meeting of the Mk 3 Re-entry Body Coordinating Committee was held in the offices of the Lockheed Missile and Space Company (LMSC) in Sunnyvale, California on August 3–4, 1966. Attendance consisted of representatives from the Navy's Special Project Office (SPO), the Sandia Corporation Sandia Laboratory (SCSL), the Sandia Corporation Livermore Laboratory (SCLL), the Lawrence Radiation Laboratory (LRL), LMSC, the Special

¹¹⁰N. E. Bradbury to Dr. Michael M. May, DIR-2032 (CRD) (June 1, 1966), 2 pp., A99-019, 186-2. ¹¹¹USAEC Delmar L. Crowson, Wash., D.C. to RUWPQA/USAEC L. P. Gise, Albuq., N. M. et. al. (SRD) (June 23, 1966), 3 pp., A99-019, 198-7

INCLASSIFIED

¹⁰⁸Brigadier General Delmar L. Crowson, USAF to L. P. Gise, Manager Albuquerque Operations, Subject: "Small Re-entry Vehicles for Poseidon," (SRD) (February 3, 1966), 1 p., Los Alamos records center film SFSL-3843.

¹⁰⁹ An AEC Phase 2 Feasibility Study of a Warhead for a Small Reentry Vehicle for the Poseidon C-3 Missile System," Albuquerque Operations Office, U. S. Atomic Energy Commission, Albuquerque, New Mexico report RS 3410/589 (SRD) (May 1966), 65 pp., A99-019, 217-16.



Projects Office Sunnyvale (SPL), and FC/DASA (Field Command, Defense Atomic Support Agency).¹¹² Thus, Los Alamos was again "out of the loop" at this point.

(b)(3)

The history

Thus, the

of the Livermore design effort for the W68 is available in LA-13755-H (SRD). The weapon The warhead was a entered the stockpile in 1970. (b)(3) component in the Mk 3 RB (reentry body) carried on the Navy's Poseidon C3 missile (b)(3)

weapon could be used for single-target or multiple-target missions.

The Williamson papers available in the Los Alamos archives include a brief note made by Williamson on December 6, 1966

(b)(3)

Figures showing the design of several of these are available in LA-13755-H (SRD).]

Advanced Planning: 1966–August 1969 2.

Advisory groups, called the Advanced Technical Objectives Working Groups (ATOWG), were initially established during a meeting of the Laboratory directors on June 7, 1966. The reason for these groups was reported: "The combined weapons, engineering and research experience of the laboratories represents a fundamental resource necessary to military system concept formulation. The nine Naval warfare area working groups were established to capitalize on this collective expertise." [This author believes that Williamson was appointed in 1966 to be a member in these groups.] The first of a series of reports prepared by these groups was distributed in the spring of 1967.¹¹⁴

During March 23-24, 1967, Williamson attended a meeting of the Committee on Advanced Concepts for Sea Based Deterrence. It was reported that a document titled "Advanced Missile Technology" had been issued on March 3.¹¹⁵

(b)(3)

¹¹⁴Gerald W. Johnson, Department of the Navy, Director of Navy Laboratories, "Memorandum for Chairman and Members of the Advanced Technical Objectives Working Groups," (U) (August 8, 1967), 1 p., Ralph Williamson papers, A86-049, 1-1.

¹¹⁵"Committee on Advanced Concepts for Sea Based Deterrence, Minutes of Meeting 23-24 March 1967," (U) (no date), p. 1, Ralph Williamson papers, A86-049, 1-1.

VI-26

SECRETING UNILASSIFIED

¹¹²"Weapon Development Status Report (U)," Headquarters Field Command Defense Atomic Support Agency, Sandia Base, Albuquerque, New Mexico report FC-08660494 (SRD) (August 18, 1966), p. 6, A99-019, 160-4.

Williamson attended the ATOWG meeting held on September 3, 1968. His notes from the meeting indicate that "future submarines" was a big program for the Navy. These new submarines were to be faster and quieter. Rickover was apparently one of those pushing this program. Apparently, also under consideration was the FLOP concept. This was a big floating raft that would be used instead of an aircraft carrier. Williamson noted that they also discussed both ship hardening and silencing during the meeting.¹¹⁶

In October 1968, Williamson was asked to review a proposal titled "ULMS Vulnerability & Reactions Systems" that had been prepared by the Advanced Planning and Analysis Staff of NOL (Naval Ordnance Laboratory). This report noted that the ULMS (Underseas Long-Range Missile System) had only recently been approved for further study by the DOD. The proposed ULMS was the result of a STRAT-X study that had been performed to identify future strategic weapon systems that could best survive an enemy first-strike attack and still perform the necessary retaliatory strike. The ULMS was one of the two proposed systems approved by DOD for further study. (The other was the Air Force's hardened, rock-silo system.) It was noted, "ULMS will differ from the Polaris/Poseidon systems by having greater missile range and system availability while minimizing vulnerability and cost. The responsibility for the ULMS project was assigned to the Navy's Strategic Systems Project Office (SSPO) in the Spring of 1968...^{*117}

In his notes of the ATOWG meeting held on April 8, 1969, Williamson reported that the new system had been a topic of discussion. The ULMS boat needed to be quiet, it needed to be able to carry a large number of big missiles, and it needed to utilize a minimum crew. (Williamson also reported that there was a rumor that the C3 had problems. Apparently, little specific information was given to Williamson on this issue.)¹¹⁸

3. Advanced Systems Studies: September 1969–September 1971

In a letter dated September 3, 1969, representatives from the Lockheed Missiles & Space Company (LMSC) sent out a letter requesting studies of advanced reentry systems for the Navy's ULMS (Underseas Long-Range Missile System) study. Floyd B. Baker, a group member of W-9, received his copy of the letter on September 12.

On September 15, 1969, a meeting of the reinstituted Re-entry Committee was held at LMSC to discuss the Advanced Poseidon System. The committee was chaired by the Navy's Commander R. Stinner from the SSPO. Baker attended as the Los Alamos representative. In addition to those from SSPO and LASL, representatives from SLA, SLL (Sandia Laboratories, Livermore), KN (Kaman Nuclear), LRL, NOL (Naval Ordnance Laboratory), LMSC, SPL, and FC/DASA were included. R. J. Young from T-2 attended as an observer from Los Alamos. It was noted during the meeting that the DDR&E (Director of Defense Research and Engineering) had directed the Navy to conduct the necessary studies and technology investigations associated with an improved Fleet Ballistic Missile with a minimum modification to the Poseidon missile. It was noted that ULMS, LRC-3, IFBM, Poseidon II, and Advanced Poseidon were essentially all the same, namely a future Navy longer-range system. In fact, at that time, the Navy planners were considering increased range as the primary goal. An increased

¹¹⁸Ralph Williamson, notes (U) (April 8, 1969), 2 pp., Ralph Williamson papers, A86-049, 1-1.

LA-14066-H

FERRED/RD(

UNCLASSIFIED

¹¹⁶Ralph Williamson, notes (U) (September 3, 1968), 2 pp., Ralph Williamson papers, A86-049, 1-1. ¹¹⁷D. F. Bleil, Chairman ATOWG Committee to Dr. Ralph E. Williamson (U) (October 8, 1968), 1 p. and enclosure, Ralph Williamson papers, A86-049, 1-1.



range would provide a greater weapon system capability in the face of a potential ASW (Anti-Submarine Warfare). Baker, in his trip report, noted that representatives from LMSC had given most of the presentations. The LMSC group had reported on the minimum modifications necessary in the Mk 3 to permit improved performance. [Author's note: Sybil Francis in her report on Livermore has indicated that the SSPO group did not want a new warhead for the new missile; they preferred at that time to use a modified Poseidon warhead. This choice would minimize RB and warhead development costs and would avoid costly flight testing.¹¹⁹ C. M. Gillespie in a May 17, 1972, memo also stated, "In the Mk 400 program SSPO has always wanted the W68 as the warhead and LLL as the laboratory."¹²⁰ Additional information on this issue is presented in several of the following paragraphs.] During the September 15 meeting, it was reported that the LMSC/Navy had requested data packages from LASL/SLA and LRL/SLL covering warhead yields from 20 to 1,000 kt. Navy representatives hoped to use their old data coupled with this new input information to "come up with an optimized system." Navy representative Commander Stinner warned, however, that the Navy's funding for ULMS was minimal. He noted that the members should not return to their respective organizations and build large budget ULMS programs quoting SSPO justification.

(b)(3)

¹The Los Alamos group also provided information on the Mk 18 test program and indicated that this program was relevant to ' the warhead study for the Advanced Poseidon System. During the discussions, it was indicated that the high-risk, long-lead technological changes appeared to be the submersible and the submersible/missile interface. It was noted that the entire project hinged on the availability of a new sub that would accommodate a 48-foot by 100-inch missile. At the meeting's close, it was agreed that it was important to have formal meetings and full participation by the member organizations in order to exchange the necessary technical data.¹²¹

On November 19, 1969, the requested LASL/SLA data package was submitted to

H. D. Trudeau at LMSC. Eight proposed Los Alamos designs were described in terms of yield, size, weight center of gravity, and moment of inertia.

(b)(3)

The document would also note that an integrated arming, fusing,

INCLASSIFIED

LA-14066-H

¹¹⁹Sybil Francis, Warhead Politics – Livermore and the Competitive System of Nuclear Weapon Design, Massachusetts Institute of Technology thesis document (U) (September 1995), p. 155, Los Alamos archives.

¹²⁰C. M. Gillespie, W-9 to Distribution, Subject: "Draft of Mk 500 Phase 1 and 2 Program Proposal," (SRD) (May 17, 1972), p. 2, B11, Drawer 56, Folder 1 of 4.

¹²¹F. B. Baker, W-9 to H. M. Agnew, W-DO, Subject: "Trip Report – Meeting of the Re-entry Committee of the FBM Steering Task Group at LMSC, Sunnyvale, California 9/15/69," W-9-511 (SRD)

(September 29, 1969), 1 p., A99-019, 307-3. "Minutes of the Re-entry Committee of the Fleet Ballistic Missile Steering Task Group Meeting (U)," Kaman Nuclear report KN-69-440(R) (SRD) (September 29, 1969), 123 np., A99-019, 163-6.

and firing (AF&F) system incorporating component technology developed during the Mk 3 program and other recent advanced development programs could be supplied.¹²³]

The second meeting of the Re-entry Committee of the Fleet Ballistic Missile Steering Task Group was held on January 28, 1970, at the Naval Ordnance Laboratory (NOL) in Silver Spring, Maryland. It was reported that performance goals of the new ships to replace the Polaris/Poseidon ships were beginning to be outlined. The new ships would be on patrol 80% of the time with an 83-day patrol period. They would have a 7-15 db noise level below the present ships. They would include a long-life nuclear reactor. They would be designed with a collapse depth of 1,350 feet. For the missile launcher system, the tube dimensions being considered were a diameter of 115 to 135 inches and a length of 50 feet. The ships would each be capable of carrying twenty-four missiles. Each missile would have a minimum missile range of 4,500 nm. It was the missile that would be the determining factor in the size of the submarine. The missile candidates at that time ranged from the 34-foot long, 74-inch diameter C3 missile to a proposed 49-foot long 110-inch diameter missile. Several ship designs were under consideration, including a single hull vs several types of double hull configurations. Decisions still to be made included the number of missiles a ship should carry, launch tube diameter, launch conditions, number of people on each sub, shipboard environment requirements, ship/missile accuracy, and maintenance. The schedule then under consideration called for a nominal number of new-class ships by FY 1978. In terms of improvement in the warhead, it was indicated that with a suitable timeframe available for development, an explosive electric generator could be manufactured reliably. This type of generator would reduce the need for high currents for ignition. Improvements in CEP [circular error probability] were also possible. It was also reported that by using large reentry angles, high ballistic coefficients, and low radar cross sections, the offense could reduce the defense's engagement time. Hardening options were also presented at the meeting. It was recommended that the Navy spend "time and effort pursuing hardened system designs."124

(b)(3)

Although not part of the Mk 400 program, it will be noted that in the summer of 1970, the Navy was also interested in a program called CAFE. This stood for C-3, Alternate, Front End.

(b)(3)

[Author's note: There was concern at that time that, due to political considerations, that the use of the Mk 3/W68 (MIRV) might be restricted or abolished.] The first CAFE Reentry Interchange Committee Meeting took place on July 15, 1970. The LASL/SLA laboratories submitted data for the CAFE study in a document dated August 7, 1970. The principal warhead proposal from LASL was the use of a W67-like device. It was noted, "Many of the development tests, up to flight tests, have been completed for this device; construction drawings are available." Use of a 16M type of device was also

¹²³ LASL/SLA Revised Data Package for ULMS (Undersea Long Range Missile System) Study," W-9-754 (SRD) (September 1, 1970), 12 pp., A99-019, 214-15.

¹²⁴ Minutes of the Second Re-entry Committee of the Fleet Ballistic Missile Steering Task Group Meeting (U)," Kaman Nuclear report KN-70-68 (R) (SRD) (February 7, 1970), 178 pp., A99-019, 163-6.

(b)(3)

LA-14066-H

SA CARA A MAIL UNCLASSIFIED



suggested. Although the high-yield Poseidon warhead would never become part of the stockpile, it is interesting to note that Los Alamos and Sandia Albuquerque were involved in the CAFE program.¹²⁶ The close connection between LMSC, Livermore, and the Navy was becoming less restrictive. (Additional information on the CAFE program is available in the Los Alamos archive files.)

In a TWX dated August 6, 1970, H. C. Donnelly, manager of the AEC Albuquerque Operations Office, noted that the DMA (Division of Military Application) had requested that the Los Alamos/Sandia group provide, for planning purposes, by September 8, 1970, a single design for the ULMS application. However, Donnelly warned, "The selection of a Laboratory as a source of design information on a system for this purpose is in no way intended or expected to prejudice the future selection of a Laboratory to develop the system."¹²⁷ In response to this request, Los Alamos document W-9-820, dated September 10, 1970, was prepared and sent to Donnelly.¹²⁸

The third meeting of the Reentry Committee Fleet Ballistic Steering Task Group Meeting was held at Livermore on November 5, 1970. Attendees from Los Alamos were T. A. Sandford (GMX-3), D. W. Bergen (W-9), F. B. Baker (W-9), R. J. Young (T-2), and E. A. Bernard (W-4). During the meeting, J. Marion from Livermore reviewed Livermore's small-weapon design-effort.

(b)(3)

Representatives from LMSC

discussed representative RB/bus layouts. It was noted that range extension, achieved by placing a rocket motor on the bus and clustering the RBs around the motor, was being studied. The current objectives for ULMS were then listed.

Seventy-tour percent of the fleet was to be at sea all the time. Because a 14-day refit time in port was desired, modular maintenance of the system would be important. (b)(3)

It was reported that the submarine was going to be the pacing item. Representatives of LMSC discussed their ballistic RB studies. It was noted that because a definite mission had not yet been specified for ULMS, LMSC personnel were studying a wide variety of yields, weights, ballistic coefficients, hardness levels, and materials. It was noted that use of the existing Mark 3 would require a minimum modification.

(b)(3)

R. R. Neel from SLA discussed blowoff from filament reinforced heat shields. Neel's presentation was followed by a discussion from the LMSC representatives on the materials that they were interested in for heat shields and nose tips. In concluding his trip report, Sandford

VI-30

JCLASSIFIED

¹²⁶Floyd B. Baker, Group W-9 to Mr. C. E. Grant, Lockheed Missiles and Space Company (SRD) (August 7, 1970), 1 p. and enclosure W-9-796, A99-019, 79-9.

¹²⁷USAEC, H. C. Donnelly, Mgr., Albuquerque, N.M. to C13/LASL, N. E. Bradbury, Dir. et. al. (SRD) (August 6, 1970), 4 pp., A99-019, 214-15.

¹²⁸Herman P. Deinken to Mr. H. C. Donnelly, Subject: "Joint LASL/SLA Design Information for Long Range Planning," (U) (September 10, 1970), 1 p., A99-019, 214-15.

complained that the LASL attendees had been given little opportunity to make presentations. Sandford suggested that perhaps the next meeting could be held at Los Alamos.¹²⁹

A rather comprehensive document titled "Planning Information Document for Underseas Long-Range Missile System (ULMS)" was issued in the final quarter of 1970 by the Albuquerque Operations Office. This document incorporated the information contained in W-9-820. The document noted that the ULMS was a submarine-launched missile system that included a new design submarine as the launch platform and a new longer-range missile.

(b)(3)

The layout of the proposed warhead, including material quantities, is shown in Figure 2 of the Operations Office report. The report includes other details of the design and further detailed information.¹³⁰ [Author's note: The reader will note that this design historically is applicable to the W78.] Almost a year later, on October 8, 1971, Herman P. Deinken, Assistant Group Leader of W-9, would write Donnelly informing him that the LASL staff had reviewed the planning document and, as a result of updated information, the staff wanted several changes incorporated.¹³¹

On January 13, 1971, F. Baker, C. Gillespie, E. Bernard from LASL, and B. R. Emrick from SLA attended a meeting at LMSC to discuss the progress of the concept paper on ballistic RBs for the ULMS. The group was informed that the study had been completed and was scheduled for distribution. The purpose of the study had been to determine what yields, hardness, heat shield, and related components should be studied in detail if ballistic RBs were chosen for the ULMS mission. Another concept paper was being prepared by staff at LMSC. It was also noted during this meeting that there appeared to be increasing Navy concern over boat survivability. As a result, a Poseidon Extended Capability Study (PECS) was to be made. This study would include a feasibility study of the possibility, in order to increase the range of the missile, of removing the four central RBs carried on the present missile and replacing them with an additional bus engine.¹³²

On February 9, 1971, Baker again visited LMSC for the purpose of discussing the PECS study. He was informed that the PECS had turned into the EXPO (Extended Poseidon Operation.) The scope of the study had been enlarged to include consideration of a new warhead as well as a new heat shield on the RB. The main objective of the study was to show how best to fill the gap between the present 1970 Poseidon and the proposed 1980 ULMS system. The two warhead options being considered for EXPO were to retain the W68 or to develop a new warhead, (b)(3)

¹²⁹T. A. Sandford to Jesse Aragon, Subject: "Third Reentry Committee Fleet Ballistic Steering Task Group Meeting (U)," GMX-3-9191 (SRD) (November 13, 1970), 4 pp., A99-019, 82-23. ¹³⁰"Planning Information Document of Undersea Long-Range Missile System (ULMS)," AEC Albuquerque Operations Office document PID-9-00 (SRD) (October 1970), 24 pp., A99-019, 214-15. ¹³¹Herman P. Deinken to Mr. H. C. Donnelly, Subject: "Long Range Planning Warhead Data for HYB, ULMS and SCAD," (SRD) (October 8, 1971), 3 pp., A99-019, 214-15.

¹³² F. B. Baker and C. M. Gillespie, W-9 to L. C. Horpedahl, W-9, Subject: "Trip Report: Lockheed Missiles and Space Company (LMSC), ULMS Discussion - January 13, 1971," W-9-943 (SRD) (January 25, 1971), 3 pp., A99-019, 307-5.

LA-14066-H

CREWEN/ NULASSIFIED

VI-31

>

Stechtruck

(b)(3) The extended range for EXPO would come by replacing the four central RBs in the C-3 bus with a 3,000-pound motor. There was also consideration being given to modification of the submarine to carry a 37-foot missile as compared to the present 34-foot missile.¹³³

The fourth Reentry Committee Meeting of the Fleet Ballistic Missile Steering Task Group was held at the LMSC facility in Sunnyvale, California, on March 30, 1971. During the meeting, C. M. Gillespie, representing the Los Alamos Laboratory, discussed the Los Alamos program relevant to the ULMS program.

(b)(3)

Hugh R. Lehman (W-9) discussed his vulnerability work and noted that he felt that the neutron vulnerability of nuclear weapons was an important issue. He reported, "...in the ULMS era some sort of terminal [USSR] ABM system may exist in large numbers sufficient perhaps to exhaust our surviving retaliatory RBs."¹³⁵

Baker returned to LMSC for a visit on April 14, 1971. After his visit, he reported that consideration of a new warhead in the EXPO study had been eliminated; only the W68 in a new RB would be considered in the final report. This decision had come from the SSPO office in Washington. Baker noted that the restriction to the use of the W68 warhead was probably more political than technical. However, in terms of time constraints, Baker reported, "...there is time to make a case for a new warhead to meet this IOC [Initial Operational Capability] of 1976." Baker included in his trip report a list of questions that should be addressed if a new warhead were to receive any serious consideration for use in an EXPO missile. Baker also reported that the design constraint in the Poseidon envelope would permit a new second-stage motor and an increase of about 2 feet in the overall length of the missile.¹³⁶

On April 27–29, 1971, C. M. Gillespie attended the meeting of the Advanced Technology Working Group for Sea-Based Deterrence. He noted in his trip report that he was replacing Ralph Williamson as the Los Alamos representative. Gillespie indicated that the Navy was aiming for approval of a Development Concept Paper for the ULMS by January 1, 1972. Gillespie reported that this paper would outline "what the system is all about and how one is going to get it." The proposed project was not cheap—the R&D costs alone for the ULMS system including boats, missile, guidance, and related equipment was estimated at 1,900 million

¹³³F. B. Baker to L. C. Horpedahl, Subject: "Trip Report: Visit to Lockheed Missile and Space Company (LMSC). Poseidon Extended Capability and ULMS, February 9, 1971", W-9-970 (SRD) (February 17, 1971), 2 pp., A99-019, 307-5.

(b)(3)

¹³⁵Hugh R. Lehman, Group W-9 to Kaman Sciences Corporation, Subject: "Transmittal of Presentation," (SRD) (April 21, 1971), 1 p., plus enclosure W-9-1040, A99-019, 82-23.

EXPANIEN

CLASSIFIED

LA-14066-H

¹³⁶F. B. Baker to L. C. Horpedahl, Subject: "Trip Report – Visit to LMSC on EXPO, April 14, 1971," W-9-1044 (SRD) (April 26, 1971), 3 pp., A99-019, 307-5.

dollars. In comparison, a Poseidon 640-class boat with a load of missiles cost approximately 240 million dollars.¹³⁷

On May 5, 1971, Baker once again visited staff at LMSC to provide data on Los Alamos proposed sealed warhead designs for the EXPO/ULMS systems

(b)(3)

details of the design were going to be included in a letter to LMSC personnel.

Baker explained that

(b)(3)

This letter was then followed up by a visit on June 3 to LMSC by Baker. The Baker presentation was an attempt to convince the LMSC group that the LASL base-line proposal was much better than the Livermore W68 proposal.¹⁴⁰

(b)(3)

The LASL policy leaders must have continued to be concerned that the Livermore establishment was "hard selling their programs" to the Navy. In a memo dated June 4, 1971, Herman P. Deinken, W-9 Assistant Group Leader, reported that several weeks ago Chuck McDonald and others from Livermore had given a briefing (dog and pony show) to the Navy. Deinken wondered what should be done about this and he asked, "Should we do something better? What?"¹⁴² [Author's note: Again, in a memo dated August 9, 1971, Deinken would note that the Los Alamos group needed to improve the Los Alamos contacts in DDR&E (Director of Defense Research and Engineering).¹⁴³]

As to the status of EXPO, by the summer of 1971, its mission remained that of extending the range of the Poseidon Missile System as a counter to the rapidly expanding Soviet capability for antisubmarine warfare. It was reported that EXPO could have an IOC (Initial Operational Capability) as early as 1976. EXPO was a competitor with the ULMS, but it was at the same

¹³⁷C. M. Gillespie, W-9 to Distribution, Subject: "Meeting of the Advanced Technology Working Group for Sea-Based Deterrence on April 27-29, 1971," W-9-1053 (SRD) (May 5, 1971), A99-019, 307-5.

(b)(3)

⁷⁴⁰F. B. Baker, W-9 to L. C. Horpedahl, W-9, Subject: "Trip Report LMSC on EXPO/ULMS June 3, 1971." W-9-1088 (SRD) (June 9, 1971), 2 pp., <u>A99-019, 307-5</u>.

(b)(3)

¹¹⁴²Herman P. Deinken to Distribution, Subject: "Pentagon Briefing by LRL and SLL," (U) (June 4, 1971), 1 p., A99-019, 198-12.

¹⁴³H. P. Deinken to R. G. Shreffler, W-DO, Subject: "Report on Nuclear Warhead Development Guidance Conference (NWDG Conference)," W-9-1148 (SRD) (August 8, 1971), p. 3, A99-019,

LA-14066-H

\$FAFEERE UNCLASSIFIED

time a major step in developing the technology base for the more advanced ULMS. It was reported that a study effort for this type of system was being undertaken by Lockheed personnel for the Navy's Special Projects Office. Although the base-line design for the EXPO reentry body centered on the use of the W68 warhead, the Los Alamos group, in a memo dated July 27, 1971. continued to note that Los Alamos could provide a smaller, lighter device.

(b)(3)

A delegation from Los Alamos consisting of F. B. Baker, C. M. Gillespie, E. W. Salmi, R. W. Olwin, B. E. Hoverson, and T. P. Seitz visited LMSC on July 29-30, 1971. They were informed that very little had been done on the EXPO study since their last visit to LMSC on June 3, 1971. Lockheed personnel were waiting for word from SSPO on what further studies were required. However, in the study as then envisioned, there would be a third-stage motor in the bus, and the W68 warhead would be used. The ULMS IOC had slipped to 1981.

(b)(3)

[Author's note: Sometime shortly after this visit to Lockheed, it appears that the EXPO and ULMS programs were merged into one program.]

4. A New Ballistic Missile System

On September 14, 1971, the Secretary of Defense directed the Navy to initiate development of an advanced-technology, submarine-launched ballistic missile with improved propulsion and improved penetration ability to include MIRV. (This directive also included, if possible, an option for a maneuvering reentry body.) The goal of the program was to develop a missile that could replace the Poseidon C-3 system in the 1977–79 time frame and be usable with the new ULMS submarines,

The Navy was to continue development of the new Trident submarines.¹⁴⁰

In a memo (dated September 23, 1971) to his Los Alamos associates, W-9's C. M. Gillespie reported that it appeared that a Phase 2 on EXPO/ULMS would be forthcoming in the next few months. He noted that Deputy Secretary Packard had sent a Program Decision Memorandum to the Secretary of the Navy. The Packard memo had requested that the ULMS be implemented in two phases. Phase 1 would be the development of a new missile. Initially, this missile was to be deployed in Poseidon boats. The Initial Operational Capability would be in the late 1970s. Phase 2 would introduce a new submarine. The projected time for the deployment of this new

(b)(3)

¹⁴⁶ Report of Impact and Capabilities Study for the Mk 400 Reentry Body for Poseidon C-4 and Trident (Previously ULMS)," Draft, AEC Albuquerque Operations Office report I&C No. 5-72 (SRD) (July 1, 1972), p. 18, B11, Drawer 56, Folder 1 of 4.

FREEPHER

UNCLASSIFIED

LA-14066-H

UNCLASSIFIED SECTEDITIES

The Packard

/For

boat was to be six years after the IOC of the missile.

(b)(3)

memo had also asked the Navy to consider an optional RB that had a maneuvering capability.¹⁴⁷

(b)(3)

Author's note: This was good news for the Los Alamos group. It appeared that a Los Alamos proposal was being used in the RB design studies.]

There was also the Small Evader Experiment (SEE). During the October 20, 1971, meeting of the WWG, Gillespie informed the attendees that SEE was a minimum-weight, minimum-size maneuvering warhead for Poseidon. General Electric personnel planned to have the SEE project in a study phase through December; flight tests were to begin as quickly as possible after that.

Gillespie reported that the LASL group felt (b)(3) that they had a better design for this application. Gillespie also reported on the EXPO program. He noted that EXPO had, in essence, been renamed ULMS. The Phase 1 would require a new missile for the Poseidon boat; two alternative missile developments were desired.

(b)(3) this system, the Navy's SSPO was trying to avoid the usual Phase 2 program by repackaging the W68 as the warhead. The second alternative missile development would be concerned with a missile that would carry a maneuvering reentry system similar to SEE.¹⁴⁹

On October 20, 1971, a meeting was held at the Sunnyvale offices of LMSC. T. P. Seitz and F. B. Baker from W-9 were the Los Alamos attendees. Their trip report indicates that the discussions centered on the status of the ULMSJ

(b)(3)

The reentry body was limited to a weight of 170 pounds, a length of 65 inches. and a diameter of 15.8 inches.

(b)(3)

¹⁴⁷C. M. Gillespie, W-9 to Distribution, Subject: "Imminent Phase 2 on EXPO-ULMS," W-9-1185 (SRD) (September 23, 1971), 1 p., A99-019, 214-15.

(b)(3)

¹⁴⁹⁷ Weapons Working Group, Minutes of the 234th Meeting," WWG-234 (SRD) (October 20, 1972), pp. 10-12, A99-019, 92-18.

LA-14066-H

ALASSIFIED

(b)(3)

5. Approval for a Mk 400

Donnelly in the AEC's Albuquerque office, on November 18, 1971, sent a TWX to various DMA, SSPO (Strategic Systems Project Office) and AEC Laboratory representatives. Donnelly noted that his TWX included a report on the Navy SSPO meeting that had been held at Livermore on November 12, 1971, where he had obtained the latest information on the status of the ULMS program. The Deputy Secretary of Defense Packard had approved development of the ULMS 1 system. This system was to be used in Poseidon boats. For the missile, a new third stage would be included that would incorporate, to increase range, an added motor.

(b)(3)

The Navy had received funding. LMSC was expected to soon be under contract for the development of the ULMS 1. The division of responsibilities between the AEC and Navy was expected to be similar to those that had been assigned in the Mk 3 RB program. At the time of the Donnelly TWX, the possible alternatives for the warhead were (1) W68, (2) a higher-yield warhead with the same envelop as the W68, and (3) a warhead with a yield similar to the W68 but lighter in weight. The TWX also stated that the RB that resulted from the trade-off studies would be designated the Mk 400; this RB might or might not completely replace the Mk 3 depending on the studies to be performed. (Another type of RB, to be designated Mk 500, might also be considered for deployment later. For this application, evader RBs would be studied.) In terms of the time frame for the ULMS Mk 400 program, the following information had been given to the responsible groups. The Phase 2 report publication date was targeted for April 1972; the Phase 3 request to the AEC was to be made by April 15, 1972, and the selection of the AEC laboratory for the Phase 3 should be announced by July 1, 1972. The IOC had a date of December 1977. Donnelly ended his TWX by stating, "Our immediate need is the Phase 2 authorization, or in its absence, instruction relative to further participation in the program. Please advise at an early date."¹⁵² [Author's note: On December 1, 1972, Giller replied to the Donnelly TWX and informed him that the ULMS program was still in Phase 1.153 Giller thereby indicated that he would take no further action until formal authorization was received for a Phase 2 or a similar request.]

As well as the Donnelly TWX, there was also a report from Olwin on the November 12 meeting. In a trip report dated November 16, 1971, Olwin reported to his associates at Los Alamos that on November 12, 1971, he had attended the Navy briefing on the ULMS. In his trip report, Olwin noted that Commander Stinner had opened the meeting by stating that the money for ULMS-1 had been released and was now available to SSPO (Strategic Systems Project Office). Lockheed Missiles and Space Company was to be the prime contractor for the

(b)(3)

¹⁵²USAEC, H. C. Donnelly, Albuquerque, N.M. to ZEN/BY1/USAEC, Maj. Gen. E. B. Giller, DMA et. al. (SRD) (November 18, 1971), 5 pp., A99-019, 214-15.

¹⁵³USAEC, Edward B. Giller, Wash., D.C. to ZEN/AN3, USAEC, H. C. Donnelly, Albuq., N.M. et. al. (OUO) (December 1, 1971), 1 p., A99-019, 214-15.

UNCLASSIFILD

LA-14066-H

Navy. The company was to have their contract for this assignment in hand no later than November 19, 1971.

(b)(3)

warhead, a letter outlining the possible options had been sent to the DMA (Division of Military Application) from the DDR&E (Director of Defense Research and Engineering) seven weeks ago

(b)(3)

During the meeting, it had been decided that the Impact and Capabilities (I&C) study should to be completed by July 1, 1972.

(b)(3)

The speaker had then outlined what the characteristics of the new reference vehicle were.

(b)(3)

'Olwin in his trip report notes, "These

As to the

"numbers are different from those LMSC has previously shown. In the earlier version bodies containing LASL WH's were lighter than the LLL counterpart. An effort to resolve this question will be made next week at a meeting at LMSC." Olwin also wrote, "There are several things that LASL should start immediately to put us in the best possible position by the time the Phase 3 ______ award for warhead development is made."

(b)(3)

Olwin felt that a project manager should be

appointed immediately for the project in order to push the project as much as possible. Olwin felt that a reminder to General Giller was in order, concerning his commitment to LASL that the next strategic offensive system would be awarded to LASL. Olwin ended his trip report by stating, "It is not at all clear that we can break into the Navy-LMSC-LLL complex, but the benefits to be derived are well worth the short-term effort to be expended."¹⁵⁵

As well as Olwin from Los Alamos, T. A. Sandford and C. M. Gillespie had also attended the November 12, 1971, meeting. In addition, T. P. Seitz and P. Vander Maat were present. However, Sandford and Gillespie were not as prompt with their report of the meeting as Olwin had been. In their trip report dated December 3, 1971, they included information similar to that provided by Olwin. However, Sandford and Gillespie noted that a letter requesting a Phase 2

(b)(3)

¹⁵⁵R. B. Olwin to R. G. Shreffler, Subject: "Navy/LMSC Briefing on ULMS-1, November 12, 1971," W-10-71-356 (SRD) (November 16, 1971), 3 pp., A99-019, 307-8.

LA-14066-H

/SA/CARA/AMENE TINCL ASSIFIED

study had been prepared but was still "circulating in the Navy chain of command." They too reported that the base-line WH was the W68. Their trip report notes, "A new WH will not be seriously considered unless it allows the RV to be substantially (how much is not known) lighter, smaller, or to carry a higher yield." [Author's note: Thus, if LASL were to receive this much-desired assignment they were going to have to come up with, and sell, a much different warhead than Livermore's W68.] The Sandford/Gillespie trip report also noted that the requirement that was going to be part of the MCs (Military Characteristics) was that the RV had to survive a 66-foot drop onto a submarine deck. (The Navy had wanted a 110-foot requirement, but they had been talked out of it.)¹⁵⁶ For additional information on the Donnelly and LASL reports, the reader may refer to the cited documents.

There were several official documents issued during November. The NAVPRO Document No. 8822 (006), dated November 12, 1971, reported on the stockpile-to-target sequence. The minutes of the Mk 400 AEC/DOD Coordination Meeting held on November 12, 1971, at Livermore and the Kaman Science document K-22896 dated November 15, 1971, reported on _ program guidelines and the draft Military Characteristics.¹⁵⁷

(b)(3)

/The team then went on to specify the preliminary details of the design. Final specifications of the design were to follow after the required GMX field tests and the more detailed design studies had been completed.¹⁵⁸

In a memo also dated November 17, 1971, Assistant Director for Weapons MacDougall indicated that it appeared that a request for a Phase 2 study for the warhead for the ULMS would be made in the near future. Milt Gillespie from W-9 had been designated as the coordinator of this study. ¹⁵⁹ The Mk 400 Phase 2 study was again discussed during the WLPC meeting of November 22, 1971. The meeting minutes report, "It is admitted that the chance of our being selected to do the Phase 3 job for the warhead for the Mk 400 RB is small, but there are a number of reasons why we must participate and do a credible job on the Phase 2 study. ...The schedule suggested by the Navy for the Mk 400 Phase 2 study is very short and perhaps not entirely realistic, but we will do our best."¹⁶⁰

¹⁵⁶T. A. Sandford and C. M. Gillespie to Distribution, Subject: "ULMS Meeting (U)," GMX-3-9870 (SRD) (December 3, 1971), 5 pp., A99-019, 214-15.

¹⁵⁷"Report of Impact and Capabilities Study for the Mk 400 Reentry Body for Poseidon C-4 and Trident (Previously ULMS)," Draft, AEC Albuquerque Operations Office report I&C No. 5-72 (SRD) (July 1, 1972), p. 6, B11, Drawer 56, Folder 1 of 4.

(b)(3)

¹⁵⁹D. P. MacDougall to Members, WLPC, Subject: "Topic for Next Meeting," ADW-162 (CRD) (November 17, 1971), 1 p., A99-019, 214-15; also in A99-019, 91-11.
 ¹⁶⁰D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 78, November 22, 1971," ADW-166 (SRD) (November 23, 1971), p. 1, A99-019, 91-11.

VI-38

SECRETARIA UNCLASSIFIED

LA-14066-H

l

The Navy awarded the prime development contract to LMSC before the end of November. By that time, the Navy SSPO office had outlined a development schedule that included Phase 2 approval by SSPO by February 1972 and a Phase 3 request to DMA by April 15, 1972. The FPU (First Production Unit) was scheduled for November 1976.¹⁶¹

On December 7, 1971, W. J. Rudy and J. E. Gordon from Kaman Sciences Corporation (KSC) met at Los Alamos with several of the LASL theoretical staff. The reason for the meeting was for Rudy and Gordon to outline the Mk 400 hostile environment as currently being recommended by LMSC and KSC. The KSC representatives indicated that they would return to Los Alamos early in January to receive comments prepared by the Los Alamos group. The KSC staff would then circulate a final draft report. They would return once again in February for comments on this draft report before they published the completed hostile environment report.¹⁶²

During the December 8, 1971, meeting of the WWG, Hoyt listed some of the shots proposed for FY73.

(b)(3)

Also, there was a meeting at LMSC on December 8, 1971. Meeting with the LMSC representatives were F. B. Baker, F. W. Kramer, B. E. Hoverson, and R. B. Olwin from Los Alamos and P. L. Brown, R. L. Alvis, and W. R. Green from SLA, it was reported that three topics were discussed, (1) Los Alamos/Sandia input data for the C-4/Mk 400 study. (2) the C-4/Mk 400 Design Matrix, and (3) the enemy defense model

(b)(3)

fit was

reported that reentry body designs based on these were being prepared by the LMSC staff.¹⁶⁴ After this meeting, drawings containing warhead data for the proposed Mk 400 were sent from Sandia/Los Alamos to Lockheed personnel.¹⁶⁵

A new planning information document, a successor to the one published in the latter part of 1970, was issued by the AEC in December 1971.

(b)(3) [Author's note: However, the cover letter for the document indicated that "the nuclear system described is not applicable to ULMS I."¹⁶⁶] It was reported that a yield for the Mk 400 RB had not yet been selected. As to the ULMS

¹⁶¹"Nuclear Technology and Analysis Report (U)," Headquarters Field Command Defense Nuclear Agency, Kirtland Air Force Base, New Mexico 87115 (SRD) (December 1, 1971), pp. 65–66, A99-019, 192-9.

¹⁶²K. F. Famularo, TD-4 to Distribution, Subject: "ULMS-1 Meeting with Kaman Sciences Corporation (KSC) at LASL on December 7, 1971," TD-4-71-367 (SRD) (December 9, 1971), 7 pp., A99-019, 214-15.

(b)(3)

¹⁶⁴F. B. Baker, F. W. Kramer, B. E. Hoverson, and R. B. Olwin to H. P. Deinken, W-9, Subject: "Trip Report C-4/Mk 400 (Formerly ULMS-1) Meeting at LMSC December 8, 1971," W-9-1285 (SRD) (December 16, 1971), 5 pp., A99-019, 307-8.

¹⁶⁵C. H. Mauney to Naval Plant Representative Office, Sandia document RS 1500/1784 (SRD) (February 25, 1972), 2 pp., B11, Drawer 56, Folder 1 of 4.

¹⁶⁶Vincent C. Vespe, Director Weapons Development Division to Distribution (SRD) (December 1, 1971), 1 p., A99-019, 214-15.

LA-14066-H

FERRES/REA

VI-39

UNCLASSIFIED



mission, this document stated, "The Undersea Long-Range Missile System (ULMS) is a submarine launched missile system similar in concept to the Polaris and Poseidon Fleet Ballistic Missile system.

(b)(3)

On December 28, 1971, the Secretary of Defense directed effort toward the first deployment of the Trident submarines in the calendar year 1978.¹⁶⁸

6. At Last, a Phase 2

John S. Foster, Director of Defense Research and Engineering (DDR&E), in a January 12, 1972, letter addressed to the Chairman of the AEC, Dr. James R. Schlesinger stated, "The Navy has been directed to undertake the development of a new advanced technology submarine-launched ballistic missile with improved propulsion and improved penetration ability, including MIRV and, if possible, an option for a maneuvering re-entry body.

(b)(3) studies on possible warhead designs had been completed. Foster indicated that although the W68 warhead might satisfy the conceptual requirements, the Department of Defense would welcome proposals for a reduction of the warhead weight and volume while maintaining the same yield as the W68. Also acceptable was a design with increased yield over that of the W68 while maintaining the same size and volume. Foster stated, "Furthermore, we would like to develop a better understanding of the range of tradeoffs available between warhead yield and weight/size of re-entry body and warhead, before we establish firm characteristics for the ULMS I re-entry vehicle and warhead." Foster then stated, "It is requested that AEC participate with the Department of Defense in a Phase 2 weapon feasibility study to be initiated as soon as possible to protect an IOC date in late 1977." Foster hoped that the Phase 2 study could be completed by March 20, 1972. Foster reported that the Navy had established a separate project office designated PM-2 for the development of the ULMS. To undertake the Phase 2 study and the follow-on program a project officers group was to be convened by the Navy's Strategic Systems Projects Office (SSPO).¹⁶⁹ [Author's note: The new missile would carry the designation C4. The missile that carried the W68 was designated C3.]

On January 13, 1972, there was a Mk 400 AEC/DOD coordination meeting. During the meeting it was noted that Foster has signed the Phase 2 request on January 12/

(b)(3)

A draft STS (Stockpile-to Target Sequence) had been distributed.

HARANNA (

(b)(3)

¹⁶⁸"Report of Impact and Capabilities Study for the Mk 400 Reentry Body for Poseidon C-4 and Trident (Previously ULMS)," Draft, AEC Albuquerque Operations Office report I&C No. 5-72 (SRD) (July 1, 1972), p. 18, B11, Drawer 56, Folder 1 of 4.

¹⁶⁹John S. Foster, Jr. to Dr. James R. Schlesinger (SRD) (January 12, 1972), 2 pp., Appendix A of "Joint AEC/DOD Mk 400 Phase 2 Feasibility Study Report," (SRD), B11, Drawer 56, Folder 1 of 4; also available in B11, Drawer 56, Folder 3 of 4.

VI-40

LA-14066-H

Several

AA, CABARINGA

(b)(3)

In a letter dated February 1, 1972, to Foster, Edward B. Giller, Assistant General Manager for Military Application, stated, "The Atomic Energy Commission is pleased to cooperate with the Department of Defense and participate in a Phase 2 weapon feasibility study for the purpose of developing the new ULMS I ballistic reentry vehicle warhead as requested by your letter of January 12, 1972." However, Giller did not feel that Foster had allowed adequate time for the completion of a Phase 2 report. Giller noted, "Accordingly, it is suggested that a Phase 2 feasibility study be conducted as soon as the draft MC's and STS can be made available to the AEC and that a completion date of July 1, 1972, be set." He also commented, "In order to conduct the pre-Phase 3 meetings, I recommend the designation 'mini-POG' for the group."¹⁷¹

In addition to his letter to Foster, Giller also sent out a memo on February 1, 1972, to the manager of the Albuquerque Operations Office and to the various laboratory directors. He noted that Foster had been informed that the Phase 2 feasibility study could not be completed as soon as Foster had wanted, Giller estimated that the most realistic date was July 1, 1972. Giller instructed that work should continue on the Impact and Capability Study (I&C Study). Giller stated, "The ALO portion of the I&C Study should be completed before ALO signature release on the final Phase 2 study. Giller then requested, "Please keep me advised of your progress and of key meeting dates."¹⁷²

Despite the fact that Giller felt that the Foster-proposed Phase 2 date was unreasonable, nevertheless a quick response from the Laboratories to provide the necessary information to those preparing the Phase 2 would be required in order to meet the Giller proposed summer deadline. Realizing that Los Alamos was going to have to provide information for the Impact and Capability (I&C) Study and for a draft Phase 2 report, Gillespie sent a memo dated February 7, 1972, to MacDougall outlining the requirements. Gillespie noted that the LASL warhead data needed to be at Lockheed by February 14. The information for the Phase 2 package needed to go to SSPO by February 28. The input for the I&C Study also needed to go to ALO by February 28. In order to respond in this limited time, the Los Alamos group was going to do one basic document to serve for both the Phase 2 input and the I&C input. But Gillespie noted that the Los Alamos group was still several days away from having "even a rough first draft." However, Gillespie provided MacDougall with a proposed outline of the report. In turn, MacDougall sent a memo dated February 8, 1972, to the LASL management with the Gillespie memo attached and urged those responsible to meet the schedule. MacDougall noted, "Should problems develop which threaten to slip the time scale shown in the attached memo, I would like them brought to my immediate attention."¹⁷³

(b)(3)

¹⁷¹Edward B. Giller to Honorable John S. Foster, Jr. (SRD) (February 1, 1972), 2 pp., Appendix B (SRD) of "Joint AEC/DOD Mk 400 Phase 2 Feasibility Study Report," B11, Drawer 56, Folder 1 of 4.

¹⁷²Edward B. Giller to H. C. Donnelly, Manager, Albuquerque Operations et. al., Subject: "DOD Request for a Phase 2 Study of the ULMS I Ballistic Reentry Vehicle Warhead," (SRD) (February 1, 1972), 2 pp., B11, Drawer 56, Folder 3 of 4.

¹⁷³D. P. MacDougall to Distribution, Subject: "W-9-1344 (attached)," (SRD) (February 8, 1972), 1 p. and attachment, B11, Drawer 56, Folder 3 of 4.

LA-14066-H

UNCLASSIFIED

In an attachment to a cover memo dated February 15, 1972, Gillespie sent the various people in the LASL administration the rough draft of the proposed Los Alamos input to the Mk 400 Phase 2 submission. He wanted comments back by February 17. Gillespie wrote, "I am available day or night for discussion." The draft indicated that the LASL base-line proposals were environmentally sealed/integrated packages. This type of engineering resulted in the device having several advantages over the current Mk 3 (W68) warhead. Because the LASL/SLA proposed warhead/arming and firing systems were constructed as a unit, there was much more flexibility in terms of both the final assembly of the reentry body and in any required repairs. Sealing the warhead meant simplification in the fabrication assembly and certification. The design allowed field exchange of the reservoir.¹⁷⁴ [Author's note: One of the reasons for Los Alamos pushing this aspect of their designs was the fact that the Navy was not happy with the difficulty of making limited life exchanges or repairs in the W68.]

During the WWG meeting held on February 16, 1972, MacDougall reported that the Laboratory was currently "deeply involved in trying to prepare the Phase 2 input for the MK 400 warhead for ULMS." The meeting minutes report, "He asked that, since this is the most likely chance we have at a new Phase 3, every effort be devoted to helping Gillespie collect the information he needs for the [Phase 2] package."¹⁷⁵ [Author's note: From this directive by MacDougall, it is clear that the upper management at Los Alamos had decided to fight hard for the Mk 400 warhead assignment.]

On February 17, 1972, Olwin sent out a memo that outlined what the LASL groups were doing in terms of input for the vulnerability requirements. He noted that there was to be a meeting in Washington on March 2 to discuss the hostile environments' section of the STS (stockpile-to-target sequence) document. Olwin noted that Ken Famularo had already met with several of the staff at Kaman Science. George Spillman had also provided relevant information.¹⁷⁶

On February 23, Gillespie sent out to the Los Alamos group (through W-Division leader R. G. Shreffler) the final draft of the Los Alamos Mk 400, Phase 2 input. Gillespie stated, "Assuming there are no major changes we can work over the weekend and get it on a plane Monday afternoon."¹⁷⁷

In a February 25 letter to the Navy, C. H. Mauney of the Systems Development Department of Sandia noted that the LASL/SLA warhead proposals would be contained in the Phase 2 input submitted to SSPO on February 28, 1972. These proposals would be similar to those already submitted to LMSC. Mauney reported, "Since the warhead and RB are a tightly integrated system, we have worked closely with LMSC to arrive at the best possible warhead design."¹⁷⁸

VI-42

AFCHET AND DNCLASSIFIED

¹⁷⁴C. M. Gillespie, W-9 to Distribution, Subject: "Mk 400 Phase 2 Input" (SRD) (February 15, 1972), 1 p. and enclosure, B11, Drawer 56, Folder 3 of 4.

¹⁷⁵"Weapons Working Group Minutes of the 237th Meeting," WWG-237 (SRD) (February 16, 1972), p. 8.
¹⁷⁶R. B. Olwin, W-10 to Distribution, Subject: "Mk 400 Vulnerability Subcommittee Meeting,"
W-10-72-60 (SRD) (February 17, 1972), 2 pp., B11, Drawer 56, Folder 3 of 4.

¹⁷⁷C. M. Gillespie, W-9 to Distribution, Subject: "Mk-400 Phase 2 Input," (SRD) (February 23, 1972), 1 p. and enclosure, B11, Drawer 56, Folder 3 of 4.

¹⁷⁸C. H. Mauney to Naval Plant Representative Office, Subject: "LASL/SLA Proposals for the C4/Mk 400 Phase 2 Study," RS 1500/1784 (SRD) (February 25, 1972), 2 pp., B11, Drawer 56, Folder 1 of 4.

In a February 28, 1972, memo, MacDougall reported that Giller was anticipating that a Phase 3 request for a ULMS might be initiated in the FY 1972–1974 time period. At the same time, Giller did not want any work that did not directly support either present or anticipated future weaponization requirements to receive a high priority.¹⁷⁹

The W-9 report dated February 28, 1972, titled "LASL and SLA Input to the ULMS MK-400 Ballistic Reentry Body Phase 2 Feasibility Study" was released as W-9-1350. As previously noted, this report served as the input from Sandia and Los Alamos both on the Phase 2 and the Impact and Capabilities studies

(b)(3)

This document was sent to Commander R. J. Stinner at the Strategic Systems Projects Office. In his cover letter to Stinner, Gillespie noted, "Signature of the final Phase 2 Feasibility Study is dependent upon review and approval of the complete Phase 2 Study, including the RB synthesis work by LMSC." Additional information on the LASL study is available in the cited reference.¹⁸¹

On February 29, 1972, Giller sent AEC Chairman Schlesinger and the other four commissioners, a summary of the ULMS-1 program. Giller reported that the Secretary of Defense had, on September 14, 1971, directed the Navy to develop a new submarine for deployment in the early 1980s. At the same time, a new missile system was to be developed that was also to be compatible with the present Poseidon-type submarines. On December 28, 1971, the Secretary of Defense had directed that the effort be aimed at delivering the first ULMS submarine in December 1977. The Giller memo then outlined the program in some detail. It is interesting to note that Giller indicated that the missile would require a new reentry body/warhead combination, primarily because the current Mk 3 RB was not designed for the more severe flight environment that the new RB would be subjected to. Other design objectives were a lighter/smaller RB, higher warhead yield, and improved hardening. Giller discussed the nuclear material implications.

(b)(3)

reported on what the various Navy groups were doing or had an interest in. He noted that Rear Admiral Harvey Lyons, the ULMS program manager, was concentrating on hull-design technology.

(b)(3)

Giller reported, "They are apparently willing to pay some penalty in payload weight to achieve the higher yield."¹⁸²

¹⁷⁹D. P. MacDougall to Distribution, Subject: "Program Planning," ADW-204 (SRD) (February 28, 1972), 3 pp., B11, Drawer 49, Folder 1 of 5.

(b)(3)

¹⁸¹C. M. Gillespie to Cdr. R. J. Stinner, Subject: "Transmittal of LASL/SLA Input to the Mk-400 Phase 2 Feasibility Study, W-9-1350" (SRD) (February 28, 1972), 1 p. and enclosure W-9-1350, B11, Drawer 56, Folder 3 of 4.

¹⁸²Edward B. Giller to Chairman Schlesinger, Commissioner Ramey, Commissioner Johnson, Commissioner Larson, Commissioner Doeb, Subject: "Undersea Long-Range Missile System (ULMS) and ULMS-I Missile," (SRD) (February 29, 1972), 10 pp., B11, Drawer 56, Folder 3 of 4.

LA-14066-H

SACREMENTED

VI-43

Giller also



On March 8, Gillespie, in a letter to Vespe at the Albuquerque Operations Office, enclosed two drawings of the LASL design proposed for the I&C study.

(b)(3)

Vespe, at the Albuquerque Operations Office, with a cover memo dated March 31, 1972, sent the weapon laboratories and the plants the AEC input for the "AEC weapon's production system for the Mk 400 RB I&C Study."¹⁸⁴

In a W-Division review report dated March 31, 1972, it was indicated that the development of small primaries was an integral and important part of the total effort for strategic offensive systems. At the same time it was necessary, in order to achieve maximum range, to highly integrate the reentry vehicle and the warhead. By doing this integration, it was anticipated that a <u>5% to 10% improvement in yield-to-weight and yield-to-volume ratios would be possible.</u>

(b)(3)

In a memo to Shreffler in W-DO, dated May 2, 1972, R. B. Olwin from W-10 noted that there was a gloomy outlook on the Mk 400 program that had been displayed by the staff and management at Los Alamos. In order to see if this was indeed warranted, Olwin then discussed the Mk 400 program at that time

----- **h** --

(b)(3)

In all cases, the

Livermore proposals did not weigh as much

(b)(3)

Olwin

also noted in his memo, "The Navy was quite adamant with regard to the AEC supplying a scaled warhead package rather than allowing the seal to be a part of the RB as was done on the MK-3. Navy sensitivity to the problems brought about by the MK-3 was behind this directive. LASL responded by supplying a neatly sealed package whereas LLL ignored the request and proceeded to seal at the RB substructure." Olwin reported that the LMSC engineers considered the LLL/SLL proposed seal design to be a high risk.¹⁸⁶ [Author's note: In this memo, we see some aspects of what would become the LASL defense against LLL. The previous problems with the warheads supplied to the Navy by LLL would be carefully noted. The problems with replacing limited-life components would be outlined. It would be indicated that two types of warheads, one from each Laboratory, were needed; one as a backup in case there were problems

(b)(3)

¹⁸⁴Vincent C. Vespe to Those on Attached List, Subject: "Mk 400 RB I&C Study," (SRD) (March 31, 1972), 4 pp. and enclosure, B11, Drawer 56, Folder 3 of 4.

(b)(3)

¹⁸⁶R. B. Olwin, W-10 to R. G. Shreffler, W-DO, Subject: "MK 400/500," W-10-72-169 (SRD) (May 2, 1972), 5 pp., B11, Drawer 56, Folder 1 of 4.

VI-44

UNCLASSIFIED

with either one.

(b)(3)

The LASL management would try to impress on the Navy the LASL desire to work with the Navy and its contractors and to respond to all their concerns.]

In a May 9, 1972, letter to V. C. Vespe at the Albuquerque Operations Office, Gillespie reported that to meet the Mk 400 development program there was a need for six NTS tests. These tests could not yet be formalized because the final yield requirement for the Mk 400 had not yet been specified.¹⁸

In the spring of 1972, it was announced that there had been a change to a rounded aft end in the reentry body. This change would reduce the radar cross section. But the change also meant that the earlier higher-yield designs were also affected—their dimensions would need modification. The Los Alamos group indicated their changes in the Phase 2 input in a TWX dated May 16, 1972.¹⁸⁸

To provide data on cost for the Impact and Capabilities report, a TWX was sent out on May 15, 1972. This TWX requested cost estimates for the NTS tests that would be required to complete each design for the stockpile should that design be chosen for the Phase 3 program. As would be expected, the W68 proposal required the fewest number of tests. At the same time, the LASL group appears to have been very optimistic as they listed in most cases a requirement of only three tests.¹⁸⁹ These were apparently the number of required full-scale tests because a slightly earlier memo had reported that LASL would require a one-point safety test, two primary tests, and three full-yield tests in order to develop the Mk 400.¹⁹⁰ [Author's note: This information on the number of tests reported as required indicates that the LASL design team must have felt under a great deal of pressure to limit the number of tests, once the Phase 3 was given to LASL.]

During the May 17, 1972, meeting of the WWG, Canada described the LLL test program for FY73.

(b)(3)

¹⁸⁷C. M. Gillespie to Mr. V. C. Vespe, W-9-1423 (SRD) (May 9, 1972), 2 pp., B11, Drawer 56, Folder 1 of 4.

¹⁸⁸C. M. Gillespie thru D. P. MacDougall, University of California Los Alamos Scientific Laboratory, Los Alamos, N.M., to RULSSAA/Cdr. R. Stinner, Strategic Sys. Projects, Dept. of Navy, Washington D.C. et. al., W-9-1430 (SRD) (May 16, 1972), 2 pp., B11, Drawer 56, Folder 1 of 4.

¹⁸⁹USAEC Vincent C. Vespe, Dir. Wpns. Dev. Div., Albuquerque, N.M. to BP3/USAEC R. W. Taft, NVO et. al. (SRD) (May 15, 1972), 6 pp., B11, Drawer 56, Folder 1 of 4.

¹⁹⁰C. M. Gillespie to Mr. V. C. Vespe (SRD) (May 9, 1972). 2 pp., B11, Drawer 56. Folder 1 of 4.

(b)(3) STORENDO **JNCLASSIFIED**

VI-45

On June 5, 1972, David B. Anthony, Assistant Manager for Plans & Budgets at the Albuquerque Operations Office, wrote the Director, Strategic Systems Project Office in Washington. Anthony reported that the AEC was presently engaged in an Impact and Capability Study (I&C) to estimate the cost of refitting the Poseidon C3 boats with Mk 400 RBs. The AEC was also considering the possible addition of the Mk 400 RBs for the new Tridents boats. He noted that the AEC had a reactor products' availability problem in the Mk 400 program.¹⁹²

On June 16, 1972, H. N. Meyer from the Albuquerque Operations Office sent out the first draft of the Report of the Mk 400 I&C (Impact and Capabilities) Study for the Mk 400. A meeting was to be held in Albuquerque on June 22 to receive any final comments.¹⁹³

On June 23, 1972, C. H. Mauney from Sandia distributed the draft titled.

(b)(3)

The second meeting of the Mk 400 mini-POM (Project Officers' Meeting) group was held in Washington on June 28, 1972. The Navy's representative from SSPO announced that the ULMS program would henceforth be known as the Trident program. The word Trident would be used to describe that system and the various portions of the system. Then the representative made an interesting and important statement. The Trident C4 missile system was designed to be fitted into existing SSBN's (nuclear-powered submarine, ballistic-missile-bearing) as well as into the Trident submarine. The Trident II D-5 would be the name given to the follow-on missile system to be designed particularly for the Trident submarine. Existing SSBN's when back-fitted with the Trident I missile would not be classed as Trident submarines but would remain Trident I back-fit SSBN's. It was announced that the Phase 2 report for the Mk 400 had been approved, signed by all participants, and published. [Author's note: This appears to have been the final draft of the Phase 2 report.] It was expected that this document would be issued in the near future. It was also reported that the entire STS document was in the final stages of preparation.¹⁹⁵

The Albuquerque Operations Office draft of their Mk 400 I&C study, circulated in the summer of 1972, carries the date of July 1, 1972. This draft states that the AEC presently had the capability and the capacity, with modest additions, to support the Poseidon refit with an IOC of December 1977 and a Trident outfit with an IOC of October 1978. In this study, it was assumed that a Phase 3 development authorization would be received by the summer of 1972. It is interesting to note that the draft report states, "The advisability of additional production of W68 nuclear systems in the late 1970's and early 1980's is questionable." The report states, "Availability of special materials was assumed to be adequate as necessary to produce any of the designs with normal working inventories as weapons grade materials, fully compatible with

¹⁹⁵"Nuclear Technology and Analysis Report (U)," Field Command, Defense Nuclear Agency Technology and Analysis Directorate, Kirtland Air Force Base, New Mexico 87115 report FC/08720008 (SRD) July 21, 1972), p. 29, B11, Drawer 57, Folder 1 of 2.

(b)(3)

VI-46

SECTORIAN (

UNCLASSIFIED

¹⁹²David B. Anthony to Director, Strategic Systems Project Office (SRD) (June 5, 1972), p. 1, B11, Drawer 56, Folder 1 of 4.

¹⁹³H. N. Meyer to C. M. Gillespie, W-9 LASL et. al., Subject: "Mk 400 I&C Study Report," (SRD) (June 16, 1972), 1 p., and enclosure, B11, Drawer 56, Folder 1 of 4.

fabrication facilities."

LA-14066-H

(b)(3) Additional information is available in the cited reference.'

(b)(3)

The final draft of the Phase 2 feasibility study report was also released in the summer of 1972. (b)(3) In the Los Alamos files the draft study report carries a date, made in pencil, of July 1, 1972. The cover sheet for the document carries a date of June 26, 1972. As previously noted, during the second meeting of the Mk 400 mini-POM group held on June 28, 1972, it had been announced that the Phase 2 report had been approved.]

(b)(3)

The final yield of the new warhead had not yet been specified.

In the

¹⁹⁶ Report of Impact and Capabilities Study for the Mk 400 Reentry Body for Poseidon C-4 and Trident (Previously ULMS)," Draft, AEC Albuquerque Operations Office report I&C No. 5-72 (SRD) (July 1, 1972), 20 pp., B11, Drawer 56, Folder 1 of 4. Also in B11, Drawer 56, Folder 3 of 4.

(b)(3)

SALARE ABAX **JNCLASSIFIED**



sealed systems, the complete nuclear device/AF&F assembly was designed to withstand repeated launch-tube pressurization to 200 psi. In summary, the LASL/SLA group noted, "Our extensive work with LMSC prior to and during the Phase 2 effort has established that all the designs are compatible with the C4/MK400 system as presently defined. We have seriously addressed the problem of integrating the warhead into the RB and our proposals reflect a realistic approach to this problem. We believe it offers many advantages to the Navy over the MK3 approach."¹⁹⁸

In the Los Alamos files, despite the fact that it is dated July 1, 1972, the final "Joint AEC/DOD Mk400 Phase 2 Feasibility Study Report" carries on its front sheet a handwritten date of September 12, 1972. The cover letter for sending the report was written by Levering Smith at the Strategic Systems Project Office in Washington; this letter has a date of September 5, 1972. In this final study, the Los Alamos group indicated that they felt they could develop the device three and a half years after the award of the Phase 3. They foresaw no difficult development problems.¹⁹⁹

(b)(3)

Paul Vander Maat from TD-2 and K. F. Famularo from TD-4 on July 24, 1972, sent an extremely interesting memo to T. A. Sandford. [Author's note: This was probably sent in preparation for Sandford's forthcoming August trip to Washington.]

(b)(3)

ASSIFIED

LA-14066-H

(b)(3)

¹⁹⁸ 'Final Draft, Joint AEC/DOD Mk400 Phase 2 Feasibility Study Report," Strategic Systems Project Office, Department of the Navy, Washington, D.C., 20390 document K-24304 (SRD) (not available), 88 pp. with Appendix A and B, B11, Drawer 56, Folder 1 of 4.

¹⁹⁹Director, Strategic Systems Projects to Chief of Naval Operations (OP-21), Subject: "Joint AEC/DOD Trident Mk 400 Re-entry Body Feasibility Study," (SRD) (September 5, 1972), 1 p. and enclosures, B11, Drawer 57, Folder 2 of 2.

SPAREARE

On August 2, a TWX from William B. Haidler, at the AEC Office in Washington, stated "Request you supply information, by COB August 4, 1972 that indicates the yield you could expect to achieve from a warhead that could be developed in the approximate volume and dimensional constraints of the proposed Navy Mk 400 without degrading the CG. 'Static Margin,' etc. and with an increase in weight to approximately 400 pounds."²⁰

(b)(3)

On August 7 and 8, Sandford and Hoyt presented to various officials in Washington several important "update" briefings. These, in preparation for the Washington trip, had been prepared as LASL/SLA briefings during the last two weeks of July.

(b)(3)

[Author's note: It is also interesting to note that Hoyt and Sandford appear to have also mentioned to the Washington groups the advantages to using a

²⁰¹USAEC, William B. Haidler, Wash., D.C. to C13 LASL, H. M. Agnew, Los Alamos N.M. et. al. (SRD) (August 2, 1972), 1 p., B11, Drawer 56, Folder 3 of 4.

(b)(3)

LA-14066-H

FIED

16-inch base diameter in the RB. This is the base diameter used in the design of the W76.] Near the end of his trip report, Sandford included several pieces of interesting "gossip." He reported, "At DDR&E we talked to John J. Brett who seemed to react favorably to our new, higher yield proposals. At the SPO we briefed Cdr. Stinner, Mark Messerole, and Hal McMasters. They seemed relieved and/or amused that our WHs now were competitive with LLL's WHs." In addition, Sandford reported, "At DMA we briefed General Camm, Tom Clark, Col. Haidler, and several of their staff members. The briefing went very well with several pertinent questions being asked." Furthermore, Sanford reported, "The MK-400 I&C study indeed was stuck in the AEC. (b)(3)

Pending resolution of this problem, the Phase 3 request cannot be sent to DMA." It was also noted in the trip report that the question of the requirements for special nuclear material had been discussed by Tom Clark. The trip report states, "The only concrete message gleaned from Clark's talk was that one or two reactor restarts at Savannah River are feasible but the AEC does not want to restart any of the Richland reactors."²⁰³

Despite this report from Hoyt and Sanford of the delay in the I&C study, on August 17, the AEC proposed to send the AEC Impact and Capability study to Foster at the Defense Research and Engineering, Department of Defense. As to the reactor question, the proposed cover letter for the study noted, "In addition, we believe that, while restart of up to four reactors is theoretically possible, in light of decisions which have been made in the past the restart of the two standby reactors at Savannah River would be difficult enough to achieve; the restart of additional reactors at Richland might pose intractable problems, particularly in view of current environmental concerns." The letter added, "Certain adjustments to the assumed Mk 400 delivery schedules and to the assumed base stockpile might possibly reduce the number of reactor restarts required." The letter ended by stating, "In order to meet the desired IOC date of December 1977 it will be necessary for us to receive a Phase 3 decision by the end of August so that the AEC can arrange for the necessary capital funding."²⁰⁴

(b)(3)

He included attachments of the vufoils that had been used in the August 7 briefing given to Assistant DDR&E Director John J. Brett.²⁰⁵

(b)(3)

²⁰³T. A. Sandford to D. P. MacDougall, ADW, Subject: "Mk-400 and High Yield Bomb Briefings in Washington (U)," ADWP-1-72-19 (SRD) (August 17, 1972), 7 pp., B11, Drawer 56, Folder 3 of 4.
 ²⁰⁴Thomas R. Clark to J. A. Hornbeck, President, Sandia Laboratories et. al., Subject: "Mk 400 I&C Study," (SRD) (August 21, 1972), 1 p., and two attachments, B11, Drawer 56, Folder 3 of 4.
 ²⁰⁵T. A. Sandford to Dr. John S. Foster, Jr., Subject: "LASL/SLA MK-400 Briefing (U)," ADWP-1-72-20

²⁰⁻T. A. Sandford to Dr. John S. Foster, Jr., Subject: "LASL/SLA MK-400 Briefing (U)," ADWP-1-72-20 (SRD) (August 21, 1972), 8 pp., B11, Drawer 56, Folder 3 of 4.

SECTORATED

VI-50

ARAYIRA

(b)(3)

Hoyt noted that the members of the WPRC had recommended that the TD proposal be accepted.²⁰⁶

On August 28, 1972, Sandford wrote Camm to provide him the information in writing that had been provided during the August 8 briefing. He also included hard copies of the slides that ______had been used in the briefing

(b)(3)

^w designs allow a 10 to 15 pound RB weight saving, meaning that for a fixed payload yield, greater range could be realized or, for a fixed range, more yield could be delivered to the target."

(b)(3)

I'his test will incorporate the 'new' technology cited in our Mk 400 study." Sandford assured Camm that the costs of all the LASL warhead proposals were covered in the current Mk 400 I&C Study.²⁰⁷

(b)(3)

8. Where is the Phase 3?

During the WLPC meeting held on September 11, 1972, Hoyt reported that a letter had been prepared by the Navy requesting that the Mk 400 program be put into Phase 3. Hoyt noted that he was hopeful that LASL would receive the Phase 3 assignment.²⁰⁹

(b)(3)

²⁰⁷T. A. Sandford to Major General Frank A. Camm, Asst. Gen'l Manager for Military Application, Subject: "LASL/SLA MK-400 Briefing (U)," ADWP-1-72-24 (SRD) (August 28, 1972), 12 pp., B11, Drawer 56, Folder 3 of 4.

(b)(3)

²⁰⁹D. P. MacDougall to WLPC Members, Subject: "WLPC Meeting No. 99, September 11, 1972," (SRD) (September 12, 1972), 2 pp., B11, Drawer 53, Folder 1 of 2

LA-14066-H

(b)(3) LASSIFIED

VI-51

/l'hese new

(b)(3)

Additional information is available in the cited reference.²¹¹

The W-Division program report dated September 22, 1972, stated that there had been continued LASL coordination with all the Mk 400 agencies in terms of design studies, flight and ground testing, and simulation test planning. This effort should permit a smooth transition when the Phase 3 award was made. Analytical studies were currently in progress to investigate areas where the LASL group could assist LMSC in obtaining a minimum weight, integrated reentry system.²¹²

On September 25, 1972, John Foster wrote AEC Chairman, James R. Schlesinger, that the Navy had not yet forwarded their recommendations on the Phase 2 study. Foster therefore anticipated that a specific selection of a Phase 3 design would not be decided for several months. It appeared that the Mk 400 IOC might be delayed until somewhere around December 1978. However, despite all this uncertainty in the program, Foster wrote, "I am therefore suggesting that the AEC take the necessary planning actions consistent with an IOC for the Mk 400 as early as December 1977. A specific Phase 3 request will be forwarded as soon the DOD review of the Phase 2 study is completed."²¹³

The next meeting of the Mark 400 Mini-POM was held at Lockheed on September 27–28, 1972. Representing LASL were K. F. Famularo (TD-4), J. C. Fuller (WX-6), C. M. Gillespie (ADWP-1), F. W. Kramer, (WX-3), and T. A. Sandford, (ADWP-1). During the meeting, it was announced that Secretary Rush had issued a program decision directive that delayed the Mark 400 IOC by 10 months to October 1978. Thus, the Mk 400 and Trident submarine would have the same IOC dates. The directive delayed the Poseidon backfit until 1979 or 1980. This directive implied that the first C4 missiles would go on the Trident submarines instead of being backfitted into the Poseidon. The Mk 500 was canceled. The Navy's Commander Stinner reported that the Phase 3 letter was being reviewed by the Navy. In lieu of an immediate Phase 3 request, Stinner hoped that a letter dated September 25, 1972, from Foster to Camm (Major General Frank A. Camm, Assistant General Manager for Military Application) that guaranteed a Phase 3 go-ahead, would be used to designate a design laboratory.

(b)(3)

The Navy's BuMed manual

requirements for radiation safety would have to be met by the Mk 400.

(b)(3)

Representatives from Lockheed then reported on their work.

(b)(3)

/D. Aspinwall from Lockheed noted that, to eliminate range

SERVET TRU UNCLASSIFIED

LA-14066-H

²¹¹C. A. Anderson and F. W. Kramer to Jesse Aragon, Subject: "Meeting – Mark 400 Packaging Design Exercises at LASL and SLA (U)," WX-3-72-35 (SRD) (October 3, 1972), 2 pp., B11, Drawer 56, Folder 2 of 4.

²¹² W Division Program Reporting System," WX-72-6 (SRD) (September 22, 1972), pp. 138–140, B11, Drawer 56, Folder 4 of 4.

²¹³USAEC William B. Haidler, Wash., D.C. to AN3 USAEC H. C. Donnelly, Albuq., N.M. et. al. (SRD) (November 10, 1972), 6 pp., B11, Drawer 56, Folder 1 of 4.
reductions associated with nose radii greater than optimum, they were considering a selfdeploying aerospike on the nose fairing of the C4 missile.

(b)(3)

The flight test program was also discussed. The AF&F sequence of the Mk 400 was also reviewed. There were several presentations made during the meeting on the possible types of overcoats that might be used to reduce vulnerability.²¹⁴

The quarterly progress report from the Laboratory for the period ending September 30, 1972, noted that for the Mk 400 there had not been a Phase 1 program. Phase 2 had been informally started in November 1971; however, the formal direction from DMA to participate in the Phase 2 had not been received until February 3, 1972. The LASL Phase 2 data package had been submitted on February 28, 1972. Although the final Phase 2-study report had been dated July 1, 1972, the report had not been formally distributed by SSPO until September 1972. In August 1972, the LASL groups had presented to SSPO, DDR&E, and DMA representatives the new design improvements that had been made. In turn, new reentry-body weights for the new warhead proposals had been developed.²¹⁵

The Trident system was discussed at length during the October 5, 1972, meeting of the Theoretical Weapons Group (TWG). Gillespie stated that the purpose of the Fleet Ballistic Missile System was to provide deterrence through an invulnerable force capable of inflicting unacceptable damage to the Soviet urban/industrial complex. Gillespie noted that by treaty the United States was limited to 44 submarines. As of October 1972, the U.S. had 10 Polaris submarines that carried the A3 missile.

There were 31 Poseidon submarines that carried the C3 missile (b)(3) The new Trident

would carry the C4 missile/

(b)(3) Trident had an IOC of 1978. It was twice as large as the Poseidon boat and could carry the larger D5 missile.

(b)(3)

Gillespie also noted the

VI-53

importance of minimum weight in terms of maximum range. Gillespie then discussed, assuming that LASL was assigned the Phase 3 in January 1973, the future test program

(b)(3)

Several additional tests would be necessary before a final weaponized design was achieved. The minutes of the meeting state, "In closing, Gillespie outlined the reasons for supporting LASL as the recipient of the MK 400 Phase 3 in terms of past experience, the present LASL program and the availability of program support

(b)(3)

²¹⁴F. W. Kramer, K. F. Famularo, J. C. Fuller, C. M. Gillespie and T. A. Sandford to Distribution, Subject: "Mark 400 Mini-POM, September 27-28, 1972 (U)," ADWP-1-72-52 (SRD) (October 12, 1972), 4 pp., B11, Drawer 56, Folder 1 of 4.

ASSIFIED

LA-14066-H

²¹⁵Leslie M. Redman and Cecil C. Carnes, Jr., "LASL Weapons Quarterly (U), for the Period Ending September 30, 1972," Los Alamos Scientific Laboratory report LA-5130-PR (SRD) (January 1972), p. 68.

(b)(3)

Gillespie noted

LA-14066-H

that the W74 program would start to decrease in about a year. The people released from this program could then provide the extra needed support for the Mk 400 program,

(b)(3)

During the October 16, 1972, WLPC meeting, information on the Mark 400 was again presented. It was reported that SPO had prepared a letter (that was circulating for review through the Navy) that requested that the AEC put the Mark 400 into Phase 3. In addition, the DDR&E had written a letter to AEC Chairman Schlesinger. This letter noted that the Navy planned in the not-too-distant future to put the Mark 400 into Phase 3; the letter requested that that AEC select a Laboratory to work with the Navy on this development project. The rumor, however, was that the AEC was going to reply that it was reluctant to assign this project to a particular Laboratory because the requirements had not yet been specified. It was also indicated during the WLPC meeting that the Mark 500 had been approved once again.

(b)(3)

Indeed, the rumor concerning a delay in the selection of the weapon design laboratory, in terms of the Phase 3 assignment, was correct. In reply to the Foster letter of September 25, 1972, James R. Schlesinger, AEC Chairman, had stated, "Since the specific selection of a Phase 3 design can have a great impact on the selection of a laboratory, we shall hold that selection in abeyance until we have received your Phase 3 request." He had also added, "Our FY 1974 budget does not include the capital funds needed to support the 1977 IOC." The Schlesinger letter was followed by a memo sent from the AEC Washington D.C. office to the AEC laboratories on October 27. This memo states, "The Phase 3 request is not expected for several months as the Navy has not yet forwarded their recommendations on the Phase 2 study to DDR&E." The memo also notes, "A Laboratory pair for the development engineering of the Mk 400 will not be selected until such time as the receipt and review of the Phase 3 request with the MC's and STS. The Navy Department has been requested to keep all AEC laboratories advised and abreast of all planning concerning the Mk 400 until a lead laboratory pair can be selected."²¹⁸

Late in October, the administration at LLL apparently proposed to Camm that because of their 14-year working relationship with the Navy and because they had developed a 2D explosion code, the Phase 3 on the Mk 400 should be given to LLL.²¹⁹

(b)(3)

²¹⁸USAEC, William B. Haidler, Wash., D.C. to AN3, USAEC, H. C. Donnelly, Albuq., N.M. et. al. (SRD) (October 27, 1972), 2 pp.; USAEC, William B. Haidler, Wash., D.C. to AN3, USAEC, H. C. Donnelly, Albuq., N.M. et. al. (SRD) (November 10, 1972), 6 pp., B11, Drawer 56, Folder 1 of 4. ²¹⁹ Military Applications Planning Committee Minutes of the 17th Meeting," TDW-48 (SRD) (December 15, 1972), p. 2, B11, Drawer 53, Folder 1 of 2.

SACKEN/BO(

UNCLASSIFIED

\$\$\$\$\$\$\$

In a letter dated October 31, 1972, R. B. Olwin from Los Alamos wrote to the Navy Strategic Systems Project Office at Lockheed. The purpose of the letter was to indicate that the Los Alamos team had made significant improvements in the Mk 400 since the Phase 2 input dated February 29, 1972.

(b)(3)

In papers dated November 2, 1972, and titled, "MK-400 X-Ray Hardening Weight Penalty Calculations" and "Warhead X-Ray Hardening," R. S. Dingus discussed the x-ray hardening considerations for the Mk 400.²²¹

On November 8, 1972, Olwin wrote a letter to inform the groups working on the Mk 4 of the need for drawings describing the basic WH layout of the Mk 400.

(b)(3)

In a phone call to Chuck Gilbert at DMA, Livermore's McDonald objected to this proposal. However, the Navy representatives called a meeting to be held at Lockheed to discuss how to respond to the Camm proposal. A preliminary meeting of the AEC representatives was held at Livermore on November 15

(b)(3)

²²¹R. S. Dingus, "Warhead X-Ray Hardening," WX-6-72-114 (SRD) (November 2, 1972), 9 pp. R. S. Dingus, "Mk -400 X-Ray Hardening Weight Penalty Calculations," WX-6-72-115 (SRD) (November 2, 1972), 9 pp., B11, Drawer 56, Folder 2 of 4.

(b)(3)

LA-14066-H

NCLASSIFIED

Another meeting was held near the end of November between LASL and LMSC representatives.

(b)(3)

SACRAN/BOA

UNCLASSIFIED

The trip report from the Los Alamos attendees included the following gossip, "Both with Howard [Howard Trudeau, Manager, Reentry Systems] and all other LMSC personnel I have noticed a definite change in their attitude from polite tolerance to an anticipation of working together on the Mk 400."

(b)(3)

This is an extremely

LA-14066-H

sore point with the LMSC working troops and they are convinced that ALO has been brainwashed by Jim Wright, SLL." Additional information on detailed designs and studies discussed by LASL and LMSC representatives at this meeting is available in the cited reference.²²⁴

A TWX was sent out on November 28, 1972. from Vespe in the Albuquerque Operations Office. (b)(3)

The minutes of the December 4, 1972, WLPC meeting have an interesting bit of information. The minutes report that on the Friday before the WLPC meeting, Hoyt had presented to General Camm in Washington "...the case for assigning the Mark 400 warhead Phase 3 job to LASL." LASL representatives were optimistic that the "assignment of this warhead to us is in the bag."²²⁶

On December 5, 1972, F. W. Kramer and L. A. Ney from WX-3 and representatives from Sandia met with the LMSC representatives to discuss possible Mk 400 mounting arrangements...

(b)(3)

/Indeed, the LMSC group might propose it as the base line underlay at the December meeting of the vulnerability subcommittee.²²⁷

(b)(3)

²²⁵D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 105, December 4, 1972,"
 ADW-323 (SRD) (December 5, 1972), 3 pp., B11, Drawer 53, Folder 1 of 2.
 ²²⁷T. A. Sandford, ADWP-1 to Distribution, Subject: "Trip Report – Mk 400 Discussions at LMSC (U),"
 ADWP-1-72-93 (SRD) (December 19, 1972), 2 pp., B11, Drawer 56, Folder 3 of 4.

SALARATIKADA

UNCLASSIFIED

(b)(3)

- VI-56

The Phase 2 Study Group Meeting was held at LMSC on December 12, 1972. R. B. Olwin and Paul Vander Maat represented LASL. Olwin reported that Marc Meserole appeared to "be dragging his feet" in terms of sending out a Phase 2 supplement.

(b)(3)

R. S. Thurston from WX-6 attended the December 13–14 vulnerability subcommittee meeting. As a result of the discussions during this meeting, Thurston requested that representatives from WX-3, 5, 6 and TD-4, 6 meet on December 22 to discuss how much neutron hardening would be required to meet the Mk 400 threat specification. Thurston noted, "Consequently, a realistic appraisal of the neutron vulnerability problem is necessary to properly assess the engineering problem."²³⁰ In addition to Thurston, LASL was represented at the vulnerability meeting by D. R. Koenig, R. S. Dingus, A. W. Charmatz, S. R. Skaggs, T. P. Seitz, W. J. Johnson, and R. B. Olwin. The trip report from Olwin noted that there had been additional funding cuts. These cuts had forced SSPO into a second slip of the Trident program; a ten-month slip would be required in the overall program. However, the IOC was still October 1978.

(b)(3)

Jake Jacobs from LMSC had noted that the mid-frustum experiments were also slipping; one reason for the delay was the delay in the selection of a Phase 3 laboratory. Olwin also reported that the budget cuts had caused a great deal of SSPO interest in the ABC program. (The reader will recall that the ABC program was in some respects the <u>Mk 18 program.</u>)

(b)(3)

/The Skaggs presentation generated a lot of interest in the attendees from the Navy. Olwin ended his trip report with the statement, "We should not relax our efforts to pry the Phase 3 letter loose."²³¹

The final quarterly progress report from the Laboratory for 1972 indicated that several projects to study the weaponization, in terms of mounting and vulnerability, of the LASL-proposed Mk 400 warhead were being undertaken

(b)(3)

²³¹R. B. Olwin to H. C. Hoyt, Subject: "Mk 400 Phase 2 Study Group Meeting December 12, and Mk 400 Vulnerability Working Group Meeting, December 13-14," ADWP-72-33 (SRD) (December 18, 1972), 5 pp., B11, Drawer 56, Folder 4 of 5.

(b)(3)

SAADEMIN

UNCLASSIFIED

¥1-57

LA-14066-H

SACREAME

Hoyt reporting back on January 8, 1973, to the WLPC concerning his meeting of the previous week in Washington noted that the Phase 3 letter on the Mark 400 warhead was still not ready for signing. In fact, it was not clear exactly where the letter was in the Navy Department. However, DDR&E had drafted a letter to the AEC that requested a Phase 3 for the program.

(b)(3)

Indeed, all the engineering

development work had not yet been completed; how such a system could operate was not precisely known.²³⁶

Ever hopeful that LASL would receive the Phase 3 award, on January 31, 1973, Olwin and Sandford outlined some of the items that should be considered in the LASL program for the Mk 400 in the coming year. They included a tentative development schedule. However, they noted that the Navy had received deep cuts in funding for the program; the schedule for the Trident program would have to slip. The IOC was now October 1978.²³⁷

9. Continued Hope for a Phase 3? Maybe

At the February 1, 1973, of the WPRC meeting, Hoyt predicted that the Phase 3 assignment for the Mk 400 warhead would be made during the second half of March

(b)(3)

²³⁶UCLLL William B. Shuler, L-31, Livermore, Calif. to C13/LASL Mail and Records Attn. Harry C. Hoyt/Richard B. Olwin (SRD) (January 30, 1973), pp. 1–2, B11, Drawer 115, Folder 1 of 4.
 ²³⁷R. B. Olwin and T. A. Sandford to Distribution, Subject: "Guidance for FY 74 Mk 400 Operating Plans," ADWP-73-6 (SRD) (January 31, 1973), 4 pp., B11, Drawer 115, Folder 1 of 4.

SEGRET

م الله مال الم الله الله الله الله

LA-14066-H .

SALAR VIE

The Advanced Technical Objectives Working Group for Sea-based Deterrence met at Livermore on January 30, 31, and February 1, 1973. The Trident program was discussed, including the development schedule, in detail. The minutes of the meeting are quite detailed and cover many interesting topics. For additional information, the reader is referred to the cited reference.239

Early in February, a Mk 400 Mini-POM meeting was held with representatives from SLL, LLL, SLA, LMSC, and KSC attending as well as representatives from Los Alamos. It was reported that a draft_Phase 3 letter had been sent out by DDR&E on January 26, 1973

(b)(3)

During the meeting, Howard Trudeau from LMSC had presented schedules based on the latest slip to an IOC (for both the Trident subs and missiles) to October ·1978,

(b)(3)

The LASL group had assured Stinner that a failure was not likely.)²⁴⁰ At the February 14, 1973, meeting of the WWG, Gillespie reported that the Phase 2 report had been completed many months ago.

Gillespie noted (b)(3) that the team was also working with the Navy and Lockheed on another look on the use of sealed warheads. The Phase 3 request was being held up in the Pentagon.²⁴¹

Early in 1973, the Lockheed engineers had started to reevaluate the baseline Mk 400 RB design. Included were ablators, substructures, sealed WHs, and sealed reentry bodies. Because of the many design changes from the initial Phase 2 work, the DMA on February 26, 1973, requested the AEC laboratories to provide the DMA with the current thinking on nuclear designs for the Mk 400. (The LASL input was submitted on March 1.)²⁴²

At a meeting held on February 27-28, it was informally indicated that the C4 missile development might be bypassed in favor of the development of the C5 or D5.²⁴³ Thus, the C4 missile/Mk 400 RB program continued to hang in limbo.

In a letter from Agnew to Camm, dated February 27, 1973, Agnew noted that during recent months, the emphasis in the Mk 400 program had been on the overall design studies. These concentrated on "marrying" the warhead and the reentry body. LMSC scientists had completed much of the required engineering work. In this effort, the Los Alamos and Lockheed groups had established a good working relationship. Agnew reported that at Los Alamos he planned to have Dick Olwin manage the Mk 400 program. Agnew then stated, "Since Olwin has been on board

²⁴²Leslie M. Redman, "LASL Weapons Quarterly (U), for the Period Ending March 31, 1973," Los Alámos Scientific Laboratory report LA-5330-PR (SRD) (June 1973), pp. 57-58.

²⁴³T. A. Sandford and T. P. Seitz to Distribution, Subject: "Simi-Annual ABRES Review Meeting on February 27 and 28, 1973 (U)," ADWP-1-73-32 (SRD) (March 6, 1973), p. 3, B11, Drawer 113, Folder 2 of 3.

LA-14066-H

VI-59

ARCHER AND UNCLASSIFIED

²³⁹"Minutes of the Meeting of DNL Advanced Technical Objectives Working Group for Sea-Based Deterrence (U), 30-31 January and 1 February 1973," (SRD) (no date), 6 sections, B11, Drawer 115, Folder 4 of 4.

²⁴⁰R. B. Olwin to Harry C. Hoyt, Subject: "Mk 400 Mini-POM," ADWP-73-10 (SRD) (February 5, 1973), 3 pp., B11, Drawer 115, Folder 1 of 4.

²⁴¹"Weapons Working Group, Minutes of the 248th Meeting," ADWB-22 (SRD) (February 23, 1973), p. 2, B11, Drawer 111, Folder 2 of 3.



and active in the Mk 400 Program for some time now, we feel that LASL is in the best possible position to proceed with the Mk 400 Phase 3 Program."²⁴⁴

The February 27 Agnew letter to Camm was followed by a letter to Camm dated February 28, 1973. This time the letter was from the Assistant Director for Weapon Planning, Harry C. Hoyt,

(b)(3)

'Hoyt stated, "Each of the warhead designs will meet all requirements of the current Mk 400 STS document."²⁴⁵

At the WPRC meeting held on March 1, 1973, it was announced that the Phase 3 assignment for the Mk 400 was expected by the middle of March.²⁴⁶ However, at the WPRC meeting held on March 22, 1973, it was announced that the Phase 3 assignment for the Mk 400 was not expected until late April.

(b)(3)

At the March 26, 1973, meeting of the WLPC, Hoyt reported that it was believed that the Mk 400 program had been approved by the Chief of Naval Operations and that the approval was waiting for a signature by the Secretary of the Navy.²⁴⁹

The supplement to the Phase 2 Feasibility Study was released late in March. The objective for the supplement was summarized, "The objective of this supplement is to update the MK400 weight, base diameter and range values and to describe in detail the advantages and disadvantages of the proposed tritiated warhead."²⁵⁰

During the April 18, 1973, meeting of the WWG, Olwin reported on the status of the Mk 400 program. He indicated that the Phase 3 letter on the Mk 400 was believed to be in Foster's office, but there was no information as to which laboratory would receive the assignment. Olwin pointed out that the Trident system would use a stellar inertial guidance system; this meant that the target accuracy would be less dependent on the launch team knowing the exact position of the submarine.^{(b)(3)}

²⁴⁴H. M. Agnew to Major General Frank A. Camm, Assistant General Manager for Military Applications, DIR-2299 (SRD) (February 27, 1973), 4 pp., B11, Drawer 115, Folder 1 of 4.

²⁴⁵Harry C. Hoyt to Major General Frank A. Camm, ADWP-73-17 (SRD) (February 28, 1973), 4 pp., B11, Drawer 115, Folder 1 of 4.

²⁴⁶Roger B. Perkins to Distribution, Subject: "Minutes of the 1 March 1973 WPRC Meeting," AP-11 (SRD) (March 5, 1973), p. 1, B11, Drawer 111, Folder 2 of 3.

(b)(3)

²⁴⁹D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 109, March 26, 1973," ADW-357 (SRD) (March 27, 1973), 3 pp., B11, Drawer 111, Folder 2 of 3.

²⁵⁰"Joint AEC/DOD MK400 Phase 2 Feasibility Study Report (U), (Supplement)," Strategic Systems Project Office, Department of the Navy, Washington, D.C. 20360 report K-26473 (SRD) (March 27, 1973), 25 pp., B11, Drawer 115, Folder 1 of 4.

/ SALCINE AURON

UNCLASSIFIED

LA-14066-H

(b)(3)

Agnew suggested that a copy be sent to General Giller.²³¹

H. N. Meyer, Jr., from the AEC's Albuquerque Operations Office, wrote an interesting memo dated April 19, 1973. Meyer reported that the Navy's representatives had informed him that the Navy was considering refitting a Poseidon boat with C4 missiles/Mk 400 RBs. This operation had a proposed date of having the refitted boat at sea in August 1978. The Navy would follow this with the loading of C4 missiles/Mk 400 RBs on the first Trident boat. This operation had a proposed date (for the boat to be at sea) of September 1978. In October, a second refitted Poseidon boat would be at sea, and in November the third refitted Poseidon boat would also be at sea. This new schedule then called for about 2 1/3 times more warheads than the original Navy proposal for the fitting of one Trident in October 1978. Meyer noted, "Mr. J. K. Davis, SSPO, was told that these delivery requirements represent a major change in our [AEC] planning. To seriously address our delivery capability would require a detailed examination that would take a minimum of a few months." Meyer noted that the Navy's Admiral Zumwalt apparently intended to pursue the new plan with the appropriate people in DOD/OSD (Office of the Secretary for Defense) for presentation to Congress.²⁵²

(b)(3)

10. Finally-A Phase 3 Assignment

In a letter from John S. Foster, Jr., Director of Defense Research and Engineering, Department of Defense to AEC Chairman, Dixy Lee Ray, dated April 23, 1973, Foster requested the AEC to join with the Department of Defense in a Phase 3 development program of the Mk 400 warhead.

He noted that a reduction to the use of eight reentry bodies per missile had been accepted. (b)(3) The original specifications for size and weight requirements had been relaxed. Foster noted, "The need to minimize system total acquisition cost, AEC as well as DOD, cannot be overemphasized." One-point safety was specified. An RB weight of about 200 pounds and a length of 65 inches maximum were also specified. The Navy was designated cognizant agent for the Department of Defense portion of the development; the Navy was to submit the proposed Military Characteristics to the Military

(b)(3)

LA-14066-H

MARETONN LASSIFILD

²⁵¹"Weapons Working Group, Minutes of the 249th Meeting," WWG-249 (SRD) (February 23, 1973), pp. 8–9, B11, Drawer 111, Folder 2 of 3.

pp. 8–9, B11, Drawer 111, Folder 2 of 3. ²⁵²H. N. Meyer, Jr. to Memo to Files, Subject: "Mk 400 Planning," (SRD) (April 19, 1973), 2 pp., B11, Drawer 115. Folder 1 of 4.

Liaison Committee for review and transmittal to the AEC. An IOC of October 1978 was desired.²⁵⁴

During the April 27 meeting of the WLPC, it was announced that the Mk 400 and its warhead had been put into Phase 3. The warhead development had been assigned to LASL/SLA. [Author's note: The Camm TWX, dated April 27, 1973, assigning the Mk 400 to LASL is discussed in the following paragraph.] Olwin, who had been in Hoyt's organization, would transfer to the Weapon Program Office where he would be the manager for the Mk 400 warhead program. The requirements for the warhead had been spelled out in a letter from Foster to the AEC Chairman. The desired IOC was October 1978. After these announcements, Agnew instructed that an immediate review of the test program take place in order to make sure that it was consistent with this potential task "that has now become a reality." It was noted that the reason cited by General Camm for assigning this task to LASL was that LLL had developed all of the strategic missile warheads for the past ten years, and to maintain the validity of the assertion that the two Laboratories were competitive across the board, LASL should have one of these jobs. MacDougall remarked in his minutes of the meeting that it was obvious that the Laboratory must do its absolute best to make the program a success.²⁵⁵

During the May 3, 1973, WPRC meeting, it was again announced that LASL had received a Phase 3 request to develop the Mk 400 warhead. The letter from Foster and a TWX from General Camm was included as an appendix in the meeting minutes. The TWX from General Camm dated April 27, 1973, assigned the project to Los Alamos. The Camm TWX stated, "After carefully reviewing factors relevant to Mk 400 Phase III development engineering, I have concluded assignment to LASL would contribute most to overall viability of the two-laboratory competitive concept. This concept has proven its value many times over in meeting National Defense needs in imaginative ways which led to dramatic improvements in defense capabilities while at the same time reducing greatly overall system costs for achieving specific Military effects." But Camm also noted, "Selection of LASL rather than LLL should in no way be construed as a reflection on LLL capabilities and contributions. On the contrary LLL contributions have essentially monopolized strategic RV warheads for the last decade to the extent that the Mk 400 is an appropriate opportunity for LASL to design one." Camm ended his TWX by stating, "Accordingly, I am assigning the Mk 400 to LASL."²⁵⁶

In a May 3 letter to Captain Wayne L. Beech at the Division of Military Application, Hoyt outlined the Laboratory's plans for the Mk 400 program. He reported that suitable designs for the primary and secondary were being developed.²⁵⁷

STREAMED UNCLASSIFIED

LA-14066-H

²⁵⁴USAEC William B. Haidler, Wash, D.C. to AN3 USAEC, H. C. Donnelly, Albuquerque, N. M. et. al. (SRD) (April 24, 1973), 5 pp., B11, Drawer 115, Folder 1 of 4.

²⁵⁵D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 111, April 27, 1973," ADW-370 (SRD) (April 30, 1973), p. 1, B11, Drawer 111, Folder 2 of 3.

²⁵⁶Roger B. Perkins to Distribution, Subject: "Minutes of the 3 May 1973 WPRC Meeting," AP-14 (SRD) (May 8, 1973), 2 pp. and two enclosures, B11, Drawer 111, Folder 2 of 3.

²⁵⁷Harry C. Hoyt to Captain Wayne L. Beech, ADWP (SRD) (May 3, 1973), 1 p. and enclosure ADWP-73-25, p. 12, B11, Drawer 110, Folder 2 of 4.

In a memo dated May 10, 1973, to D. W. Bergen, MacDougall stated, "We are now starting a new program, the development of the W76. As long as I am around, I intend to ride pretty hard on you and Olwin. Since this one has a management team from the start, I hope that we can manage it better. I want to insist as hard as I can that things be better documented.

...In connection with the W76 program, I expect the program plans to indicate what you and Olwin expect in the way of accomplishments from each participating Division, with dates. If any Division is not living up to its commitments, I expect you and Olwin to come to me, if you can't get the thing straightened out yourselves with the Division Leader concerned."²⁵⁸

The AEC's acceptance of the program was contained in a letter signed by AEC Chairman Dixy Lee Ray and dated May 10, 1973. She noted that the design responsibility for the warhead had been assigned to the Los Alamos Scientific Laboratory and Sandia Laboratories. She stated, "We will continue to investigate designs utilizing minimal quantities of special nuclear materials and to search for the design that optimizes development and production cost reductions in meeting the envelope as specified in the Desired Characteristics." She also noted, "In the fall we will present the 17-week MLC [Military Liaison Committee] review on the status of weapon development engineering. At that time, the AEC and the DOD will jointly evaluate the program."²⁵⁹

Official authorization for the Phase 3 responsibility was contained in a memo from Camm at the AEC Division of Military Application dated May 17, 1973. This memo states, "The purpose of this memorandum is to assign the Phase 3 development responsibility for this warhead to the Los Alamos Scientific Laboratory and the Sandia Laboratories." The memo also states, "The interface responsibilities should be resolved and firmed up by the POG as soon as possible. [The Project Officers Group (POG) would be formed in June 1973.] If problems arise that cannot be easily resolved concerning this matter, do not hesitate to bring them to the attention of DMA." Included as part of the enclosure to this memo was a copy of the Desired Characteristics. It was reported, "The Military Characteristics and the Stockpile-to-Target Sequence are being staffed within the Department of Defense and will be provided as soon as they are approved and received."²⁶⁰ This authorization report is considered to be the defining memo in the assignment of the Phase 3.²⁶¹

During the May 21, 1973, meeting of the WLPC, it was reported that the reentry body for the C4 missile previously referred to as the Mk 400 had now been designated the Mk 4. The warhead for this reentry body had been designated the W76. The program had been assigned to Los Alamos on April 27.²⁶²

²⁶¹"Final Development Report for the W76-0/MK4 Reentry Body (U)," Sandia National Laboratories and Los Alamos Scientific Laboratory report SAND78-2064 (SRD) (April 1980), p. 5.

²⁶²D. P. MacDougall to Members, WLPC, Subject: "WLPC Meeting No. 112, May 21, 1973," ADW-385 (SRD) (May 22, 1973), p. 1, B11, Drawer 111, Folder 2 of 3.

LA-14066-H

NCLASSIFIED

²⁵⁸D. P. MacDougall to D. W. Bergen, Subject: "Program Management, the W74 and the W76," ADW-376 (SRD) (May 10, 1973), 2 pp., B11, Drawer 116, Folder 1 of 2.

²⁵⁹Dixy L. Ray, Chairman to Honorable John S. Foster, Jr., Director of Defense Research and Engineering, Department of Defense (SRD) (May 10, 1973), 1 p., B11, Drawer 116, Folder 1 of 2.
²⁶⁰Frank A. Camm to H. C. Donnelly, Manager, Albuquerque Operations; M. Sparks, President, Sandia Laboratories; R. D. Thorne, Manager, San Francisco Operations; H. M. Agnew, Director, Los Alamos Scientific Laboratory, Subject: "Phase 3 Authorization for a Reentry Body for the Trident I (C-4) Submarine Launched Ballistic Missile," (SRD) (May 17, 1973), 2 pp. and enclosures, B11, Drawer 116, Folder 1 of 2.

(b)(3)

The second quarter report from the Laboratory for 1973 noted that the Mk 400 was the reentry body for the C4 Trident missile; it was being designed for use on the new Trident submarine

(b)(3)

! (Phase 3 development of the W76 WH for the Mk 4 RB was awarded to LASL and SLA on April 27, 1973.)"²⁶⁴

11. Significance

It had been a long and at times bitter fight. But the laboratory at Los Alamos had won the long desired strategic warhead assignment. However, the laboratory staff was now under extreme pressure to develop and design a warhead that would (1) meet the yield requirement, (2) meet the size and weight requirements, (3) achieve the vulnerability and safety requirements, and (4) at the same time satisfy the minimum number of NTS development tests, the strict time restraints, as well as the budget limits that had been placed on weapon development.

It was never envisioned that the in the year 2003 planning would be in progress to retain the W76 in the U.S. stockpile.

C. Harold Agnew

1. Proponent

While the entire staff at Los Alamos worked hard to obtain the Phase 3 for the Mk 400, one of the chief proponents for this award was Harold Agnew. (From the information given in Chapter I, the reader will recall that Agnew became Laboratory Director in 1970.) Agnew felt that there were many reasons why the award should go to Los Alamos

(b)(3)

²⁰⁴Leslie M. Redman, "LASL Weapons Quarterly (U), for the Period Ending June 30, 1973," Los Alamos Scientific Laboratory report LA-5401-PR (SRD) (September 1973), p. 58.

SECRETARD

UNCLASSIFIED

VI-64

LA-14066-H

Agnew voiced his viewpoint in numerous letters and TWXs to Washington. Considering the fact that he was on many panels and committees that included military personnel, Agnew must have also presented his viewpoint informally at appropriate moments.

The Agnew concerns will be discussed in greater detail in the following sections.

2. Reliability

The folklore among the "old timers" at Los Alamos is that it was Agnew that pressed for the testing of Livermore's mechanical safing systems employed in some of their weapons. The objective of these tests was to determine if the safing system could actually be withdrawn in order to prepare the warhead for activation. The reader of this author's previous document, LA-13755-H (SRD), will recall that when these tests were undertaken not all the safing systems worked as required. The question then became "How many duds in the system?" Agnew apparently noted this problem at opportune moments. For example in a November 29, 1966, letter to General Earle G. Wheeler, Chairman, Joint Chiefs of Staff, Agnew wrote, "I have been worrying for several years about the actual implementation of what we commonly refer to as our assured destruction capability. Recently a great deal of effort has been directed toward reducing the vulnerability of the forces we have for this role against possible enemy action throughout all stages of their stockpile-to-target sequence. However, I wonder if we have protected ourselves to the degree we might have against basic system component design failures which to some degree or another always turn up. The most recent example is the MK-47 Polaris problem." In discussing the prospect for the Mk 3 and Mk 18 carrying the same warhead, Agnew wrote, "If a defect turned up in the warhead some years from now the complete force could be in trouble." He went on to say, "If I were in a position to make such decisions I would suggest that in a missile system such as Poseidon or the new Minuteman that there be a mix of basic missiles, perhaps a different mod for every one or two hundred missiles and a mix of RV's with warheads. I would have a warhead mix such that no more than a quarter of the missile warheads were identical assuming a total build of over a thousand. The different RV's could also pose additional problems for any enemy ABM system."265

Delmar Crowson, Director of Military Application, agreed with Agnew. A few months after the Agnew letter, in a memo dated February 9, 1967, for the Chairman, Military Liaison Committee, Crowson wrote, "I believe that Dr. Agnew's suggestion has considerable merit. ...Three examples in the strategic missile warhead stockpile serve as cases in point: the recent ANA (Actuator, Nuclear Arming) failure in the MK 47Y2 warhead for POLARIS; the highaltitude failure of the arm-safe inspection port in the MK 58 warhead (MK 2 R/V) for POLARIS; and the ANA failure in the MK 56 warhead for MINUTEMAN. In each case it has been comforting to have alternate warheads in the stockpile..."²⁶⁶ [Author's note: The three warheads cited by Crowson were all designed at Livermore.]

LA-14066-H

SEGRATION UNCLASSIFIED

²⁶⁵H. M. Agnew to General Earle G. Wheeler, USA, W-1989 (SRD) (November 29, 1966), p. 1, B11, Drawer 56, Folder 1 of 4.

²⁶⁶Delmar L. Crowson, Memorandum for Chairman, Military Liaison Committee to the U. S. Atomic Energy Commission, Subject: "Concept of Mixed Warheads for Strategic Missiles," (SRD) (February 9, 1967), 2 pp., B11, Drawer 56, Folder 1 of 4.



Again in a November 8, 1968, letter to John S. Foster, Director of Defense Research and Engineering, Agnew stated, "In addition to the general point about having to worry about the whole Poseidon force being down due to some unforeseen failure in hardware, recent results at the LASL and LRL lead me to believe that we are getting awfully close to complete unreliability in very small advanced systems and we don't know why now! Our test results (LRL and LASL) certainly cannot be dismissed without concern. We seem to be getting squeezed between one point safety and repeatable performance." ²⁶⁷

3. Advantages of Diversity

A slightly different, but related argument that Agnew made was that not "all the eggs should be placed in one basket." By this he meant that he felt that the stockpile would be more reliable if both laboratories, Livermore and Los Alamos, contributed devices.

In the November 8, 1968, letter to John Foster, Agnew in particular seems to have been worried about the Poseidon. He stated, "If I were in charge I would require that the Poseidon build have two or three different nuclear systems and I would insist that Minuteman be different." Agnew closed his letter by stating, "...don't let the AEC make the nuclear warheads all the same unless you really want to get the country in worse shape than it is already."²⁶⁸ By these arguments, Agnew appears to have been implying that the two-laboratory approach would be the best.

Agnew was at least somewhat successful in convincing the Military of the benefit of the two-laboratory approach. In reporting on a briefing for the Under Secretary of State, Elliot L. Richardson, conducted on August 26, 1969, Agnew stated, "Giller made a strong pitch about not putting all our eggs in one basket..."²⁶⁹

In a letter to Camm dated August 10, 1972, Agnew indicated that the Military was "very desirous of obtaining a mix of warheads representing the AEC laboratories' different design philosophies in order to insure better against a catastrophic failure in the future that might occur if all their 'eggs' continue to be in one basket."²⁷⁰

Agnew once again repeated his concerns in a letter to Major General F. A. Camm, dated November 27, 1972. Agnew stated, "The strategic missile systems, with the exception of Titan, now use LLL-designed warheads. We question the advisability of relying on a single source for such an important part of our deterrent force."²⁷¹

Agnew also made the argument that the two-laboratory approach kept each Laboratory "honest." In his letter to Camm on August 10, 1972, Agnew wrote, "Neither Laboratory can afford to make either wildly optimistic claims, or, on the other had, make only overly conservative statements about what can be achieved."²⁷²

VI-66

/SECREDIED

UNCLASSIFIED

LA-14066-H

 ²⁶⁷H. M. Agnew to John S. Foster (SRD) (November 8, 1968) 1 p., B11, Drawer 56, Folder 1 of 4.
 ²⁶⁸H. M. Agnew to John S. Foster (SRD) (November 8, 1968) 1 p., B11, Drawer 56, Folder 1 of 4.

²⁶⁹H. M. Agnew to WLPC, W-2208 (SRD) (August 27, 1969), 2 pp., A99-019, 91-10.

²⁷⁰H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), pp. 6–7, B11, Drawer 56, Folder 1 of 4.

²⁷¹H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), p. 3, B11, Drawer 56, Folder 1 of 4.

²⁷²H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), p. 6, B11, Drawer 56, Folder 1 of 4.

UNCLASSIFIED AARAAMA

4. **Engineering Philosophy**

Another argument, related somewhat to the "don't put all your eggs in one basket" argument, was the point that Agnew made about the differences in engineering philosophy between the two laboratories.

In a letter to Camm dated September 11, 1972, Agnew wrote, "One strength of this Laboratory has been its consistent ability to meet the actual needs of the military with realistic design concepts and practical engineering. In addition, we are very conscious of development and production costs and involve our weapon engineering people in the design from the start. This has been an important aspect of our ability to minimize costs and stay within budgets while at the same time delivering reliable hardware. We feel that our past experience with Mk 18 and ABC, together with our current Mk 400 efforts, will enable us to continue this for the Mk 400 warhead."273

In his letter to Camm on November 27, 1972, Agnew reported, "As we have made clear, there are differences in basic physics design philosophies and engineering approaches between the two laboratories." He indicated, "In particular, LASL design philosophy is to rely more on demonstrated fabrication techniques (often worked out and demonstrated at LASL), and simplicity in design wherever possible. We feel that our approach has led to significant differences between our warhead design and those of LLL. A mixture of LLL and LASL warheads definitely should improve the confidence in the strategic missile deterrent." Agnew also noted "The AEC has always supported the two laboratories to make certain that, through competition, different options would be available to them and to the DoD. Not to take advantage of these options is to ignore the basic rationale for supporting the opportunity to provide the options in the first place. In today's climate this point should receive very serious weighting in your deliberation."²⁷⁴

5. Cooperation

Agnew tried to make it clear that the LASL group was willing to cooperate to the fullest extent and to provide whatever the Military wanted.

On September 11, 1972, Agnew wrote Major General Camm, "It is important also to realize that we have worked actively with Lockheed and SSPO for the past two years on the Mk 400 and before that on the CAFE program and have a good working relationship with both... Stan Burriss, President of Lockheed, is very friendly to the LASL having been a senior member of LASL before joining Lockheed."275

LA-14066-H

SECRETARIO UNCLASSIFIED

· VI-67

²⁷³H. M. Agnew to Major General Frank A. Camm, DIR-2293 (SRD) (September 11, 1972), 3 pp., B11, Drawer 56, Folder 1 of 4.

²⁷⁴H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), 7 pp., B11,

Drawer 56, Folder 1 of 4. ²⁷⁵H. M. Agnew to Major General Frank A. Camm, DIR-2293 (SRD) (September 11, 1972), 3 pp., B11, Drawer 56, Folder 1 of 4.

In his November 27, 1972, letter to Camm, Agnew implied that all the LASL team was eager to work with the Navy and the Navy's contractors and vice versa. Agnew stated, "In a nutshell, the Navy, Lockheed and the Air Force wish to work with us." Agnew also noted, "Industry and the DoD sense that they are missing an important design input and working relationship by not interacting directly with the LASL in the strategic offensive warhead area."276

UNCLASSIFIED

ł

LA-14066-H

6. **Program Balance**

Agnew felt that to remain a viable laboratory, each laboratory must work on all the different types of weapons going to the stockpile.

In a letter dated November 17, 1970, to Michael May at Livermore, Agnew stated, "...I believe it would not be a healthy condition for the country, the AEC, or the laboratories if the LASL were to concentrate on providing only tactical offensive warheads and bombs to the stockpile and LRL were to limit its endeavor to strategic offensive warheads." Agnew informed May, "... we are concentrating a sizeable portion of our advanced development technology on getting into a position to respond rapidly to the need for a new strategic offensive warhead whenever the AEC is called upon to provide one.

(b)(3)

In a letter dated August 10, 1972, Agnew noted that during the late 1950s time period it had been agreed that both the Los Alamos and Livermore laboratories should maintain competence in all areas of weapon design; each laboratory would not just specialize in one area of nuclear weapon technology. He stated, "At the same time, it was agreed by both laboratories and by DMA that both laboratories would maintain competence 'across the board' rather than have one laboratory specialize in other areas of nuclear weapon technology."278

In his November 27, 1972, letter Agnew noted, "To achieve proper balance, a design laboratory must receive design responsibility for weapons program (i.e., Phase 3 assignments) in all areas of nuclear weapons within a reasonable time period. Otherwise, capabilities can atrophy or disappear because of the feeling that 'We never get those jobs.' When this happens the advantages of two-laboratory competition in the nuclear weapon field will disappear." 279

Agnew also reported that he expected the W74 effort to decline at the same time that the Mk 400 effort was projected to increase. Agnew noted, "Consequently we anticipate no staffing problems for the Mk 400 Program. We need the work."280

²⁷⁶H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), 7 pp., B11, Drawer 56. Folder 1 of 4.

(b)(3)

²⁷⁸H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), 7 pp., B11, Drawer 56, Folder 1 of 4.

²⁷⁹H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), 7 pp., B11, Drawer 56, Folder 1 of 4.

²⁸⁰H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), 7 pp., B11, Drawer 56, Folder 1 of 4. /SFARATINEM UNCLASSIFIED

7. Yield: The Confetti Argument

Agnew felt that the yield of the W68 was too low to be really effective. In addition, in terms of the overall total yield available from all the W68 warheads, the W68 design was very costly in terms of the amount of required special nuclear materials.

In an April 1972 TWX to Assistant Director for Safety and Liaison (Division of Military Application) Colonel Robert T. Duff, Agnew reported that he was worried about maintaining the U.S. nuclear deterrent. Agnew noted, "It occurs to me that as we go to lower and lower yields in our strategic missile warheads and the Soviet Union builds up a better and better civil defense position, the reality of this deterrent may become questionable.

(b)(3)

If the Soviet leadership believes this, then our strategic deterrent will have lost a good deal of its force. If our MIRV trend continues we'll be threatening to throw confetti at a potential aggressor. Confetti has high penetration and survivability but little deterrent power."²⁸¹

In a letter dated October 10, 1972, to Giller, at that time Assistant General Manager for National Security, Agnew again noted several reasons why low yield warheads might not be the best solution for maximizing the deterrence capability of the stockpile. He reported that considering the number of required submarines and the low efficiency in their use of special nuclear material, the low-yield warheads were not very cost effective. Moreover, Agnew pointed out that for the Hiroshima device, the effects on Hiroshima in terms of loss of substantial buildings and the people in them "wasn't all that impressive." In terms of loss of life, the USSR had lost more than ten million people in WWII. Although the Soviets had an extensive civildefense network in place, even if that did not work to reduce loss of civilian lives, the Soviets might not mind losing a few people. Agnew wrote, "Again, to me, to continue to increase warhead numbers at the cost of a decrease in yield per warhead could eventually lead to no deterrence in the minds of those we hope to deter." Agnew stated, "I feel very strongly that we should endeavor to convince the DoD that what they should have on the next round is a mix of yields.

(b)(3)

8. Capability

Agnew in his August 10, 1972, letter to Camm pointed out that the Los Alamos group had been developing suitable technology applicable to the new strategic missile warheads. He wrote, "In summary then, we have been working very hard to provide the very latest technology in warhead designs incorporating the most advanced minimum weight hardening techniques to provide an optimum warhead for the next round of strategic missile warheads. In fact, our work has been of such outstanding quality that we have been invited by Admiral Levering Smith to

²⁸¹H. M. Agnew, University of California, Los Alamos Scientific Laboratory, Los Alamos, N.M. to BY3/Colonel Robert T. Duff, USAF, Assistant Director for Safety and Liaison, Division of Military Application USAEC, Wash., D.C. (SRD) (April 14, 1972), pp. 1–2, B11, Drawer 56, Folder 1 of 4.

LA-14066-H

(b)(3) **SACANA** UNCLASSIFIED

join his Steering Task Group for the FBM Weapon System Program in anticipation of our supplying TRIDENT with its warhead."²⁸³

Agnew felt that the LASL group had had extensive experience in the areas of vulnerability, hardening, and RV/warhead integration. Again, in a letter to Camm dated September 11, 1972, Agnew wrote, "Also, our in-house work and underground testing program in the vulnerability and hardening area have made a significant contribution to the nation's overall capability in this area." He also noted, "One unique advantage we have is that while we have a solid background of experience in reentry system design we are not tied to our past achievements and thereby inhibited in our approach to new designs."²⁸⁴ In his November 27, 1972, letter to Camm, Agnew gave examples of how the LASL team had been the leader in several aspects of the vulnerability program.

(b)(3)

/This experience)

had been demonstrated in the successful Mk 18 and Mk 400 programs. With money very tight and the need to limit the expense of testing at NTS, the LASL team was in the best position to develop the Mk 400 warhead with a minimum number of tests. Agnew told Camm, "Once the Phase 3 has been awarded, we will design a package in which both the primary and secondary are so configured as to provide the best possible warhead to satisfy the specific DoD requirements."²⁸⁶

9. Promise of the Next Strategic Missile Warhead

In his August 10, 1972, letter to Camm in which Agnew discussed the history of previous weapon assignments, Agnew noted that at the time of the Mk 3/W68 warhead assignment to Livermore the Los Alamos group had been promised the development responsibility "for the next strategic missile warhead, whatever it might be..."²⁸⁷

In a letter dated September 11, 1972, to Camm, Agnew again reminded Camm that LASL had been told that they would receive development responsibility for the next strategic reentry system. To meet this obligation, the members of the LASL weapon groups had been developing and testing warheads for the Mk 19, Mk 18, ABC, and Mk 400 programs. This work had resulted in the Laboratory being very involved in these types of systems. (b)(3)

Agnew reported, "I nese two tests cover the spectrum of possible secondary designs for the Mk 400." Agnew also noted that the design of a suitable primary for the Mk 400 program was underway. He concluded his September 11 letter by stating "We feel

²⁸³H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), 7 pp., B11, Drawer 56, Folder 1 of 4

²⁸⁴H. M. Agnew to Major General Frank A. Camm, DIR-2293 (SRD) (September 11, 1972), 3 pp., B11, Drawer 56, Folder 1 of 4.

(b)(3)

²⁸⁶H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), 7 pp., B11, Drawer 56, Folder 1 of 4.

²⁸⁷H. M. Agnew to Major General Frank A. Camm, DIR-2292 (SRD) (August 10, 1972), p. 4, B11, Drawer 56, Folder 1 of 4.

VI-70

UNCLASSIFIED

LA-14066-H

that our past experience with Mk 18 and ABC, together with our current Mk 400 efforts, will enable us to continue this [successful effort] for the Mk 400 warhead."²⁸⁸

10. Morale

Agnew was fearful concerning the effect that an award of the Phase 3, Mk 400 warhead to Livermore might have on Los Alamos weapon personnel. Agnew in his November 27, 1972, letter to Camm noted, "After having had the vision to work in this field and having been extremely successful, not to receive this assignment would have a very severe impact on our staff morale."²⁸⁹

²⁸⁸H. M. Agnew to Major General Frank A. Camm, DIR-2293 (SRD) (September 11, 1972), 3 pp., B11, Drawer 56, Folder 1 of 4.

²⁸⁹H. M. Agnew to Major General F. A. Camm, DIR-2296 (SRD) (November 27, 1972), p. 7, B11, Drawer 56, Folder 1 of 4.

LA-14066-H

READER ICLASSIFIED

This Page Intentionally Left Blank

UNCLASSIFIED

LA-14066-H

No. of Copies

1

1

1

1

1

1

UNCLASSIFIED

November 3, 2003

1

DISTRIBUTION

Terence R. Fehner, Historian History & Records Dept. ME-75, F-016/GTN U. S. Department of Energy P. O. Box A Germantown, MD 20874-0963

Brigadier General Ronald Haeckel NA-10, 4A-019/FORS U. S. Department of Energy P. O. Box 23865 Washington, DC 20026-3865

David H. Crandall NA-11, 4C-014/FORS U. S. Department of Energy P. O. Box 23865 Washington, DC 20026-3865

David E. Beck NA-12, 4A-045/FORS U. S. Department of Energy P. O. Box 23865 Washington, DC 20026-3865

Kerry P. Webb NA-113.2, B-321/GTN U. S. Department of Energy P. O. Box A Germantown, MD 20874-0963

Patricia Dedik, DIR NA-241, GA-007/FORS U.S. Department of Energy P. O. Box 23865 Washington, DC 20026-3865

Office of the Deputy Assistant to the Secretary of Defense (NM) 3050 Defense Pentagon Room 3C 125, Pentagon Attn: Rollin Whitman, OATSD (NCB) Nuclear Matters Washington, DC 20301-3050

FILL

LA-14066-H History

HET NAM

November 3, 2003

No. of Copies

1

1

1

1

1

1

2

Office of the Deputy Assistant to the Secretary of Defense (NM) 3050 Defense Pentagon Room 3C 125, Pentagon Attn: Steve Henry, DATSD (NM) Nuclear Matters Washington, DC 20301-3050

Office of the Assistant to the Secretary of Defense (NM) 3050 Defense Pentagon Room 3C 125, Pentagon Attn: Tom K. Scheber, OATSD (NCB) Nuclear Matters Washington, DC 20301-3050

Office of the Under Secretary of Defense for Acquisition and Technology 3150 Defense Pentagon Room 3D136, Pentagon Attn: Dr. Ronald Sega, 3E808 Washington, DC 20301-3150

Department of the Navy Chief of Naval Operations 2000 Navy Pentagon (N09B11) Attn: RADM Joseph Sestak, N-51 Washington, DC 20350-2000

Strategic Systems Programs 287 Somers Ct., Ste 10041, NW Attn: RADM Dennis Dwyer, SP-00 Washington, DC 20393-5446

Sandia Corporation Sandia National Laboratories Attn: Mail Services Section, Marjorie Gaddy Kadlec For: Technical Library P. O. Box 5800 Albuquerque, NM 87185-5800

Director Defense Threat Reduction Agency Attn: Document Control 8725 John J. Kingman Road MSC 6201 Ft. Belvoir, VA 22060-6201

ADDIGIED

1

1

1

UNCLASSIFIED

November 3, 2003

3

No. of Copies (Cont.)

Commander Albuquerque Field Operations Defense Threat Reduction Agency 1680 Texas Street, SE Attn: Robert W. Carter, FCIESC Kirtland AFB, NM 87117-5669

National Security Council Attn: Frank Miller, Room 379EEOB Records Management Eisenhower Executive Office Building Washington, DC 20504

U. S. Department of Energy National Nuclear Security Administration Nevada Site Office Attn: Loretta Bush, Technical Library, MS 505 P. O. Box 98518 Las Vegas, NV 89193-8518

University of California Lawrence Livermore National Laboratory P. O. Box 808 Livermore, CA 94551-0808 Mail Stop L-313 Attn: Distribution For: Mike Anastasio Bruce Tarter John Nuckolls **DNT Vault** Kent Johnson **Richard Ward B** Div Vault **Division Leader McMillan** Gary Carlson Frank Graziani Dana Rowley **Bob Tipton Dave Stanfel** Scott Carman Peter Stry Alan Wan Ivan Otero

Paul Miller

UNCLASSIFIED

November 3, 2003

No. of Copies (Cont.)

University of California Lawrence Livermore National Laboratory P. O. Box 808 Livermore, CA 94551-0808 Mail Stop L-313 Attn: Distribution (Cont.) For: Joe Bauer Omar Hurricane Jim Rathkopf Charlie Verdon

Los Alamos National Laboratory Mail Station 5000 P. O. Box 1663 Los Alamos, NM 87545 For: John D. Immele, Director's Office, MS A150

Carolyn A. Mangeng, Director's Office, MS A100 Ray Juzaitis, ADWP, MS F633 Richard Mah, ADWEM, MS F633 David Ponton, ESA-WSE, MS C936 Paul Hommert, X-DO, MS B218 Dawn G. Flicker, X-4, MS T086 William J. Krauser, X-DO, MS T082 Donald R. McCoy, ADWP, MS F633 Betty L. Perkins, DX-7, MS C322 Joseph A. Brown, S-7, MS F674 Michael C. Pankratz, S-7, MS F674 Michael Webb, TRO, MS B248 Gary Dean Wall, X-4, MS T086 Donald C. Wolkerstorfer, X-4, MS T086 Bruce Trent, X-4, MS T086 Don Sandoval, X-4, MS T086 Jim Hill, X-4, MS T086 David W. Scudder, P-22, MS D410 Diane E. Martinez, DX-7, MS P915 John Sarracino, X-5, MS F663 Robert Douglas Fulton, P-23, MS H803 James W. Ogle, DX-5, MS D411 Donald D. Cobb, ADTR, MS A150 Houston T. Hawkins, Director's Office, MS A150 Steve Girrens, ESA-DO, MS P946 Mary Hockaday, DX-DO, MS P915 Patrice Stevens, ADWEM, MS F633 Ezekiel D. Aragon, ADWEM, MS F633

A VALCEDSIC LELL

> CD 1

No. of Copies (Cont.)

UNCLASSIFIED

Los Alamos National Laboratory (Cont.) Mail Station 5000 P. O. Box 1663 Los Alamos, NM 87545 For: David A. Trujillo, ADWEM, MS F633

Roberta D. Idzorek, ADWEM, MS F633 Michael Haertling, ADWEM, MS F633 Thomas D. Kunkle, EES-11, MS F665 Ping Lee, Director's Office, MS A100 Michael Bernardin, X-2, MS T085 John Dallman, DX-DO, MS P915 David B. Harris, X-2, MS T085 Paul Dunn, MST-6, MS G770 J. David Olivas, NMT-10, MS E508 Jeffrey Paisner, DX-DO, MS H803 Fred N. Mortensen, X-2, MS T085 Rosemary Baltrusaitis, X-2, T085 Joyce A. Guzik, X-DO, T085 Harold H. Rogers, Jr., X-2, MS T085 Bernard P. Ginsberg, DX-1, MS P950 LCDR J. S. Baehr, D-2, MS F609

Roger A. Meade, IM-5, MS C322 (6 copies)

Jackson H. Carter, X-5, MS F663 (20 copies)

LANL Research Library, (50 copies) Report Section, STB-RL, MS P364

U. S. Department of Energy Office of Scientific and Technical Information Attn: Weapon Data Index P. O. Box 62 Oak Ridge, TN 37831

ASSIFIED

November 3, 2003

This Page Intentionally Left Blank

re*tvirkø* ASSIFIED

SECTREM ASSIFIED



\$ECREMIND

Los Alamos NM 87545